Western University Scholarship@Western

Digitized Theses

Digitized Special Collections

2011

DEGRADATION OF 1,2-DICHLOROETHANE WITH NANO-SCALE ZERO VALENT IRON PARTICLES

Ambareen Atisha

Follow this and additional works at: https://ir.lib.uwo.ca/digitizedtheses

Recommended Citation

Atisha, Ambareen, "DEGRADATION OF 1,2-DICHLOROETHANE WITH NANO-SCALE ZERO VALENT IRON PARTICLES" (2011). *Digitized Theses*. 3306. https://ir.lib.uwo.ca/digitizedtheses/3306

This Thesis is brought to you for free and open access by the Digitized Special Collections at Scholarship@Western. It has been accepted for inclusion in Digitized Theses by an authorized administrator of Scholarship@Western. For more information, please contact wlswadmin@uwo.ca.

DEGRADATION OF 1,2-DICHLOROETHANE WITH NANO-SCALE ZERO VALENT IRON PARTICLES

(Spine Title: Degradation of 1,2-DCA with nZVI Particles) (Thesis Format: Integrated-Article)

 $\mathbf{B} \hat{\mathbf{y}}$ the case of the d

的复数复数 医白白素

Ambareen ATISHA

1:

Graduate Program in Engineering Science

Department of Civil and Environmental Engineering

A thesis submitted in partial fulfilment of the requirements for the degree of

Master of Engineering Science

School of Graduate and Postdoctoral Studies

The University of Western Ontario

London, Ontario, Canada is accepted in particle fulfilling a contract superscription April, 2011 and of basics of filled outsting Science

© Ambareen Atisha 2011

Dr. 1.C. Serrey Saires Colline Statuta in Ara

13841<u>3月</u>14日 2011日

THE UNIVERSITY OF WESTERN ONTARIO School of Graduate and Postdoctoral Studies

The articles of the second s

CERTIFICATE OF EXAMINATION

and fighter. There we shall be the respectively and set and could

Supervisor de de sector de Examiners de souve estatate de g

distribution by all appendixed bill close

승규님께 많은 것은 안 같은 것

Dr. Jose E. Herrera de la desta de la della

els enformations en la finita de ferral aphaneta pa

Co-Supervisor

Dr. Denis M. O'Carroll

The second second

Dr. Clare Robinson and a contract of the

a anti anti anti

, ¢

10

site reactive to take the first fifthe

Dr. Jonathon Southen

Dr. Jin Zhang

The thesis by

Ambareen Atisha

Entitled:

DEGRADATION OF 1,2-DCA WITH

NANO-SCALE ZERO VALENT IRON PARTICLES

is accepted in partial fulfillment of the requirements for the degree of Master of Engineering Science

Date: <u>April 19, 2011</u>

Dr. J.Q. Shang Chair of the Thesis Examination Board

ii

ABSTRACT

The application of nanoscale zero-valent iron particles (nZVI) for abiotic remediation of chlorinated compounds is proving among the most viable technologies for environmental remediation. However, although most polychlorinated C2 compounds are easily dechlorinated by nZVI, 1,2-dichloroethane (1,2-DCA), has shown resistance to dechlorination by this nanomaterial. The present contribution shows how a combination of a catalyst and nZVI together with the addition of a hydrogen donor can be used to achieve dechlorination of 1,2-DCA under aqueous conditions similar to those found in the field. The best results for dechlorination were observed using formic acid as a H₂ donor and Pd as catalyst doped onto CMC stabilized nanoscale zero-valent iron particles-at a temperature of 45°C. This leads to significant degradation (close to 18%) of 1,2-DCA at the end of seven days. As degradation products, evolution of ethane and propane were observed from the very first day of reaction.

Keywords: nZVI, nano-scale zero valent iron, hydrogen donor, catalyst, 1,2-DCA.

collected is a set coefficient for an inclusion of and my distribution for their living support and

ACKNOWLEDGEMENTS

Above all, I thank the Almighty for giving me the strength to complete this work. Then I should thank my Supervisors Dr. Jose E. Herrera and Dr. Denis O'Carroll for their n. An an an the Charles and the state of the state continuous support and guidance during my research period at the UWO. وية. المانية: وفراري الإرام منهمة من الولوية ووياستين الولوية الإرام الارام.).... المواريعين المعرولة المراجو المتحاد Dr. Nataphan Sakulchaicharoen and Dr. Hardiljeet Boparai who had provided their expert knowledge and essential feedback throughout my work should be acknowledged. Special thanks to Ahmed Ishtiaque Amin Chowdhury for being very helpful during my studies at UWO. - 1911年1月1日 | 1911年1日 - 1911年1日 -| 1911年1日 - 1911年1日 -Moreover, I would like to thank my examiners - Dr. Clare Robinson, Dr. Jonathon Southen and Dr. Jing Zhang. a to the second state of the المراجع ويوجع المراجع المراجع والمتعام والمراجع والمراجع والمراجع والمراجع المراجع والمراجع المراجع المراجع Finally, I am really grateful to my husband and my daughter for their living support and encouragement to accomplish this endeavor. I would like to dedicate my work to my father, And the second secon Habibullah Sirajee. a sector of the 1.3.1 To rade is an I.3.490 v. The first in finite Course constitution in the 31 Else weekfahrt was erste to Decent statisfiel so white CNR, a consistent by device a so

TABLE OF CONTENTS

CERTIFICATE OF EXAMINATIONii
ABSTRACTiii
ACKNOWLEDGEMENTS
TABLE OF CONTENTS
LIST OF TABLES
LIST OF FIGURES ix
LIST OF ABBREVIATIONSxi
n an
1.0 INTRODUCTION
1.1 BACKGROUND
1.2 RESEARCH OBJECTIVES
1.3 THESIS OUTLINE
1.4 REFERENCE
2.0 LITERATURE REVIEW
•2.1 INTRODUCTION
2.2 NANO-SCALE ZERO VALENT IRON TECHNOLOGY
2.2.1 Synthesis of Iron Nanoparticles
2.2.2 Reactivity of Nanoiron with Chlorinated Compounds
2.2.3 Stabilization of Iron Nanoparticles
2.3 DEGRADATION OF 1,2-DCA
2.3.1 Degradation Pathway of 1,2-DCA
2.3.2 Degradation of 1,2-DCA with Metals
2.3.3 Degradation of 1,2-DCA with Hydrogen and a Catalyst
[•] 2.3.4 Modified Approach to Degrade 1,2-DCA with nZVI, a catalyst and Hydrogen 34

n sidden ander en	yarak rostabul r
2.4 SUMMÁRY	
2.5 REFERENCE	39
3. 0 DEGRADATION OF 1,2-DCA WITH nZVI PARTICLES	43
3.1 INTRODUCTION	43
3.2 MATERIALS AND METHODS	46
3.2.1 Chemicals	46
3.2.2 Synthesis of non-stabilized nZVI particles	
3.2.3 Synthesis of CMC-stabilized nZVI particles	50
3.3 BATCH EXPERIMENTS	
·3.4 ANALYTICAL METHODS	54
3.4.1 Solvent Extraction Analysis	
3.4.2 Headspace Analysis	55
3.4.3 Chloride Analysis	
3.5 RESULTS AND DISCUSSION	
3.5.1 The formulation for 1,2-DCA dechlorination	56
3.5.2 Formic Acid as H ₂ source	62
3.5.3 Optimization of Experimental Conditions	64
3.5.4 Degradation Pathways during dechlorination of 1,2-DCA	70
3.6 SUMMARY AND CONCLUSION	75
3.7 REFERENCE	77
4.0 SUMMARY AND RECOMMENDATIONS	
4.1 SUMMARY AND CONCLUSIONS	80
4.2 RECOMMENDATIONS	81

i.

Appendix A: Comparison of Reactivity between 1,2-DCA and different catalysts	3
Appendix B: Chromatogram from GC/MS during dechlorination of 1,2-DCA	1
Appendix C: Calibration Curves for TCE and 1,2-DCE	5
Appendix D: Calibration Curves for 1,2-DCA86	5
Appendix E: Calibration Curves for Hydrocarbons	7
Appendix F: Replicate of Experiment # 1588	8
Appendix G: Comparison of Experiments # 8 and 1189)
Appendix H: Comparison of Experiments # 10 and 1390	Ĵ
Appendix I-1: Table I-1 Experiments with 1,2-DCA	1
Appendix I-2: Table I-2 Experiments with 1,2-DCA (contd)92	2
Appendix I-3: Table I-3 Experiments with 1,2-DCA (contd)93	3.
Appendix I-4: Table I-4 Experiments with 1,2-DCA (contd)94	1
Appendix J-1: Table J-1 Experiments with TCE95	5
Appendix J-2: Table J-2 Experiments with TCE (contd)96	5
Appendix J-3: Table J-3 Experiments with TCE (contd)97	7
Vita98	3

LIST OF TABLES

Table 2.2 Redox Table Image: Comparison of the second se	29
Table 2.3 Half Reactions Gibb's Free Energies of Reductive Dechlorination	30
Table 2.4 Relative Rates for Vicinal Dechlorination of Chlorinated Ethanes in Reducin Sediment Water Slurry	ıg 31
Table 2.5 Summary of Catalytic Hydrodechlorination of 1,2-DCA	33
Table 3.1 Experimental Conditions for 1,2-DCA and TCE Dechlorination	53 N
Gegere D. F. T. G. Marger, et R. M. Efreger, Henry and the state of the state for the state of the state of the	in the second se
ningen anderen versteren en er som en som en som en som er so Here anderen som er som er Here anderen som er s	
englinge Sule (EX.)	840.
Pirjana Balt Fiscala S. J., a theorida doo to docatafadon of Not beyde of SCH scale, als geb all'M and D Scale Scale (1910)	
General States of the state of the state of the state of the states of t	1980) 20
gipus 1.5 Benedera e el 10 se el serverte angle 10 sel serverte angle 12 sel serverte angle 12 serverte angle 1 25 se	1 50 6
Higher (1993) an an an ann an an ann an Anna a An an Anna an A	
en en en de la sector de la completa de la complet El sector de la completa de la comple	的道

viii

LIST OF FIGURES

Figure 2.1: Core-shell Structure of nZVI
Figure 2.2 Pathways of TCE dechlorination with Fe ⁰ 16
Figure 2.3 Schematic of Reactivity of Pd/Fe Nanoparticles with Chlorinated Solvents 19
Figure 2.4 1,2-DCA Transformation Pathways under different Conditions
Figure 2.5 Reaction Scheme for Catalytic Hydrodechlorination of 1,2-DCA
Figure 2.6 Schematic of Proposed Hydrogen Transfer Mechanism on Pd
Figure 2.7 Possible Mechanism of Pd-Catalyzed Hydrodechlorination in the presence of formic acid a) Hydride Mechanism b) Radical Mechanism
Figure 3.1 TEM image of Pd/Fe bimetallic particles with scale bar 100 nm
Figure 3.2 TEM image of CMC (90K) stabilized Pd/Fe bimetallic particles with scale bar 5 nm
Figure 3.3: Degradation of 100 mg/L of TCE with 1.5 g/L of nZVI and 0.1wt% Pd at 25°C.
Figure 3.4: Evolution of Chloride due to degradation of 100 mg/L of TCE with 1.5 g/L of nZVI and 0.1wt% Pd at 25°C
Figure 3.5: Degradation of 5 mg/L of 1,2-DCA with 20 g/L of nZVI and 0.2wt% Pd at 25°C.
Figure 3.6: Degradation of 10 mg/L of 1,2-DCA with 20 g/L of nZVI and 0.2wt% Pd at 25°C
Figure 3.7: Degradation of 5 mg/L of 1,2-DCA with 2.5 g/L of nZVI and 0.2wt% Pd in 250 mg/L HCOOH at 25°C
Figure 3.8: Degradation of 10 mg/L of 1,2-DCA with 5 g/L of nZVI and 0.2wt% Pd in 500 mg/L HCOOH at 35°C

	Figure 3.9: Degradation of 10 mg/L of 1,2-DCA with 10 g/L of nZVI and 0.2wt% Pd, 0.2 wt% Rh and 0.1 wt% Pt in 500 mg/L HCOOH at 35°C.	
	Figure 3.10: Degradation of 10 mg/L of 1,2-DCA with 1.5% CMC stabilized 5 g/L of nZVI and 0.5wt% Pd in 500 mg/L HCOOH at 25°C.	
	Figure 3.11: Degradation of 10 mg/L of 1,2-DCA with 1.5% CMC stabilized 5 g/L of nZVI and 0.5wt% Pd in 500 mg/L HCOOH at 45°C.	
	Figure 3.12: Dechlorination of 1,2-DCA by Vanstone et al (2008)	
	Figure 3.13: Proposed 1,2-DCA Transformation Pathway while reacting with bimetallic Pd/Fe and HCOOH	
	CAME EXCLOSES CONTRACTOR Address COMPLETE EXCLOSES CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR COMPLETE EXCLOSED CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR COMPLETE EXCLOSED CONTRACTOR C	
	 State of the Annual State Performance of the Annual State of the Annual State Annual State of the Annual State o	•
۹		
	NY ANALANA ANALANA MANANA AMIN'NA AMIN' NY TAONA MANANA MIN'NA AMIN'NA A Amin'ny taona manana amin'ny taona amin'ny taona amin'ny taona amin'ny taona amin'ny taona amin'ny taona amin'ny	jođ -
	FAA XCA	
	af yataragi a bala da bala da ang ang ang ang ang ang ang ang ang an	

х

LIST OF ABBREVIATIONS

14

.с. С

C.

N. ANARS

1,1,1,2-TeCA	: 1,1,1,2-tetrachloroethane
1,1,2,2-TeCA	: 1,1,2,2-tetrachloroethane
1,1,2-TCA	: 1,1,2-trichloroethane
1,1-DCA	: 1,1-dichloroethane
1,1-DCE	: 1,1-dichloroethylene
1,2-DCA	: 1,2-dichloroethane
A de la composition de la comp	: Silver
cis-DCE	: cis-dichloroethylene
СМС	: Carboxymethyl Cellulose
COCs	: Chlorinated Organic Compounds
DI	: De-ionized Water
E ₀	: Standard Reduction Potential
HCA	: Hexachloroethane
НСООН	: Formic Acid
IPA	: Isopropanol
Ni	: Nickel
NZVI	: Nano-scale Zero Valent Iron
PAA	: Poly Acrylic Acids
PCA	: Pentachloroethane
PCBs	: Polychlorinated biphenyls
PCE	: Tetrachloroethylene
Pd	: Palladium

xi

 $\langle \psi \rangle$

PRBs	: Permeable Reactive Barriers (1) 2014
Pt (1), a set estated.	: Platinum
PVP	: Polyvinylpyrrolidone
RDX	: Hexahydro-1,3,5-trinitro-1,3,5-triazine
Rh	: Rhodium
Ru	: Ruthenium
SDBS	: Sodium Dodecylbenzene Sulfonate
TCE	a din dia sense ang dia mgi na panahasin' na silangan si nakyi pana maka ang di : Trichloroethylene ang dia
The second secon	· · · · · · · · · · · · · · · · · · ·
trans-DCE	: trans-dichloroethylene
vc ^{arsi dalara}	: Vinyl Chloride
ZVI	: Zero Valent Iron
	a bear proved to a local provide the state of providence the most
	de un beinde maar nijndeel oor naar heline beschien planker politikere be
	e na na statistica e a service de la Caladi State de Bardig de La
	or Same real Colored (1997), para l'arce a ch'Anneag, 1992), files e chanairtís.
ing worth and st	

pales annaech d' a malamhan ar tri a ché einste a suiscrite la fiscardo-suisea a tradición de Alia. Ascalage manages, est un líter, es la sue o trago assisti stalager ar un la situé commert

(Assubstate of provide college cardinal both to dirpy waits the book as the black of the moment for a barge both state of the state of the college of the state of the stat

<u>CHAPTER 1</u>

14

a a the second a first on the first design and

ser 1.0 INTRODUCTION destance of Plane

1.1 BACKGROUND

Groundwater contamination problems caused by halogenated organic compounds, heavy metals and energetic materials such as hexahydro-1,3,5-trinitro-1,3,5-triazine (RDX) and 2,4,6-trinitrotoluene (TNT) are increasing in Canada. This is mainly due to the large number of toxic compounds utilized in agriculture and industry (Zhang et al, 1998; Environment Canada, Groundwater). Among these, chlorinated ethylenes such as tetrachloroethylene (PCE), trichloroethylene (TCE), trans-dichloroethylene (t-DCE), cisdichloroethylene (c-DCE), 1,1-dichloroethylene (1,1-DCE), vinyl chloride (VC) as well as chlorinated ethanes such as hexachloroethane (HCA), 1,1,2,2-tetrachloroethane (1,1,2,2-TeCA), 1,1,1,2-tetrachloroethane (1,1,1,2-TeCA), 1,1,2-trichloroethane (1,1,2-TCA), 1,1-dichloroethane (1,1-DCA) and 1,2-dichloroethane (1,2-DCA) are the most prevalent contaminants in soils and aquifers and have been listed as priority pollutants by both the US Environmental Protection Agency and the Superfund National Priority List (Zhang et al, 1998; Song and Carraway, 2005; Lien and Zhang, 2005). These chemicals are used in dry cleaning, wood preservation, pesticide manufacturing, metal cleaning, and paint removal. Accumulation of these chlorinated solvents in the sub-surface results from leaking underground or above-ground storage tanks, storage areas or buried chemical distribution pipelines, spillage at chemical loading and off-loading facilities or even from intentional disposal into the sub-surface. Once these solvents are spilled in aquifers in sufficient quantities, they have the capacity to penetrate below the water table and thus serve as a long-term source of contamination. The length of time required to dissolve a solvent source zone can be hundreds of years under natural conditions. When groundwater is used as a source of drinking water, the presence of these compounds becomes a serious risk to human health. These risks include liver or kidney damage and spontaneous abortions; moreover, some of these compounds are considered carcinogenic. Therefore, a great deal of time and effort has been put into the development of remediation techniques to treat contaminated sites (Geiger et al, 2009).

the pipe in a granifing near of here of here is a submy file or quine bookers with the strate

Depending on the type and location of the contaminants, both in-situ and ex-situ remediation options are being used for their removal from soil and groundwater. Electroreclamation, enhanced volatilization, vitrification, bioventing, biostimulation and natural attenuation are some of the commonly used in-situ treatment technologies, whereas, chemical extraction, dehalogenation, redox reactions, incineration and pyrolysis can be classified as ex-situ treatment technologies (Lodolo, 2007). However, the application of some of these techniques is hampered due to specific drawbacks associated with their implementation such as- high costs, incomplete decontamination resulting in formation of more toxic intermediates, slow outputs and complex process control (Lodolo, 2007).

One of the most promising technologies for the remediation of subsurface contamination involves the use of zero-valent metallic materials (Zhang et al, 1998; Wang et al, 2008; Sakulchaichareon et al, 2010), specifically zero-valent iron. In this case, groundwater flows through an iron rich reactive zone, resulting in chemical transformations of the contaminants into benign compounds (Wang and Zhang, 1997; Zhang et al, 1998; Schrick et al, 2002). Equation 1.1 can be used to express the reactions between

. The surge small discovered at the strength of a surge subject mode. Fight strethod as many server di

chlorinated organic compounds and iron in aqueous solution. Here, iron acts as an electron donor (reductant) to remove chlorine. $C_xH_yCl_z + zH^+ + zFe^0 \rightarrow C_xH_{y+z} + zFe^{2+} + zCl^-$ (Equation 1.1)

the set of the state of the set of the set of the state of the

Despite the promising use of iron fillings in permeable reactive barriers (PBRs), there still exist some challenges. For instance, the lower reactivity of zero-valent iron powder with lightly chlorinated compounds may produce and accumulate by-products such as, 1,2-dichloroethylene and vinyl chloride which are of even more toxicological concern than their parent compounds (Zhang et al, 1998; Wang et al, 2008). To address the above-mentioned shortcomings researchers have proposed nanoscale zerovalent iron (nZVI) as an alternative. (Zhang et al, 1998; Schrick et al, 2004, Li et al, 2006). Indeed, the use of nZVI for the treatment of sub-surface chlorinated hydrocarbons is of significant interest due to its low toxicity and significant contaminant degradation capabilities (Zhang et al, 1998; Schrick et al, 2005; Sakulchaicharoen et al, 2010)

The very small dimension, high surface area to volume ratio, high surface energy as well as high surface reactivity is the reason behind the high reactivity of nZVI particles with environmental contaminants (Zhang et al, 1998; Zhang, 2003). As opposed to PBR technologies, instead of installing metal walls, nanoparticles, in the form of water slurry, can be injected into the contaminated plume requiring treatment by means of pressure or gravity or both. Their nanoscale size makes it possible to transport them to a certain extent by groundwater flow (Zhang, 2003; Li et al, 2006). Furthermore, the addition of

promoters in the form of catalytic metals such as palladium (Pd), platinum (Pt), silver (Ag) or nickel (Ni) can lead to accelerated dechlorination rates and therefore, lead to the formation of benign hydrocarbons (Lien and Zhang, 2001; He and Zhao, 2005) rather than undesirable chlorinated byproducts. Nevertheless, these doped Fe nanoparticles tend to react rapidly with dissolved oxygen or even water, due to their extremely high reactivity, leading to quick loss in reactivity (He and Zhao, 2005). Another consideration is magnetic and van der Waals attractive forces between nZVI particles which cause agglomeration and the formation of larger particles or flocs (Sakulchaicharoen et al, 2010; He and Zhao, 2007). This agglomeration decreases available nZVI surface area for reaction (Sakulchaicharoen et al, 2010; He and Zhao, 2007). To prevent agglomeration, stabilizers such as Carboxymethyl Cellulose (CMC), polyvinylpyrrolidone (PVP) and guar gum can be added with iron nanoparticles. The addition of these stabilizers enhances dispersion of particles through electrostatic repulsion and/or steric hindrance (Sakulchaicharoen et al, 2010). Particle size reduction and the resulting gain in surface area leads to more reactive nanoparticles for the degradation of chlorinated solvents.

Despite the successful application of nZVI particles for remediation of chlorinated hydrocarbons, nitroaromatics and polychlorinated biphenyls, 1,2-dichloroethane (1,2-DCA) is found so far to be totally resistant to degradation using nZVI (Zhang et al, 1998; Song and Carraway, 2005; De Wildeman and Verstraete, 2003; Kopinke et al, 2004). 1,2-DCA is a chlorinated hydrocarbon that affects human liver and kidneys, neurological, cardiovascular, and immune systems (De Wildeman and Verstraete, 2003). Metal-based reduction has been found to be marginally successful in 1,2 DCA chemical

4

iç,

dechlorination whereas catalytic hydrodechlorination and bioremediation are reported to be successful. However, the high temperature and molecular H₂ requirements make catalytic hydrodechlorination inconvenient for field application, whereas the very slow anaerobic bioremediation process is unsuited for the rapid dechlorination rate usually required in field applications (Ukisu et al, 1998; Klecka et al, 1998). An alternative and effective approach for successful dechlorination of 1,2-DCA at field scale is therefore required. Though to date iron has been unsuccessful in dechlorinating 1,2 DCA (Larson and Weber; De Wildeman and Verstraete, 2003), the process is thermodynamically possible. This makes the remediation of 1,2-DCA by nZVI particles through the application of an appropriate methodology of huge interest.

1.2 RESEARCH OBJECTIVES

The goal of this research work is to find a nZVI-based formulation able to degrade 1,2dichloroethane (1,2-DCA) and to optimize degradation conditions. As presented in the following chapters 1,2-DCA dechlorination is a significant challenge for nZVI based remediation technologies. Optimization of the experimental conditions and investigation of the degradation pathways are the two other main objectives of this work.

areas and a finance of a supplication of the stating and decomposite the sub-

1.3 THESIS OUTLINE

This thesis is written in "Integrated Article Format". A brief description of each chapter is presented below.

Chapter 1 introduces some general background information and states the objectives of the study, as marked in the second of the source terminary state and entry the backage of C2 nderen er benen ersenen för eksende er andere förde för skande som etter att som er som er som er som er som e Regenster att som er som er

and shall we black from the Chapter 2 reviews previous research work focused on the degradation of 1,2-DCA. In radar et er e neer dier die ste addition, nanoparticle structure, synthesis and application for remediation of chlorinated organic compounds are also discussed. The Problem occupied as with CMC Bartrateneous 经总计划运输通过 法公共公共通过 网络公共规范 鞋囊

a de la construction de Chapter 3 presents the experimental methodology as well as the results and discussion of the experiments carried on this research work. The address Green Electrony in Antonio States, by the address

1.4440. , 21월 22일 - 21일 Chapter 4 summarizes the research conducted in this study, and presents conclusions and en an fear stat is des seises a say listenations recommendations for future work. t Added Markey de Bits Beere aa

dere dere de Las de We show the she when we wanted as a second of N. E.L. ling the state of the second state of the st 后接到 自然的 计 计算法转移 化化学学 化化学学 化化学学 化化学学 化化学学 化化学学

the following to a solution of a particles belowing to a gala the

the first state of the and the physical devices the many e per communication de la construction de la construction de la secondad de la constituier de la constituier de

schedelsche Erzens Verschere, 1994 (1994) beite bereiten der Mittelscheren, "Preise es bereitels active and a state the ball of default of the 1991

deven in det fram och fra her det i blad och det middelige et er filmen hen an de ferreten verste die eine het skere enner verste film 🖓

6

Billio disanta Andra Indra A

Nacional Approximation of the State of the S

1.4 REFERENCES

De Wildeman, S., W. Verstraete, "The quest for microbial reductive dechlorination of C2 to C4 chloroalkanes is warranted". <u>Appl Microbiol Biotechnol</u>, 2003, 61, 94-102.

1990 a. 1997

Geiger, Cherie L, Kathleen M. Carvalho-Knighton, "Environmental Applications of Nanoscale and Microscale Reactive Metal Particles" <u>ACS Symposium Series</u>

1027, Americal Chemical Society 2009.

He, Heng, Dongye Zhao, "Preparation and characterization of of a new class of starchstabilized bimetallic nanoparticles for degradation of chlorinated hydrocarbons in water". Environmental Science and Technology, 2005, 39, 3314-3320.

- He, Heng, Dongye Zhao, "Stabilization of Fe-Pd nanoparticles with CMC for enhanced transport and dechlorination of TCE in soil and groundwater". Ind. Eng. Chem. Res, 2007, 46, 29-34.
- Klecka, G.M, C.L. Carpenter, S.J.Gonsior, "Biological transformations of 1,2-DCA in sub-surface soils and groundwater" Journal of Contaminant Hydrology, 1998, 34, 139 154.
- Kopinke Frank-Dieter, Katrin Mackenzie, Koechler, Robert Georgi Anette, "Alternative sources of H2 for hydrodechlorination of chlorinated organic compounds in water on Pd catalysts". <u>Applied Catalysis A: General, 2004, 271, 119-128</u>.
- Lodolo, Andrea, "Further description: remediation option". <u>EUGRIS: Portal for Soil and Water</u> <u>Management in Europe, 2007.</u>
- Li, L., M.H. Fan, R.C. Brown, J.H. Van Leeuwen, J.J. Wang, W.H. Wang, Y.H. Song, and P.Y. Zhang, "Synthesis, properties, and environmental applications of nanoscale iron-based materials: A review". <u>Critical Reviews in Environmental</u> <u>Science and Technology</u>, 2006. 36, 405-431.
- Li, X.Q., D.W. Elliott, and W.X. Zhang, "Zero-valent iron nanoparticles for abatement of environmental pollutants: Materials and engineering aspects." <u>Critical Reviews in</u> Solid State and Materials Sciences, 2006. 31, 111-122.
- Lien, Hsing-Lung, Wei-Xian Zhang, "Hydrodechlorination of chlorinated ethanes by nanoscale Pd/Fe bimetallic particles". Journal of Environmental Engineering, 2005, 131.
- Lien, Hsing-Lung, Wei-xian Zhang, "Nanoscale iron particles for complete reduction of chlorinated ethenes". <u>Colloids and Surfaces A: Physiochemical and Engineering</u> <u>Aspects</u> 2001, 191, 97-105
- Ponder, S.M., J.G. Darab, J. Bucher, D. Caulder, I. Craig, L. Davis, N. Edelstein, W. Lukens, H. Nitsche, L.F. Rao, D.K. Shuh, and T.E. Mallouk, "Surface chemistry and electrochemistry of supported zerovalent iron nanoparticles in the remediation of aqueous metal contaminants". <u>Chemistry of Materials</u>, 2001, 13, 479-486.
- Sakulchaicharoen, Nataphan, D.M.O'Carroll and Jose .E. Herrera, "Enhanced Stability and dechlorination activity of pre-synthesis stabilized nanoscale FePd particles". Journal of Contaminant Hydrology. 2010, 118, 117-127.
- Schrick, B., J.L. Blough, A.D. Jones, and T.E. Mallouk, "Hydrodechlorination of trichloroethylene to hydrocarbons using bimetallic nickel-iron nanoparticles". Chemistry of Materials, 2002, 14, 5140-5147.

- Song, Hocheol, Elizabeth R Carraway, "Reduction of chlorinated ethanes by nano-scale zero-valent iron: kinetics, pathways and effects of reaction conditions". <u>Environmental Science and Technology</u>, 2005, 39, 6237-6245.
- Ukisu, Yuji, Satoshi Kameoka, Tatsuo Miyadera, "Rh-based catalysts for catalytic dechlorination of aromatic chloride at ambient temperature" <u>Applied Catalysis B:</u> <u>Environmental</u>, 1998, 18, 273-279.
- Wang, C.B. and W.X. Zhang, "Synthesizing nanoscale iron particles for rapid and complete dechlorination of TCE and PCBs". <u>Environmental Science & Technology</u>, 1997, 31, 2154-2156.
- Zhang, W.X., "Nanoscale iron particles for environmental remediation: An overview". Journal of Nanoparticle Research, 2003, 5, 323-332.

Zhang, W.X., C.B. Wang, and H.L. Lien, "Treatment of chlorinated organic contaminants with nanoscale bimetallic particles". <u>Catalysis Today</u>, 1998, 40, 387-395.

. The marel is a shared a larger barrier of a lower sole, the eleganetic processing and some a signal sole of a 🕺

terminally, and a target of a set to a star with speer integration, with enclose and

searche monant and average the horizon of the horizon have been used for grave being

sanaa sa da walay na kawanda da baban na sadi da **da da da da ka in** ang ina na

you and in first the assessment of the first state of the sufficiency development have made down have

interactive disclote (1947). Biologic of the product order order discrimination is effective

and a contraction of the second of the second state of the second s

there align the following and the considerations of the area and in the

2024). He setted that has a shift on show procedure that the tweeter hubble drawberher

Cananda dault and the est (Monte et al. 2019). Distances which terrained the

vare attantion of the lost on Concellent information store plyake mentile of the polynamize bi

believed tool of discriminance entries between a manifestation for monthing by a calle

Leonited a sub-state for the second management to be been been as the second factors of the second factors of the

and all the data when the state of the of annual state from all and the

网络马达 机动力

2.0 LITERATURE REVIEW

2.1 INTRODUCTION

osterani.

Chlorinated Organic Compounds (COCs) have caused widespread environmental contamination in both groundwater and soil since they are persistent in the environment over relatively long time periods and are difficult to directly degrade under most natural or abiotic conditions. Consequently, the environmental engineering community has devoted significant efforts to the degradation of these chlorinated organic compounds. Both in-situ and ex-situ approaches comprising of thermal, biological, physical and chemical treatments are commonly used to remove these contaminants (Wang et al, 2008).

Thermal treatments utilize high temperature to volatilize, burn and degrade contaminants. Specifically, in-situ thermal treatments such as soil vapour extraction, vitrification and ex-situ thermal treatments like incineration and pyrolysis have been used for remediation purposes. Nevertheless, incomplete combustion giving rise to dioxins and furans and high cost and fugitive emissions such as dust and particulates often make their application less attractive (Lodolo, 2007). Biological treatment requires optimal conditions and selective microorganisms to decompose COCs by metabolic processes (Wang et al, 2008). Bioventing, biostimulation, natural attenuation are some of the proposed in-situ biological treatments, whereas, contaminants are treated ex-situ in bioreactors (Lodolo, 2007). However, this is a relatively slow procedure and COCs often inhibit degradation through their toxic effects (Wang et al, 2008). Physio-chemical treatments are a combination of physical and chemical treatments. Here phase transfer of the pollutants is induced and chemical structure is altered by chemical reactions to produce less toxic

compounds (Lodolo, 2007; Wang et al, 2009). Electroreclamation and enhanced volatilization are two examples of in-situ physio-chemical treatments. Although these technologies have been widely used for volatile organic compound degradation, heterogeneities in the treated zone and high concentration of contaminants heavily influence their activity (Lodolo, 2007). On the other hand, chemical extraction, dehalogenation and redox reactions have been proposed as ex-situ treatments. Chemical extraction is less effective for high molecular weight compounds and incomplete decontamination may form more toxic compounds (Lodolo, 2007). Therefore, a more convenient technology is required that can be used to remediate the contaminant in-situ with the maximum benefits.

Recently, the degradation of COCs by direct reduction using zero-valent iron (ZVI) technology has received great interest (Zhang et al, 1998; Wang et al 2008; Sakulchaichareon et al, 2010). Microscale zero-valent iron, usually applied in permeable reactive barriers (PRBs), reacts with contaminants like chlorinated aliphatics, aromatics and polychlorinated biphenyls to produce mostly benign compounds: hydrocarbons, chloride and water (Zhang et al, 1998; Schrick et al, 2002). However, the implementation of zero-valent metal technology in the sub-surface still faces various challenges, such as the low reactivity of iron particles towards less chlorinated hydrocarbons, in turn leading to the production of chlorinated by-products, which are sometimes even more toxic than the parent compounds. Moreover, the time required to complete the dechlorination is often very long (Wang et al, 2008). Furthermore, the construction of walls in aquifers at depths more than 30m poses engineering difficulties.

Current research indicates that nano-scale zero-valent iron (nZVI) can be a very good alternative to PRBs (Wang and Zhang, 1997; Zhang et al, 1998; Nurmi et al, 2005). Their very tiny dimensions allow them to be injected in the form of water slurry into the contaminated plume requiring treatment. The injection is done through pressure, gravity, or both. The nZVI-water slurry reacts more rapidly to dechlorinate organic contaminants in the sub-surface than the larger ZVI particles (Li et al, 2006). The ability to deliver nanoparticles directly to the contaminated zone in the sub-surface, even in areas where many conventional methods are inaccessible (such as beneath buildings and airport runways) make nZVI a very attractive remediation technology (Elliot and Zhang, 2001).

2.2 NANOSCALE ZERO-VALENT IRON TECHNOLOGY

The structure of iron nanoparticles is important for their environmental remediation function. These nanoparticles are generally spherical in shape, colloidal in nature and have a strong tendency to aggregate (Li et al, 2006) (Fig 2.1). The very small dimensions (~1-100 nm), relatively high specific surface area (~30 m²/g) and greater density of reactive sites are responsible for the high intrinsic reactivity of nZVI particles (Zhang et al, 1998; Elliot and Zhang, 2001; Nurmi et al, 2005). Under aqueous conditions, iron nanoparticles contain a Fe⁰ core, which is a moderate reducing agent and is slowly oxidized in air or water (equations 2.1, 2.2) (Li et al, 2006). This core is surrounded by a shell of iron oxides, hydroxides or oxyhydroxides (Eqns 2.3, 2.4, 2.5) (Fig 2.1) with a shell thickness approximately 5 nm (Liu et al, 2005a). It has been reported that oxidation of the Fe⁰ core causes transportation of electrons through the iron oxide shell that in turn

and a start start was the for a start for the start start and the start start start was

reduces organic contaminants (Yan et al, 2010). The mixed valence iron oxide shell protects the Fe^0 core from further oxidation (Li et al, 2006).

here and the first appendix the set of while the first the first the



Fig 2.1- Core-Shell Structure of nZVI $Fe^{0} + 2H_{2}O \rightarrow Fe^{2+} + H_{2} + 2OH^{-}$ (Equation 2.1) $2Fe^{0} + O_{2} + 2H_{2}O \rightarrow 2Fe^{2+} + 4OH^{-}$ (Equation 2.2) $4Fe^{2+} + 4H^{+} + O_{2} \rightarrow 4Fe^{3+} + 2H_{2}O$ (Equation 2.3)

 $Fe^{3+} + 3OH^- \rightarrow Fe(OH)_3$ (Equation 2.4)

 $Fe^{3+} + 2H_2O \rightarrow FeOOH + 3H^+$ (Equation 2.5)

2.2.1 Synthesis of Iron Nanoparticles

Borohydride reduction of an aqueous iron salt is the most commonly used method to produce nZVI in research labs (Li et al, 2006; Zhang and Elliot, 2006). This is a very

An in the state of the second s

simple method, which can be carried out in aqueous phase at room temperature without the need for highly specialized instruments. Moreover, this method requires only two common reagents, i.e. iron precursor and sodium borohydride (NaBH₄). Many studies have utilized aqueous solutions of ferric chloride (FeCl₃.6H₂O) and ferrous sulphate (FeSO₄.7H₂O) as precursors to iron nanoparticles (Lien and Zhang, 2001; Zhang et al, 2003; Li et al, 2006). For instance, Lien and Zhang (2001) synthesized nZVI by adding 1:1 volume ratio of 0.25M NaBH₄ into 0.045M FeCl₃.6H₂O (Equation 2.6).

 $4Fe^{3+} + 3BH_4 + 9H_2O \rightarrow 4Fe^0 + 3H_2BO_3 + 12H^+ + 6H_2$ (Equation 2.6)

mentering of the second of the contract of the contract of the second second second second second second second

and such the first dealers and the second second second second for the

The excess amount of borohydride accelerates the synthesis reaction as well as ensures uniform growth of nanoiron (Zhang et al, 2003). Although ferric chloride hexahydrate is often used as the iron precursor, ferrous sulphate has been proposed as an alternative due to the potential health and safety concerns associated with handling the highly hygroscopic and acidic ferric chloride salt. Zhang and Elliot (2006) prepared nZVI from 0.28M FeSO₄.7H₂O by reducing it with 0.5M of NaBH₄ solution (equation 2.7)

 $2Fe^{2+} + BH_4^+ + 3H_2O \rightarrow 2Fe^0 + H_2BO_3^+ + 4H^+ + 2H_2$ (Equation 2.7)

nere in a vinere se proving the angle in cash and the . This of Bearle's presences

Less borohydride is required to reduce ferrous sulphate (stoichiometric excess of 3.6) than the ferric chloride (stoichiometric excess of 7.4) (Zhang and Elliot 2006; He and Zhao, 2007). Moreover, the use of sulphate allows for an easier monitoring of chloride

production during dechlorination reactions of chlorinated organic compound (Zhang and Elliot 2006, He et al, 2007).

2.2.2 Reactivity of Nano Iron with Chlorinated Compounds

 $\{r_{i}, r_{i}\}$

The contaminants that can be treated with nZVI particles include chlorinated ethylenes, methanes, ethanes, nitroaromatics, polychlorinated biphenyls, chlorophenols, heavy metals and inorganic anions (Zhang et al, 1998; Liu et al, 2005a; Liu et al, 2005b; Song and Carraway, 2008). The dechlorination of halogenated compounds by nZVI involves several steps: mass transport of the contaminant from the solution to the surface of nanoparticles, the adsorption of the contaminant to the reactive surface of nanoparticles, reactions at the surface, desorption of products from the surface and finally mass transport of products to the solution (Matheson and Tratnyek, 1994; Lien and Zhang, 2007). The slowest reaction step requires the highest activation energy and controls the kinetics of a reaction (Lien and Zhang, 2007).

When Fe^0 is oxidized to Fe^{2+} (equation 2.8), it donates two electrons and these can be used in a variety of reactions to transform contaminants. The addition of two electrons from the reduction of nZVI to a chlorinated ethylene may result in the release of chloride ion with the concurrent formation of a new C-H bond (equation 2.9) (Bylaska et al, 2008). These reactions can follow two different pathways: β -elimination (equation 2.10) and hydrogenolysis (equation 2.11). It has been proposed, based on mechanistic studies, that the electron transfer to the surface adsorbed molecule is the rate-limiting step (Schrick et al, 2002; Bylaska et al, 2008).





and sense works with his alternary due the masses of a sense of different que which a first had not

the Addition of the State of the Contract of Marcal Markets States (2018) (1) the

We believe they have the DCD Discondinal ${f Q}{f I}$ and indefinition ${f H}^{(1)}$ in fingtower biomic to the mouth they have been by β is detailed β -elimination dense c = c' , baseline is Hydrogenolysis with β . The i de **él** de la del **'Cl**ére hadre d'Ale de reger de la 1996 de la un de par en el la constant de la debarre de la constant de la const ada sebara da la constructiva inservise del constructiva del constructiva del constructiva del constructiva del which a firm particular product graph with the other there exists to be press $\hat{\mathbf{C}}$ == $\hat{\mathbf{C}}$ Cl→→C==C→→ Harrow (Gala Arres, 17 Notes, Physical results of the train H the process Chloroacetylene is a set of the edit of the constraint deputy deputy deputy of cis-1,2-DCE is Hydrogenolysis Hydrogenolysis production of $[e^{-1}]$ about the first of the constitution of \mathbf{Cl}_{i} is $[\mathbf{Cl}_{i}]$ in $[\mathbf{V}]$ and $[\mathbf{H}]$ devia la \mathbb{J} l'étér de la desir contes, per el ci, italiès, l'emp $\mathbf{\hat{c}} = \mathbf{\hat{c}}$ esse $\mathrm{Prior}(\mathbf{H} \longrightarrow \mathbf{C} = \mathbf{C} \longrightarrow \mathbf{H}$ is all the second constants of \mathcal{H} such that \mathbf{H} H intende la Acetylene « separation in alman en actuare a faure, section **Vinyl Chloride** al All model by Angel Strates and the strategy of the state of the Hydrogenolysis of the states. takaget and post-menution blocking to the shakes, a cashe chinabalish broke converters 物质 建硫酸盐医丁胆酸 医马赫氏管 Cale denses where γ , where the solution of r , defined by performing \mathbf{H}'_{r} . At H. ter due selection de la la definita de la la la consideración de la la construction de la la construction de la Fig: 2.2 Pathways of TCE dechlorination with Fe⁰ and the address of the relations of the Laboratory research has showed that the transformation of chlorinated ethylenes, such astetrachloroethylene (PCE), trichloroethylene (TCE), cis- dichloroethylene (c-DCE), trans-

dichloroethylene (t-DCE), and vinyl chloride (VC) using nanoscale iron particles lead to fully dechlorinated products (e.g.ethane) with reactive intermediates like c-DCE, VC, ethylene for a short period of time. Liu et al (2005b) studied the reactivity of nZVI with TCE at high and low iron to TCE ratios. In both cases, only monometallic iron was used. At low iron/TCE ratios (TCE = 290 mg/L and $Fe^0 = 0.035$ g/L), complete dechlorination took a longer period (several days). Ethane (70%) and C3-C6 coupling products (30%) were identified as end products (Liu et al, 2005b). Ethylene was found as a reactive intermediate that later transformed into ethane, whereas no acetylene was detected. At high iron (monometallic) to TCE ratios (TCE = 4.4 mg/L and Fe⁰ = 0.194 g/L) the end products were mainly ethane (80%) and C3-C6 coupling products (20%) and took only \sim 1.5 hours to complete the dechlorination (Liu et al, 2005b). Zhang et al (1998) investigated the dechlorination of chlorinated ethylenes (20 mg/L) with nZVI (20g/L) and found that TCE was completely dechlorinated within 1 hour, while PCE, cis-DCE and VC took 2, 3 and 3 hours respectively. Ethylene, ethane, propylene, propane, butylene, butane and pentane were identified as end products; organic chlorinated by-products were not observed (Zhang et al, 1998). At a this sector of the other of the other sector detector and

Chlorinated ethanes, such as hexachloroethane (HCA), pentachloroethane (PCA), 1,1,2,2tetrachloroethane (1,1,2,2-TeCA), 1,1,1,2-tetrachloroethane (1,1,1,2-TeCA) 1,1,1trichloroethane (1,1,1-TCA) and 1,1-dichloroethane (1,1-DCA) along with chlorinated ethylenes are another major group of groundwater contaminants with low reduction potentials (~ -0.68 to + 0.15 V) (Song and Carraway, 2005). Their structural stability makes them persistent in the environment. The reactions of chlorinated ethanes with

velenere, de die jesteleneren en die eentide meiden in die endere het.

nZVI particles form many chlorinated by-products and generally the reactivity rate is slower than those of chlorinated ethylenes (Lien and Zhang, 2005). The degree of chlorination of the chlorinated ethane controls their reactivity with nZVI particles (Lien and Zhang, 2005). When there are more than four chlorine atoms the reaction rates are high whereas, lesser chlorinated ethanes with chlorine number less than two show little reactivity (Lien and Zhang, 2005). Reactions of chlorinated ethanes with conventional iron powders produce mainly chlorinated by-products and no ethane (Lien and Zhang, 2001). According to Song and Carraway (2005), HCA was one of the most reactive chlorinated ethanes and when it reacted with 0.08g/L nZVI, it formed PCE at the end of 1.4 hours. Song and Carraway (2005) also showed that 1,1,1,2-TeCA transformed into 1,1-DCE and then ethane after an 80 hours reaction with 0.08 g/L of nanoiron. In case of 1,1,2-TCA the same amount of nZVI resulted in a much slower rate but also formed ethane at the end of 110 hours (Song and Carraway, 2005).

During Fe^0 corrosion in aqueous system, H_2 is generated through water reduction (equation 2.1). The dissociative chemisorption of this H_2 generates a new activated reductant, atomic hydrogen, on the nanoparticle surface in the presence of a catalyst, which is responsible for bimetallic reactivity (Schrick et al, 2002). In a bimetallic nZVI system, Fe^0 acts as the electron donor and the catalyst functions as hydrogen dissociator (Song and Carraway, 2008). This hydrogen initiates rapid dehalogenation of the contaminant forming benign hydrocarbons and inorganic chlorides (Fig 2.3). However, in the absence of a catalyst, hydrogen cannot contribute directly to the dechlorination; rather the presence of an excess amount of hydrogen on the metal surface may inhibit the iron corrosion (Matheson and Tratnyek, 1994). The large amount of hydrogen gas produced as minute gas bubbles might perform as barriers inhibiting the contact of iron particles and target pollutants (Wang et al, 2009). Therefore, the catalytic effects of a second metal through a direct hydrogen reduction (Schrick et al, 2002) and a galvanic corrosion leading to the increased corrosion rates (Zhang et al, 1998) are the main reasons behind the enhanced reactivity of bimetallic nZVI particles. The use of catalysts with nZVI reduces the activation energy required for hydrogen dissociation over the catalyst surface (Lien and Zhang, 2007). The lower activation energy for bimetallic nZVI particles indicates the dechlorination reaction to be catalytic, resulting in an increase in the dechlorination rate and decreasing the production of toxic intermediates such as dichloroethylenes and vinyl chloride (Song and Carraway, 2008; Yan et al, 2010).



er al. 1919). In the off all futeral science of the bland discussion when all back

Fig. 2.3 Schematic of reactivity of Pd/Fe nanoparticles with chlorinated solvents (Ref: Zhang et al, 1998)

I di sena ante a viso de la constante a de la sera de la consta (sera de la consta (sera)) veras

Noble metals, such as- Pd, Pt, Ru, or Ni have been found to serve as catalyst for dechlorination using nZVI particles (Zhang et al, 1998; Liu et al, 2004). The differences in the surface atomic structure and electronic properties of Pd, Pt, Ru and Ni give them different catalytic properties when used as part of bimetallic nZVI particles (Zhang et al, 1998). Pd/Fe shows the highest activity among bimetallic particle formulations (e.g Pd/Fe, Ru/Fe, Pt/Fe) used for dechlorination of TCE as it has low activation barrier for hydrogen dissociation (2 kcal/mol) on Pd surface (Lien and Zhang, 2007). It has been found that metals such as Pd or Pt have low cathodic hydrogen overpotentials. On the other hand metals like Fe⁰ have high hydrogen overpotential (Lien and Zhang, 2007). The much lower cathodic hydrogen overpotential of Pd compared to that of Fe⁰ causes easy catalyzation of hydrogen dissociation nergies for Pd/Fe bimetallic nanoparticles and bare nZVI particles for dechlorination reactions are 31.1 and 44.9 kJ/mol respectively (Lien and Zhang, 2007).

Chlorinated ethylenes (PCE, TCE, 1,2-DCE, cis- and trans- DCE, VC) are completely reduced by Pd/Fe forming ethane (60 ~ 90%) and ethylene (3 ~ 20%) (Lien and Zhang, 2001). Zhang et al (1998) showed that the reaction times using bimetallic Pd/Fe (20 g/L) for PCE, TCE, cis-DCE and VC (each with initial concentration of 20 mg/L) dechlorination were reduced to 0.25, 0.25, 0.8 and 1.5 hours respectively (i.e., about 100 times faster than that with monometallic nZVI). In the presence of Pd, the reduction of chlorinated ethanes also tends to be more complete leading to ethane formation. With 5g/L of bimetallic Pd/Fe, ethane was detected as the major end product (~87%) when

HCA was the reactant (Lien and Zhang, 2005). On the other hand, 1,1,1-TCA with 5g/L

Dearstars

Sheer

1.A. Martine

111 60.04

.

 $\sum_{i=1}^{n} \frac{1}{i} \sum_{\substack{k=1,\dots,n\\ k \in \mathbb{N}, k \neq n}} \sum_{i=1}^{n} \frac{1}{i} \sum_{\substack{k=1,\dots,n\\ k \in \mathbb{N}, k \neq n}} \sum_{i=1}^{n} \frac{1}{i} \sum_{\substack{k=1,\dots,n\\ k \neq n}} \sum_{i=1}^{n} \frac{1}{i} \sum_{\substack{k=1,\dots,n\\ k \neq n}} \sum_{i=1}^{n} \frac{1}{i} \sum_{\substack{k=1,\dots,n\\ k \neq n}} \sum_{\substack{k=1,\dots,n}} \sum_{\substack{k=1,\dots,n$ pression property

ind Administ

n in the second s

Textor and the

years and a solution

of Pd/Fe formed ethane (~60%) just after 7 hours.

Table 2.1 gives a summary of the studies discussed above regarding the dechlorination of

.

chlorinated ethylenes and ethanes with nZVI particles.

à

n i sing di ang tan sa ang tan sa ang tan sa sa

General,

4

a harris

and the second s

r flyteg men terese e energianeter for th

1. 1. 1. 1. 1.

Table 2.1: Summary	of dechlorination	reaction o	f chlorinated	ethylenes and	1 ethanes
with nZVI					

Research GroupParent compound (mg/L)nZVI concentration (g/L)Type of CatalystDegradation timeDegradation Products ObservedLiu et al, 2005bTCE=290 2005b0.035noneSeveral daysEthane=70%, C3- C6 coupling products=30%Liu et al, 2005bTCE=4.40.194none1.5 hoursEthane=80%, C3- C6 coupling products=20%Zhang et al, 1998TCE=2020none1 hourEthane=80%, C3- C6 coupling products=20%Zhang et al, 1998TCE=2020none1.4 hoursPCESong and Carraway, 2005PCA = 220.08none1.4 hoursPCECarraway, 2005PCA = 160.08none1.3 hoursPCE, TCESong and Carraway, 20051,1,1,2-0.08none110 hoursEthaneCarraway, 2005TCE=2020Pd0.25 hourEthaneCarraway, 2005TCE=2020Pd0.25 hourEthaneLien and 2005TCE=205Pd1.25 hourEthane=87%, EthaneLien and Zhang, 2001PCE=205Pd1.5 hoursEthane=87%, Ethane=87%, Ethane=87%, Ethane=67%, ethene = 6%, PCE=7.5%Lien and Zhang, 20011,1,1-5Pd7 hoursEthane=67% Ethane=67%	Research GroupParent compound (mg/L) $nZVI$ concentration (g/L)Type of CatalystDegradation timeDegradation Products Observed C6 coupling products=30%Liu et al, 2005bTCE=2900.035noneSeveral daysEthane=70%,0 C6 coupling products=20%Liu et al, 2005bTCE=2020none1.5 hoursEthane=80%,0 C6 coupling products=20%Zhang et al, 1998TCE=2020none1 hourEthane=80%,0 C6 coupling products=20%Song and Carraway, 2005HCA = 240.08none1.4 hoursPCESong and Carraway, 2005PCA = 220.08none1.3 hoursPCE, TCESong and Carraway, 20051,1,1,2-0.08none80 hours1,1-DCE, ethatSong and Carraway, 20051,1,2-0.08none110 hoursEthane=Song and al, 19981,1,2-0.08none110 hoursEthaneSong and carraway, 2005TCE=2020Pd0.25 hourEthaneSong and al, 19981,1,2-0.08none110 hoursEthaneCarraway, 2005TCE=2020Pd0.25 hourEthaneLien and Zhang, 2001TCE=205Pd1.25 hourEthane =87%, Ethylene=Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%		and the second				
Group (mg/L)compound (mg/L)concentration (g/L)Catalyst (Lin et al., 2005btime (mg/L)Products Observed Ethane=70%, C3- C6 coupling products=30%Liu et al. 2005bTCE=2900.035noneSeveral daysEthane=70%, C3- C6 coupling products=20%Zhang et al. 1998TCE=2020none1.5 hoursEthane=80%, C3- C6 coupling products=20%Zhang et al. 1998TCE=2020none1 hourEthene, ethane, propene, propane, butene, butane and pentaneSong and Carraway, 2005HCA = 240.08none1.4 hoursPCESong and Carraway, 2005PCA = 220.08none1.3 hoursPCE, TCESong and Carraway, 20051,1,1,2-0.08none110 hoursEthaneSong and Carraway, 20051,1,1,2-0.08none110 hoursEthaneSong and Carraway, 20051,1,2-0.08none110 hoursEthaneCarraway, 2005TCE=2020Pd0.25 hourEthaneLien 'and Zhang, 2001TCE=205Pd1.25 hourEthane=87%, Ethylene=Lien and Zhang, 2001PCE=205Pd1.5 hoursEthane=87%, ethene=6%, PCE=7.5%Lien and Zhang, 20011,1,1-5Pd7 hoursEthane=60%	Group (mg/L)concentration (g/L)Catalyst (g/L)time (mag/L)Products Observed (G C6 coupling products=30%)Liu et al, 2005bTCE=200.035noneSeveral daysEthane=70%,0 (C6 coupling products=30%)Liu et al, 2005bTCE=4.40.194none1.5 hoursEthane=80%,0 (C6 coupling products=20%)Zhang et al, 1998TCE=2020none1 hourEthane=80%,0 (C6 coupling products=20%)Song and Carraway, 2005HCA = 240.08none1.4 hoursPCESong and Carraway, 2005PCA = 220.08none1.3 hoursPCE, TCESong and Carraway, 20051,1,1,2-0.08none1.3 hoursPCE, TCESong and Carraway, 20051,1,2-0.08none110 hoursEthaneSong and 20051,1,2-0.08none110 hoursEthaneSong and	Research	Parent	nZVI	Type of	Degradation	Degradation
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	(mg/L)(g/L)ObservedLiu et al, 2005bTCE=290 0.035 noneSeveral daysEthane=70%, 6 C6 coupling products=30%Liu et al, 2005bTCE=4.4 0.194 none 1.5 hoursEthane=80%, 6 C6 coupling products=20%Zhang et al, 1998TCE=2020none1 hourEthane, ethane propene, propa butene, butane and pentaneSong and Carraway, 2005HCA = 24 0.08 none 1.4 hoursPCESong and Carraway, 2005PCA = 22 0.08 none 1.3 hoursPCE, TCESong and Carraway, 2005 $1,1,1,2$ - TCA = 16 0.08 none 1.3 hours $1,1$ -DCE, ethane and pentaneSong and Carraway, 2005 $1,1,2$ - TCA = 16 0.08 none 110 hoursEthaneSong and carraway, 2005 $1,1,2$ - TCA = 12 0.08 none 110 hoursEthaneSong and carraway, TCA=12 $1,2$ - 0.08 none 110 hoursEthaneSong and carraway, TCA=12 $1,2$ - 0.20 Pd 0.25 hourEthaneZhang et al, 1998TCE=20 20 Pd 0.25 hourEthane =87%, Ethylene=Lien and Zhang, 2001PCE=20 5 Pd 1.5 hourEthane =89%Lien and Zhang, 2001PCE=20 5 Pd 1.5 hourEthane =89%	Group	compound	concentration	Catalyst	time	Products and the
	Liu et al, 2005bTCE=290 0.035 noneSeveral daysEthane=70%, 6 C6 coupling products=30%Liu et al, 2005bTCE=4.4 0.194 none 1.5 hoursEthane=80%, 6 C6 coupling products=20%Zhang et al, 1998TCE=2020none 1 hourEthane=80%, 6 C6 coupling products=20%Zhang et al, 1998TCE=2020none 1 hourEthane=80%, 6 C6 coupling products=20%Song and Carraway, 2005HCA = 24 0.08 none 1.4 hoursPCESong and Carraway, 2005PCA = 22 0.08 none 1.3 hoursPCE, TCESong and Carraway, 2005 $1,1,1,2$ - TCA = 16 0.08 none 1.0 hours $1,1$ -DCE, ethaSong and Carraway, 2005 $1,1,2$ - TCA = 12 0.08 none 110 hoursEthaneSong and 2005 $1,1,2$ - TCA = 12 0.08 none 110 hoursEthaneSong and 2005 $1,1,2$ - TCA = 12 0.08 none 110 hoursEthaneLien and Zhang, 2001TCE=20 20 Pd 0.25 hourEthane = 87% , Ethylene=Lien and Zhang, 2001TCE=20 5 Pd 1.25 hourEthane = 87% , Ethylene=Lien and Zhang, 2001PCE=20 5 Pd 1.5 hourEthane = 89%		(mg/L)	(g/L)			Observed
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	2005bC6 coupling products=30%Liu et al, 2005bTCE=4.40.194none1.5 hoursEthane=80%, G C6 coupling products=20%Zhang et al, 1998TCE=2020none1 hourEthene, ethane propene, propa butene, butane and pentaneSong and Carraway, 2005HCA = 240.08none1.4 hoursPCESong and Carraway, 2005PCA = 220.08none1.3 hoursPCE, TCESong and Carraway, 20051.1,1,2-0.08none1.3 hoursPCE, ethane and pentaneSong and Carraway, 20051,1,1,2-0.08none110 hoursEthaneSong and Carraway, 20051,1,2-0.08none110 hoursEthaneCarraway, 2005TCA=1220Pd0.25 hourEthaneLien and Zhang, 2001TCE=205Pd1.25 hourEthane earra%, Ethylene= 5.8~7.1%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane=89% earra%	Liu et al,	TCE=290	0.035	none	Several days	Ethane=70%, C3-
\sim products=30%Liu et al. 2005bTCE=4.40.194none1.5 hoursEthane=80%, C3- C6 coupling products=20%Zhang et al, 1998TCE=2020none1 hourEthane=80%, C3- C6 coupling products=20%Zhang et al, 1998TCE=2020none1 hourEthane, ethane, propene, propane, buttene, butane and pentaneSong and Carraway, 2005PCA = 220.08none1.4 hoursPCECarraway, 2005PCA = 120.08none1.3 hoursPCE, TCECarraway, 2005TeCA = 160.08none110 hoursEthaneSong and Carraway, 20051,1,2-0.08none110 hoursEthaneSong and 20051,1,2-0.08none110 hoursEthaneSong and 20051,1,2-0.08none110 hoursEthaneSong and 20051,1,2-0.08none110 hoursEthaneCarraway, 2005TCE=2020Pd0.25 hourEthane, ethane, propene, propane, butene, butane and pentaneLien and 2001TCE=205Pd1.25 hourEthane=87%, ethane=83%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane=87%, ethane=89%Lien and Zhang, 2001HCA = 305Pd2 hoursEthane=87%, ethane=60%Lien and Zhang, 20011,1,1-5Pd7 hoursEthane=60%	Liu et al, 2005b $TCE=4.4$ 0.194 none 1.5 hours $Ethane=80\%, 6$ C6 coupling products=20%Zhang et al, 1998 $TCE=20$ 20 none 1 hour $Ethene, ethanepropene, propabutene, butaneand pentaneSong andCarraway,2005HCA = 240.08none1.4 hoursPCESong andCarraway,2005PCA = 220.08none1.4 hoursPCESong andCarraway,2005PCA = 220.08none1.3 hoursPCE, TCESong andCarraway,20051,1,1,2-TCA = 160.08none1.3 hoursPCE, ethaSong andCarraway,20051,1,2-TCA = 120.08none110 hoursEthaneSong andCarraway,TCA=121.2-0.08none110 hoursEthaneLien andZhang, et2001TCE=205Pd1.25 hourEthane = 87\%,Ethylene=5.8 \sim 7.1\%Lien andZhang,2001PCE=205Pd1.5 hourEthane = 89\%$	2005b					C6 coupling
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Liu et al, 2005b $TCE=4.4$ 0.194none1.5 hoursEthane=80%, 6 C6 coupling products=20%Zhang et al, 1998 $TCE=20$ 20none1 hourEthane, ethane propene, propa butene, butane and pentaneSong and Carraway, 2005HCA = 240.08none1.4 hoursPCESong and Carraway, 2005PCA = 220.08none1.3 hoursPCE, TCESong and Carraway, 2005PCA = 160.08none1.3 hoursPCE, thatSong and Carraway, 20051,1,1,2-0.08none110 hoursEthaneSong and Carraway, 20051,1,2-0.08none110 hoursEthaneSong and Carraway, 20051,1,2-0.08none110 hoursEthaneLien and Zhang, et 2001TCE=2020Pd0.25 hourEthane ea7%, f EthaneLien and Zhang, 2001PCE=205Pd1.25 hourEthane e87%, f Ethylene= 5.8~7.1%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%	Marker -	<u>A na sa sa</u>			Asta Mas di	products=30%
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	2005bC6 coupling products=20%Zhang et al, 1998TCE=2020none1 hourEthene, ethane propene, propa butene, butane and pentaneSong and Carraway, 2005HCA = 240.08none1.4 hoursPCESong and Carraway, 2005PCA = 220.08none1.3 hoursPCE, TCESong and Carraway, 2005PCA = 160.08none1.3 hoursPCE, TCESong and Carraway, 20051,1,1,2-0.08none80 hours1,1-DCE, ethaSong and Carraway, TCA = 161,1,2-0.08none110 hoursEthaneSong and Carraway, TCA=121,1,2-0.08none110 hoursEthaneSong and Carraway, TCA=121,1,2-0.08none110 hoursEthaneSong and carraway, TCA=121,1,2-0.08none110 hoursEthaneSong and carraway, TCA=121,1,2-0.08none110 hoursEthaneSong and carraway, TCA=121,1,2-0.08none110 hoursEthaneSong and carraway, TCA=121,1,2-0.08none110 hoursEthaneSong and carraway, TCA=121,1,2-0.08none1.5 hourEthane =87%, Ethylene= 5.8~7.1%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Zhang, 200120055Pd1.5 hourEthane =89%	Liu et al,	TCE=4.4	0.194	none	1.5 hours	Ethane=80%, C3-
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	Zhang et al, 1998TCE=2020none1 hourproducts=20%Song and Carraway, 2005HCA = 240.08none1.4 hoursPCESong and Carraway, 2005PCA = 220.08none1.4 hoursPCESong and Carraway, 2005PCA = 220.08none1.3 hoursPCE, TCESong and Carraway, 20051.1,1,2-0.08none1.3 hoursPCE, thatSong and Carraway, 20051,1,2-0.08none110 hoursEthaneSong and Carraway, TCA = 161,1,2-0.08none110 hoursEthaneSong and Carraway, TCA=121,1,2-0.08none110 hoursEthaneSong and Carraway, TCA=121,1,2-0.08none1.10 hoursEthaneLien and 2005TCE=2020Pd0.25 hourEthane =87%, Ethylene= 5.8~7.1%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Zhang, 20011.5 hourEthane =89%2.4 hoursEthane =89%	2005b					C6 coupling
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Zhang et al, 1998TCE=2020none1 hourEthene, ethane propene, propa butene, butane and pentaneSong and Carraway, 2005HCA = 240.08none1.4 hoursPCESong and Carraway, 2005PCA = 220.08none1.3 hoursPCE, TCESong and Carraway, 2005PCA = 160.08none1.3 hoursPCE, ethane aSong and Carraway, 20051,1,1,2-0.08none80 hours1,1-DCE, ethaneSong and Carraway, 20051,1,2-0.08none110 hoursEthaneSong and Carraway, 20051,1,2-0.08none110 hoursEthaneSong and al, 19981,1,2-0.08none1.10 hoursEthaneLien and 2001TCE=2020Pd0.25 hourEthane =87%, Ethane=87%, Ethane=87%, Ethane=89%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%			· · · · · · · · · · · · · · · · · · ·			products=20%
al, 1998propene, propane, butene, butane and pentaneSong and Carraway, 2005HCA = 24 0.08 none 1.4 hoursPCESong and Carraway, 2005PCA = 22 0.08 none 1.4 hoursPCESong and Carraway, 2005PCA = 22 0.08 none 1.3 hoursPCE, TCESong and Carraway, 2005 $1,1,1,2$ - 0.08 none 80 hours $1,1$ -DCE, ethaneCarraway, 2005TCCA = 16 2005 -1006 -1006 -1006 Song and Carraway, TCA=12 $1,1,2$ - 0.08 none 110 hoursEthaneCarraway, 2005TCE=20 200 Pd 0.25 hourEthane, propene, propane, butene, butane and pentaneLien and Zhang, 2001TCE=20 5 Pd 1.25 hourEthane = 87% , Ethane = 87% , EthaleLien and Zhang, 2001PCE=20 5 Pd 1.5 hourEthane = 89% Lien and Zhang, 2001HCA = 30 5 Pd 2 hoursEthane= 89% , ethene = 6% , PCE= 7.5% Lien and Zhang, TCA=30 $1,1,1$ - 5 Pd 7 hoursEthane= 60%	al, 1998propene, propa butene, butane and pentaneSong and Carraway, 2005HCA = 24 0.08 none 1.4 hoursPCESong and Carraway, 2005PCA = 22 0.08 none 1.4 hoursPCESong and Carraway, 2005PCA = 22 0.08 none 1.3 hoursPCE, TCESong and Carraway, 2005 $1,1,1,2$ - 0.08 none 80 hours $1,1$ -DCE, ethaSong and Carraway, 2005 $1,1,2$ - 0.08 none 110 hoursEthaneSong and Carraway, 2005 $1,1,2$ - 0.08 none 110 hoursEthaneZhang et al, 1998TCE=20 20 Pd 0.25 hourEthene, ethane propene, propa butene, butane and pentaneLien and 2001TCE=20 5 Pd 1.25 hourEthane = 87% , Ethylene= $5.8 \sim 7.1\%$ Lien and Zhang, 2001PCE=20 5 Pd 1.5 hourEthane = 89%	Zhang et	TCE=20	1991 20 a 1994	none	1 hour	Ethene, ethane,
Song and Carraway, 2005HCA = 24 0.08 none 1.4 hoursPCESong and Carraway, 2005PCA = 22 0.08 none 1.4 hoursPCECarraway, 2005PCA = 22 0.08 none 1.3 hoursPCE, TCECarraway, 2005TCCA = 16 0.08 none 80 hours $1,1-DCE$, ethaneCarraway, 2005TCCA = 16 0.08 none 80 hours $1,1-DCE$, ethaneCarraway, 2005TCCA=12 0.08 none 110 hoursEthaneSong and 2005 $1,1,2 0.08$ none 110 hoursEthaneCarraway, 2005TCE=20 20 Pd 0.25 hourEthaneLien and Zhang, 2001TCE=20 5 Pd 1.25 hourEthane =87%, Ethane =87%, Ethane =87%, Ethane =89%Lien and Zhang, 2001PCE=20 5 Pd 1.5 hourEthane=89%Lien and Zhang, 2001HCA = 30 5 Pd 2 hoursEthane=87%, ethene = 6%, PCE=7.5%Lien and Zhang, TCA=30 $1,1,1 5$ Pd 7 hoursEthane=60%	Song and Carraway, 2005HCA = 240.08none1.4 hoursPCESong and Carraway, 2005PCA = 220.08none1.4 hoursPCESong and Carraway, 2005PCA = 220.08none1.3 hoursPCE, TCESong and Carraway, 20051,1,1,2-0.08none80 hours1,1-DCE, ethaSong and Carraway, TeCA = 161,1,2-0.08none110 hoursEthaneSong and Carraway, TCA=121,1,2-0.08none110 hoursEthaneSong and Carraway, TCA=12TCE=2020Pd0.25 hourEthaneZhang et al, 1998TCE=2020Pd1.25 hourEthane =87%, Ethylene= 5.8-7.1%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Lien and Zhang, 2001PCE=205Pd2.5 hourEthane =89%	al, 1998					propene, propane,
Song and Carraway, 2005HCA = 240.08none1.4 hoursPCESong and Carraway, 2005PCA = 220.08none1.3 hoursPCE, TCESong and Carraway, 20051,1,1,2-0.08none1.3 hoursPCE, tTCESong and Carraway, 20051,1,1,2-0.08none80 hours1,1-DCE, ethaneCarraway, 2005TeCA = 16	Song and Carraway, 2005HCA = 24 0.08 none 1.4 hoursPCESong and Carraway, 2005PCA = 22 0.08 none 1.4 hoursPCE, TCESong and Carraway, 2005 $1,1,1,2$ - 0.08 none 1.3 hoursPCE, TCESong and Carraway, 2005 $1,1,1,2$ - 0.08 none 80 hours $1,1$ -DCE, ethaSong and Carraway, 2005 $1,1,2$ - 0.08 none 110 hoursEthaneSong and Carraway, 2005 $1,1,2$ - 0.08 none 110 hoursEthaneSong and Carraway, 2005 $1,1,2$ - 0.08 none 110 hoursEthaneSong and al, 1998 $1,1,2$ - 0.08 none 110 hoursEthaneLien and Zhang, 2001TCE=20 20 Pd 0.25 hourEthane = 87% , Ethylene= $5.8 \sim 7.1\%$ Lien and Zhang, 2001PCE=20 5 Pd 1.5 hourEthane = 89%						butene, butane
Song and Carraway, 2005HCA = 24 0.08 none 1.4 hoursPCESong and Carraway, 2005PCA = 22 0.08 none 1.3 hoursPCE, TCECarraway, 2005TeCA = 16 0.08 none 1.3 hours $1,1$ -DCE, ethaneSong and Carraway, TeCA = 16 $1,1,2$ - 0.08 none 80 hours $1,1$ -DCE, ethaneSong and Carraway, TCA=12 $1,1,2$ - 0.08 none 110 hoursEthaneSong and Carraway, TCA=12 $TCE=20$ 20 Pd 0.25 hourEthene, ethane, propene, propane, butene, butane and pentaneLien and Zhang, 2001TCE=20 5 Pd 1.25 hourEthane = 87% , Ethylene= $5.8-7.1\%$ Lien and Zhang, 2001PCE=20 5 Pd 1.5 hourEthane= 89% Lien and Zhang, 2001HCA = 30 5 Pd 2 hoursEthane= 89% , ethene = 6% , PCE= 7.5% Lien and Zhang, 2001 $1,1,1$ - 5 Pd 7 hoursEthane= 60%	Song and Carraway, 2005HCA = 24 0.08 none 1.4 hoursPCESong and Carraway, 2005PCA = 22 0.08 none 1.3 hoursPCE, TCESong and Carraway, 2005 $1,1,1,2$ - 0.08 none 80 hours $1,1$ -DCE, ethatSong and Carraway, 2005 $1,1,2$ - 0.08 none 80 hours $1,1$ -DCE, ethatSong and Carraway, 2005 $1,1,2$ - 0.08 none 110 hoursEthaneSong and Carraway, 2005 $1,1,2$ - 0.08 none 110 hoursEthaneSong and Carraway, 2005 $1,1,2$ - 0.08 none 110 hoursEthaneSong and Carraway, 2005 $1,2$ - 120 Pd 0.25 hourEthaneLien and Zhang, 2001TCE=20 5 Pd 1.25 hourEthane = 87% , $Ethylene=$ Lien and Zhang, 2001PCE=20 5 Pd 1.5 hourEthane = 89%	. · · ·	-	· · ·			and pentane
$\begin{array}{c ccccc} Carraway, \\ 2005 \\ Song and \\ Carraway, \\ 2005 \\ Song and \\ Carraway, \\ 2005 \\ Song and \\ 1,1,1,2 \\ TeCA = 16 \\ 2005 \\ Song and \\ 1,1,2 \\ TCA = 16 \\ 2005 \\ Song and \\ 1,1,2 \\ TCA = 16 \\ 2005 \\ \hline \end{array} \begin{array}{c} none \\ rec \\ $	Carraway, 2005PCA = 22 0.08 none 1.3 hoursPCE, TCESong and Carraway, 2005 $1,1,1,2$ - 0.08 none 80 hours $1,1$ -DCE, ethatCarraway, Carraway, TeCA = 16TeCA = 16 0.08 none 80 hours $1,1$ -DCE, ethatSong and Carraway, 2005 $1,1,2$ - 0.08 none 110 hoursEthaneSong and Carraway, 2005 $1,1,2$ - 0.08 none 110 hoursEthaneSong and Carraway, 2005 $1,1,2$ - 0.08 none 110 hoursEthaneSong and Carraway, 2005 $1,2$ - 0.25 hourEthene, ethaneLien and Zhang, 2001TCE=20 5 Pd 1.25 hourEthane = 87% , Ethylene=Lien and Zhang, 2001PCE=20 5 Pd 1.5 hourEthane = 89%	Song and	HCA = 24	1.1444 0.08	none	1.4 hours	PCE
2005Cancel and carraway, 2005PCA = 22 0.08 none 1.3 hoursPCE, TCESong and Carraway, 2005 $1,1,1,2$ - 0.08 none 80 hours $1,1$ -DCE, ethaneCarraway, TeCA = 16 0.08 none 80 hours $1,1$ -DCE, ethaneSong and Carraway, TCA=12 0.08 none 110 hoursEthaneCarraway, TCA=12 0.08 none 110 hoursEthaneSong and 1,1,2- 0.08 none 110 hoursEthaneCarraway, TCA=12 200 Pd 0.25 hourEthene, ethane, propene, propane, butene, butane and pentaneLien and TCE=20 5 Pd 1.25 hourEthane =87%, Ethylene= $5.8~7.1\%$ Lien and PCE=20 5 Pd 1.5 hourEthane =89%Zhang, 2001 2001 11.5 hourEthane =89%Lien and HCA = 30 5 Pd 2 hoursEthane=87%, ethene = 6%, PCE= 7.5%Lien and 1,1,1- 5 Pd 7 hoursEthane =60%	2005PCA = 220.08none1.3 hoursPCE, TCECarraway, 2005 2005 $1,1,1,2$ - 0.08 none 80 hours $1,1$ -DCE, ethatCarraway, 2005TeCA = 16 0.08 none 80 hours $1,1$ -DCE, ethatSong and 2005 $1,1,2$ - 0.08 none 110 hoursEthaneCarraway, 2005TCA=12 0.08 none 110 hoursEthaneCarraway, 2005TCE=20 20 Pd 0.25 hourEthene, ethane propene, propa butene, butane and pentaneLien 2001TCE=20 5 Pd 1.25 hourEthane = 87% , Ethylene= $5.8 \sim 7.1\%$ Lien 2001PCE=20 5 Pd 1.5 hourEthane = 89%	Carraway,		•			
Song and Carraway, 2005PCA = 22 0.08 none 1.3 hoursPCE, TCESong and Carraway, TeCA = 16 $1,1,1,2$ - TeCA = 16 0.08 none 80 hours $1,1$ -DCE, ethane2005TeCA = 16 0.08 none 110 hoursEthane2005TCA=12 0.08 none 110 hoursEthane2005TCA=12 0.08 none 110 hoursEthane2005TCE=20 20 Pd 0.25 hourEthene, ethane, propene, propane, butene, butane and pentaneLien and Zhang, 2001 TCE=20 5 Pd 1.25 hourEthane = 87% , Ethylene= $5.8~7.1\%$ Lien and Zhang, 2001 PCE=20 5 Pd 1.5 hourEthane = 89% Lien and Zhang, 2001 HCA = 30 5 Pd 2 hoursEthane= 87% , ethene = 6% , PCE= 7.5% Lien and Zhang, 2001 $1,1,1$ - 5 Pd 7 hoursEthane = 60%	Song and Carraway, 2005PCA = 22 0.08 none 1.3 hoursPCE, TCESong and 2005 $1,1,1,2$ - TeCA = 16 0.08 none 80 hours $1,1$ -DCE, ethatSong and 2005 $1,1,2$ - TCA=12 0.08 none 110 hoursEthaneSong and Carraway, 2005 $1,1,2$ - TCA=12 0.08 none 110 hoursEthaneZhang et al, 1998TCE=2020Pd 0.25 hourEthene, ethane propene, propa butene, butane and pentaneLien 2001TCE=205Pd 1.25 hourEthane = 87% , Ethylene= $5.8 \sim 7.1\%$ Lien Zhang, 2001PCE=205Pd 1.5 hourEthane = 89%	2005					· · · · · ·
Carraway, 2005Image: Carraway, 2005Image: Carraway, recA = 16Image: Carraway, recA = 16Image: Carraway, recA = 16Image: Carraway, recA = 16Image: Carraway, recA = 12Image: Carraway, <b< td=""><td>Carraway, 2005Image: Carraway, 2005Image: Carraway, TeCA = 16Image: Carraway, TeCA = 16Image: Carraway, TeCA = 16Image: Carraway, TeCA = 12Image: Carraway,<b< td=""><td>Song and</td><td>PCA = 22</td><td>0.08</td><td>none</td><td>1.3 hours</td><td>PCE, TCE</td></b<></td></b<>	Carraway, 2005Image: Carraway, 2005Image: Carraway, TeCA = 16Image: Carraway, TeCA = 16Image: Carraway, TeCA = 16Image: Carraway, TeCA = 12Image: Carraway, <b< td=""><td>Song and</td><td>PCA = 22</td><td>0.08</td><td>none</td><td>1.3 hours</td><td>PCE, TCE</td></b<>	Song and	PCA = 22	0.08	none	1.3 hours	PCE, TCE
2005 $(1,1,1,2)$ $(0,08)$ none (80 hours) $(1,1-DCE, \text{ ethane})$ Carraway, 2005TeCA = 16 $(1,1,2)$ $(1,1,2)$ $(1,1,2)$ $(1,1,2)$ $(1,1,2)$ Song and Carraway, 2005 $(1,1,2)$ $(1,1,2)$ $(1,1,2)$ $(1,1,2)$ $(1,1,2)$ Zhang et al, 1998TCE=20 (20) Pd $(0,25)$ hourEthaneLien and Zhang, 2001TCE=20 (20) Pd $(1,25)$ hourEthane = 87\%, Ethane = 87\%, Ethane = 87\%, Ethane = 87\%, Ethane, $(20,1)$ Lien and Zhang, 2001PCE=20 (5) Pd $(1,5)$ hourEthane = 89\%Lien and Zhang, 2001PCE=20 (5) Pd (2) hoursEthane = 87\%, ethane = 89\%Lien and Zhang, 2001HCA = 30 (5) Pd (2) hoursEthane = 87\%, ethane = 6\%, PCE= 7.5\%Lien and Zhang, 2001 $(1,1,1)$ (5) Pd (7) hoursEthane = 60\%	2005 $1,1,1,2$ 0.08 none 80 hours $1,1$ -DCE, ethatCarraway, 2005TeCA = 16 0.08 none 80 hours $1,1$ -DCE, ethatSong and Carraway, 2005 $1,1,2$ - TCA=12 0.08 none 110 hoursEthaneZhang et al, 1998TCE=2020Pd 0.25 hourEthene, ethane propene, propa butene, butane and pentaneLien 2001and PCE=205Pd 1.25 hourEthane = 87% , Ethylene= $5.8 \sim 7.1\%$ Lien Zhang, 2001PCE=205Pd 1.5 hourEthane = 89%	Carraway,	Residente d'Arres				
Song and Carraway, 20051,1,1,2- TeCA = 160.08none80 hours1,1-DCE, ethaneSong and Carraway, TCA=121,1,2- TCA=120.08none110 hoursEthaneZhang et al, 1998TCE=2020Pd0.25 hourEthene, ethane, propene, propane, butene, butane and pentaneLien and Zhang, 2001TCE=205Pd1.25 hourEthane =87%, Ethylene= $5.8 \sim 7.1\%$ Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Lien and Zhang, 2001HCA = 305Pd2 hoursEthane=87%, ethene = 6%, PCE=7.5%Lien and Zhang, 20011,1,1-5Pd7 hoursEthane=60%	Song and Carraway, 20051,1,1,2- TeCA = 160.08none80 hours1,1-DCE, ethatSong and Carraway, 20051,1,2- TCA=120.08none110 hoursEthaneZhang et al, 1998TCE=2020Pd0.25 hourEthene, ethane propene, propa butene, butane and pentaneLien 2001and PCE=205Pd1.25 hourEthane =87%, Ethylene= 5.8~7.1%Lien 2001and PCE=205Pd1.5 hourEthane =89%	2005					
Carraway, 2005TeCA = 16none110 hoursEthaneSong and Carraway, TCA=12 $1,1,2$ - TCA=12 0.08 none 110 hoursEthane2005TCA=12 0.25 hourEthene, ethane, propene, propane, butene, butane and pentaneEthene, ethane, propene, propane, butene, butane and pentaneLien and Zhang, 2001TCE=20 5 Pd 1.25 hourEthane = 87% , Ethylene= $5.8~7.1\%$ Lien and Zhang, 2001PCE=20 5 Pd 1.5 hourEthane = 89% Lien and Zhang, 2001HCA = 30 5 Pd 2 hoursEthane= 87% , ethene = 6% , PCE= 7.5% Lien and Zhang, 2001 $1,1,1$ - 5 Pd 7 hoursEthane = 60%	Carraway, 2005TeCA = 16none110 hoursEthaneSong and Carraway, 20051,1,2-0.08none110 hoursEthaneZhang et al, 1998TCE=2020Pd0.25 hourEthene, ethane propene, propa butene, butane and pentaneLien and Zhang, 2001TCE=205Pd1.25 hourEthane =87%, Ethylene= $5.8~7.1\%$ Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%	Song and	1,1,1,2-	0.08	none	80 hours	1,1-DCE, ethane
2005 $1,1,2 0.08$ none 110 hoursEthaneCarraway, 2005TCA=12 0.08 none 110 hoursEthaneZhang et al, 1998TCE=20 20 Pd 0.25 hourEthene, ethane, propene, propane, butene, butane and pentaneLien and Zhang, 2001TCE=20 5 Pd 1.25 hourEthane =87%, Ethylene= $5.8~7.1\%$ Lien and Zhang, 2001PCE=20 5 Pd 1.5 hourEthane =89%Lien and Zhang, 2001PCE=30 5 Pd 1.5 hourEthane =89%Lien and Zhang, 2001HCA = 30 5 Pd 2 hoursEthane=87%, ethene = 6%, PCE= 7.5%Lien and Zhang, 2001 $1,1,1 5$ Pd 7 hoursEthane=60%	2005Song and Carraway, 20051,1,2- TCA=120.08none110 hoursEthaneZhang et al, 1998TCE=2020Pd0.25 hourEthene, ethane propene, propa butene, butane and pentaneLien 2001TCE=205Pd1.25 hourEthane =87%, Ethylene= 5.8~7.1%Lien 2001PCE=205Pd1.5 hourEthane =89%Zhang, 2001SPd1.5 hourEthane =89%	Carraway,	TeCA = 16				
Song and Carraway, TCA=121,1,2- TCA=120.08none110 hoursEthaneZhang et al, 1998TCE=2020Pd0.25 hourEthene, ethane, propene, propane, butene, butane and pentaneLien and Zhang, 2001TCE=205Pd1.25 hourEthane =87%, Ethylene= $5.8~7.1\%$ Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =87%, Ethylene= $5.8~7.1\%$ Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Lien and Zhang, 2001HCA = 305Pd2 hoursEthane=87%, ethene = 6%, PCE=7.5%Lien and Zhang, 20011,1,1-5Pd7 hoursEthane=80%	Song and Carraway, 20051,1,2- TCA=120.08none110 hoursEthaneZhang et al, 1998TCE=2020Pd0.25 hourEthene, ethane propene, propa butene, butane and pentaneLien and Zhang, 2001TCE=205Pd1.25 hourEthane =87%, Ethylene= 5.8~7.1%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%	2005	n de talèncie de l'ha	in de la maintaire de la m			ent dage Baan analogo
Carraway, 2005TCA=12Image: Carraway, 2005TCA=12Image: Carraway, 2005TCE=2020Pd0.25 hourEthene, ethane, propene, propane, butene, butane and pentaneLien 2001TCE=205Pd1.25 hourEthane =87%, Ethylene= 5.8~7.1%Lien 2001PCE=205Pd1.5 hourEthane =87%, Ethylene= 5.8~7.1%Lien 2001PCE=205Pd1.5 hourEthane =89%Lien 2001PCE=305Pd2 hoursEthane=87%, ethene = 6%, PCE=7.5%Lien 20011,1,1-5Pd7 hoursEthane =60%	Carraway, 2005TCA=12Zhang et al, 1998TCE=2020Pd0.25 hourEthene, ethane propene, propa butene, butane and pentaneLien and Zhang, 2001TCE=205Pd1.25 hourEthane =87%, Ethylene= 5.8~7.1%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%	Song and	1,1,2-	0.08	none	110 hours	Ethane
2005TCE=2020Pd0.25 hourEthene, ethane, propene, propane, butene, butane and pentaneLien and Zhang, 2001TCE=205Pd1.25 hourEthane =87%, Ethylene= $5.8 \sim 7.1\%$ Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =87%, Ethylene= $5.8 \sim 7.1\%$ Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Lien and Zhang, 2001HCA = 305Pd2 hoursEthane=87%, ethene = 6%, PCE= 7.5%Lien and Zhang, 20011,1,1-5Pd7 hoursEthane =60%	2005Image: TCE=2020Pd0.25 hourEthene, ethane propene, propa butene, butane and pentaneLien and Zhang, 2001TCE=205Pd1.25 hourEthane =87%, Ethylene= 5.8~7.1%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%	Carraway,	TCA=12		in an	and the second second	
Zhang et al, 1998TCE=2020Pd0.25 hourEthene, ethane, propene, propane, butene, butane and pentaneLien and Zhang, 2001TCE=205Pd1.25 hourEthane =87%, Ethylene= 5.8~7.1%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Lien and Zhang, 2001PCE=305Pd1.5 hourEthane =89%Lien and Zhang, 2001HCA = 305Pd2 hoursEthane=87%, ethene = 6%, PCE=7.5%Lien and Zhang, 20011,1,1-5Pd7 hoursEthane =60%	Zhang et al, 1998TCE=2020Pd0.25 hourEthene, ethane propene, propa butene, butane and pentaneLien and Zhang, 2001TCE=205Pd1.25 hourEthane =87%, Ethylene= 5.8~7.1%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%	2005					
al, 1998propene, propane, butene, butane and pentaneLien and Zhang, 2001TCE=205Pd1.25 hourEthane =87%, Ethylene= 5.8~7.1%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Lien and Zhang, 2001PCE=305Pd2 hoursEthane=87%, ethene = 6%, PCE= 7.5%Lien and Zhang, 20011.1.5Pd2 hoursEthane=87%, ethene = 6%, PCE= 7.5%Lien and Zhang, 20011.1.5Pd7 hoursEthane=60%	al, 1998propene, propa butene, butane and pentaneLien and TCE=205Pd1.25 hourEthane =87%, Ethylene= 5.8~7.1%Zhang, 20015Pd1.5 hourEthane =89%Lien and Zhang, 2001PCE=205Pd1.5 hourLien and Zhang, 2001PCE=205Pd1.5 hourLien and Zhang, 2001PCE=205Pd1.5 hourLien and Zhang, 2001PCE=205Pd2 hours	Zhang et	TCE=20	20	Pd	0.25 hour	Ethene, ethane,
Lien and Zhang, 2001TCE=205Pd1.25 hourEthane =87%, Ethylene= $5.8~7.1\%$ Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Lien and Zhang, 2001HCA = 305Pd2 hoursEthane=87%, ethene = 6%, PCE= 7.5%Lien and Zhang, 20011,1,1-5Pd7 hoursEthane =60%	Lien and Zhang, 2001TCE=205Pd1.25 hourEthane =87%, Ethylene= 5.8~7.1%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%	al, 1998					propene, propane,
Lien and Zhang, 2001TCE=205Pd1.25 hourEthane =87%, Ethylene= $5.8~7.1\%$ Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Lien and Zhang, 2001HCA = 305Pd2 hoursEthane=87%, ethene = 6%, PCE= 7.5%Lien and Zhang, 20011,1,1-5Pd7 hoursEthane =60%	Lien and Zhang, 2001TCE=205Pd1.25 hourEthane =87%, Ethylene= 5.8~7.1%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%	a alfadalar de			enned st		butene, butane
Lien and Zhang, 2001TCE=205Pd1.25 hourEthane =87%, Ethylene= $5.8~7.1\%$ Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Lien and Zhang, 2001HCA = 305Pd2 hoursEthane=87%, ethene = 6%, PCE= 7.5%Lien and Zhang, 20011,1,1-5Pd7 hoursEthane =60%	Lien and Zhang, 2001TCE=205Pd1.25 hourEthane =87%, Ethylene= 5.8~7.1%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%				· · ·		and pentane
Zhang, 2001Ethylene= $5.8~7.1\%$ Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Lien and Zhang, 2001HCA = 305Pd2 hoursEthane=87%, ethene = 6%, PCE=7.5%Lien and Zhang, 20011,1,1-5Pd7 hoursEthane=60%	Zhang, 2001Ethylene= 5.8~7.1%Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Lien and Zhang, 2001Pd2 hoursEthane =89%	Lien and	TCE=20	· · · · 5	Pd	1.25 hour	Ethane =87%,
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	2001 5.8~7.1% Lien and PCE=20 5 Pd 1.5 hour Ethane =89% Zhang, 2001 2001 Ethane =89% Ethane =89% Ethane =89%	Zhang,				a a cara a c	Ethylene=
Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Lien and Zhang, 2001HCA = 305Pd2 hoursEthane=87%, ethene = 6%, PCE= 7.5%Lien and Zhang, 20011,1,1-5Pd7 hoursEthane =60%	Lien and Zhang, 2001PCE=205Pd1.5 hourEthane =89%Lien and Lien and 	2001		a de la companya de l			5.8~7.1%
Zhang, 2001HCA = 305Pd2 hoursEthane=87%, ethene = 6%, PCE= 7.5%Lien and Zhang,1,1,1-5Pd7 hoursEthane =60%, Ethane =60%	Zhang, 2001	Lien and	PCE=20	5	Pd	1.5 hour	Ethane =89%
2001Image: Description of the second systemImage: Description of the second systemImage: Description of the second systemLien and 1,1,1-5Pd2 hoursEthane=87%, ethene = 6%, ethene = 6%, PCE= 7.5%Lien and 1,1,1-5Pd7 hoursEthane =60%Zhang,TCA=30Image: Description of the second systemImage: Description of the second system	$\frac{2001}{1 \text{ Lien and } 1104 - 20}$	Zhang,		international Constraints		28.649 (VA)는 4	
Lien and Zhang, 2001HCA = 305Pd2 hoursEthane=87%, ethene = 6%, 	Lion and $UCA = 20$	2001					
Zhang, 2001ethene = 6%, PCE= 7.5%Lien and 1,1,1- Zhang,5Pd7 hoursEthane =60%	$\begin{bmatrix} \text{Lien and } \text{HCA} = 30 \end{bmatrix} = 5 \qquad \begin{bmatrix} \text{Pu} \\ \text{Illen and } \text{Pu} \end{bmatrix} = 2 \text{ Hours} \qquad \begin{bmatrix} \text{Ethane} = 87\%, \\ \text{Illen and } Illen $	Lien and	HCA = 30	1	Pd	2 hours	Ethane=87%,
2001 PCE= 7.5% Lien and 1,1,1- 5 Pd 7 hours Ethane =60% Zhang, TCA=30 7 1000000000000000000000000000000000000	Zhang, ethene = 6%,	Zhang,					ethene = 6% ,
Lien and1,1,1-5Pd7 hoursEthane =60%Zhang,TCA=307DescriptionDescriptionDescription	2001 PCE=7.5%	2001	and the second				PCE=7.5%
Zhang, TCA=30 and a feature state of the control of the state of the s	Lien and 1,1,1- 5 Pd 7 hours Ethane =60%	Lien and	1,1,1-	5	Pd	7 hours	Ethane =60%
	Zhang, TCA=30 and a desired and share in the state of the state of the state of the state of the state state of the state	Zhang,	TCA=30	18 193 등 42월 <u>원</u>	and the Andrews and Andrews	na shina qu	nt sha gaan jilaadaa
2001		2001		· · · · ·			

Therefore, it can be concluded that environmentally benign and highly efficient nZVI particles show excellent performance in the remediation of chlorinated ethylenes and ethanes. Monometallic nZVI particles, when reacted with TCE, form primarily ethane with lesser amount of even numbered saturated hydrocarbons like butane and hexane and inorganic chlorides. In case of HCA, PCA, 1,1,1,2-TeCA, less chlorinated byproducts and ethane formed at the end of the reaction. On the other hand, the utilisation of bimetallic nZVI particles such as Pd/Fe ensures a complete reduction at a faster rate forming mainly ethane as end products; however a smaller amount of chlorinated byproducts may be evolved during reaction which disappears instantly.

a state of the second of the second second state.

2.2.3 Stabilization of Iron Nanoparticles Sciences of the submet of the Science of the second states of the second

In an aqueous system, direct inter-particle interactions such as Van der Waals forces and magnetic interactions cause nanoiron particles to form larger flocs and thus induce agglomeration, which in turn decreases chemical reactivity of nZVI particles (He and Zhao, 2007; Sakulchaicharoen et al, 2010). Supporting nanoparticles with selected stabilizers can diminish agglomeration, decrease particle size and thus result in a substantial gain of net reactivity by either electrostatic repulsion or steric hindrance (Schrick et al, 2002; He and Zhao, 2005; He and Zhao, 2007; Sakulchaicharoen et al, 2010). Schrick et al (2002) used carbon nanoparticles and poly acrylic acid (PAA) as supports for nZVI particles. Saleh et al (2007) utilized triblock copolymer (PMAA-PMMA-PSS) and the surfactant sodium dodecylbenzene sulfonate (SDBS) for modifying commercial nZVI to dechlorinate DNAPLs. Starch-stabilized bimetallic nanoparticles were prepared for degradation of chlorinated hydrocarbons in water by He and Zhao
(2005). In a different study, He and Zhao (2007) successfully used carboxymethyl cellulose (CMC), a water soluble, low-cost and environmentally friendly polyelectrolyte for the same purpose. The starched nanoparticles in the study of He and Zhao (2005) displayed much less agglomeration and greater dechlorination than bare nZVI; however a stronger interaction and higher dechlorination rate was observed when CMC is used instead of starch with nZVI particles. (He and Zhao, 2007). CMC molecules adsorb onto the surface of the nanoparticles forming a thin negatively charged layer that suppresses agglomeration and exhibits higher reactivity compared to nonstabilized nanoparticles (He and Zhao, 2007). Sakulchaicharoen et al (2010) found that bimetallic Pd/Fe (Fe=0.1 g/L, Pd=0.1 wt%) encapsulated with 0.2% (w/w) CMC700K completely degraded TCE (nZVI/TCE=2 wt ratio) within 6 hours whereas Pd/Fe without CMC was able to dechlorinate only 37% of TCE at the end of 24 hours. He and Zhao (2007) showed that 0.2% (w/w) CMC-stabilized Pd/Fe transformed 50 mg/L of TCE (nZVI/TCE=1.17) into ethane and Cl- within 40 minutes. On the other hand, in the same study non-stabilized Pd/Fe took 2 hours for about 40% TCE dechlorination (He and Zhao, 2007). Thus compared to non-stabilized nZVI partcles, supported nZVI particles display faster reactivity, which enhances their efficacy for groundwater remediation.

2.3 DEGRADATION OF 1,2-DCA . Refer to full test patients in the object of the

The degradation of 1,2-dichloroethane (1,2-DCA) or ethylene dichloride (EDC) is of particular interest in this study because unlike other halogenated alkanes, 1,2-DCA shows no response to nZVI treatment. 1,2-DCA is a chlorinated hydrocarbon, highly flammable and colorless liquid with a chloroform like odor. It is used in chemical manufacturing as

Witherman and Massimon Land Spin and Apply States and a 1995. The base film

an intermediate in the synthesis of polyvinyl chloride and as a solvent in pharmaceutical products synthesis (Klecka et al, 1998; Marzorati et al, 2005). The health risks associated with 1,2-DCA are linked to acute and long-term toxic effects affecting human liver, kidneys, neurological, cardiovascular, and immune systems. Chloroacetaldehyde, produced in vivo due to oxidative conversion of 1,2-DCA, has established carcinogen effects. (De Wildeman and Verstraete, 2003). The drinking water standard of 1,2-DCA in Canada is 0.005 mg/L (Health Canada). Because of its relatively high water solubility (8,700 mg/L), high density (1.253 g/cc), its potential for migration in soil and high resistance to reduction reactions, the residence time of 1,2-DCA in the environment can range from months to decades depending on environmental conditions (e.g. oxidation-reduction potential, pH, dissolved oxygen, temperature, conductivity etc.) (Marzorati et al, 2005). For these reasons, the development and implementation of reliable and effective remediation strategies for 1,2-DCA is a matter of great importance.

2.3.1 Degradation Pathways of 1,2-DCA

In the subsurface, under both aerobic and anaerobic conditions, specific microorganisms are able to transform 1,2-DCA into non-toxic end products (Zaan et al, 2009; De Wildeman and Verstraete 2003; Dyer et al, 2000; Klecka et al, 1998). Fig 2.4 describes dechlorination reactions of 1,2-DCA following different pathways depending on the prevailing reductive-oxidative conditions of an aquifer. Under aerobic conditions, through the oxidation process, 1,2-DCA is completely mineralized into CO_2 , H₂O and CI forming 2-chloroethanol and 2-chloroacetate as intermediates but these are not easily detectable in the field (the left hand pathway of Fig 2.4) (Zaan et al, 2009; Nobre et al,

2004; Klecka et al, 1998). Under anaerobic conditions, 1,2-DCA may also be oxidized under denitrifying conditions, with nitrate as the electron acceptor, forming CO₂, H₂O and CI^{\cdot} (the middle pathway in Fig 2.4) (Dyer et al, 2000; Nobre et al, 2004). Anaerobic degradation can take place through both biotic and abiotic processes (the right hand pathways of Fig 2.4). Reductions such as reductive dechlorination (equation 2.12) and dihaloelimination (equation 2.13) are involved in biotic process whereas hydrolysis (equation 2.14) takes place under abiotic conditions. Vinyl chloride, formed as an intermediate during the abiotic process of hydrolysis under alkaline condition, needs to undergo anaerobic oxidation to form CO₂ because of its low oxidation state (Dyer et al, 2000). Although anaerobic biological remediation can lead to successful dechlorination of 1¹₂-DCA, it is a very slow process and may take years to decades to complete the dechlorination reaction in the sub-surface environment.







Entitle Mark and a pression of the defended on the Net wat it bim night hiz VI va fight, strengt i te dage de 1.2504 (a hive be defenderobic Condition and the form of the mode of the Anaerobic Condition (2021) E. Marsen Cl a se a contra la contra la contra de la contra for Houman elect Oxidation of Alabert 4), 4 313 Cl 90 A H**Y**A Dihaloelimination (22-143.2-1.2-DCA Ck OH Hydrolysis and 2-chloroethanol Reductive Denitrification dechlorination <u>.</u> 199 AND STREET, ST C=C \downarrow is the Classical and CO₂, H₂O, Cl⁻ Ethylene Chloroethane 2-chloroacetate 0 Vinyl Chloride -C-C-TOH AN AND A DATA SALAR AN Anaerobic Ethane Oxidation glycolate (Iron reducers) CO_2 , H_2O , CI^- which are the constraint of the constraint of the constraint of CO_2 and CO_2 and

Fig-2.4- 1,2-DCA Transformation Pathways under Different Conditions (Nobre et al, 2004)

2.3.2 Degradation of 1,2-DCA with metals

1,2-DCA has shown slight dechlorination in abiotic systems by metals due to its extreme recalcitrance to reduction whereas other halogenated alkanes were successfully remediated through this method (Ferrey et al 2004; Zhang et al 1998; Song and Carraway

2005; De Wildeman and Verstraete 2003). While a variety of chlorinated ethanes (e.g.,

HCA, PCA, 1,1,1,2-TeCA, 1,1,2,2-TeCA and 1,1,2-TCA) have been degraded easily using Fe⁰, Zn⁰ as well as bimetallic nZVI particles, attempts to degrade 1,2-DCA have been only marginally successful (Arnold et al, 1999; Vanstone et al, 2008). Zn⁰ is known to be a highly reactive zerovalent metal with a standard reduction potential of -0.76V, facilitating experiments with slowly reacting chlorinated compounds (e.g. 1,1-DCA, 1,2-DCA) (Vanstone et al, 2008). However, even when a very large amount of Zn⁰ (Zn⁰/1,2-DCA = 8000 wt ratio) reacted with 1,2-DCA, it degraded very slowly. About 30% of total 1,2-DCA was degraded at the end of 12 days, forming ethylene as the degradation product; though chloride evolution test was not carried at (Vanstone et al, 2008). On the other hand, 1,2-DCA has been found so far to be absolutely resistant to reduction by iron (Ferrey et al. 2004; Zhang et al 1998; Song and Carraway 2005; De Wildeman and Verstraete 2003, Kopinke et al 2004). For instance, Song and Carraway (2005) failed to show any measurable reduction (<5%) of 1,2-DCA with nZVI within 40 days.

It has been proposed that the average oxidation state of the carbon atoms that bear chlorine atoms aids to analyse the recalcitrance of C2-C4 chloroalkanes in reductive or oxidative environments (De Wildeman and Verstraete, 2003). The lower the oxidation state of the carbon atoms, the less susceptible they are to reduction (De Wildeman and Verstraete, 2003).

definite des construits de la construction de la construction de la seconda de la seconda de la seconda de la s

Table 2.2 presents the nominal oxidation states for several chlorinated hydrocarbons and their dechlorinated products.

and the second second second to a second second

ality same approximation and the second statement of the second approximation of the second statement of t

	+2		-1	-2
Ethylenes	PCE →	TCE →	cis-1,2-DCE	\rightarrow VC \rightarrow	Ethylene
		un Chaisterian Anna an A	a (de ferier) Secondaria	n Brohm Dae Bei Seise anne Seise	Alexandrean (* 1997) Alexandrean (* 1997)
Ethanes	Colettai.	ing and bee	1,1,2-TCA→	→ 1,2-DCA→	Ethylene
			an a		

Table 2.2- Redox Table (Ref: De Wildeman and Verstraete, 2003) *** Entertality Huston for

la serie de la composición de la compos

Table 2.2 reflects a zone, moving to the right from oxidation state 0, where reductive dechlorination is uncommon. For instance, carbon atoms associated with PCE have an oxidation state of +2 that makes reduction very easy but oxidation difficult. On the other hand, cis-1,2-DCE and VC with oxidation state of 0 and -1 respectively show comparatively slower reduction than PCE (De Wildeman and Verstraete, 2003). Although VC, with an oxidation state of -1, is susceptible to slow reduction, 1,2-DCA despite having the same oxidation number (-1) exhibits stability to reductive dechlorination (De Wildeman and Verstraete, 2003). Almost no degradation in the reductive conditions has been discussed for lower chlorinated alkanes containing 2-4 chlorine atoms (De Wildeman and Verstraete, 2003). The role of halogenated compounds as electron acceptors in anaerobic environments has also been rationalized by the Gibb's free energy (ΔG^0) of reactions. Table 2.3 reflects the half reaction Gibb's free energies for reductive dechlorination. Here PCE, having a ΔG^0 -55.4 kJ/electron, reduces more easily than TCE (ΔG^0 -53.1 to -50.9 kJ/electron) and 1.2-DCA is placed at the bottom of the table with $\Delta G^0 = -36.2$ kJ/electron acting as the weakest electron acceptor that supports the data provided in Table 2.2. Although 1,1-DCA being just above 1,2-DCA in table 2.3 with a ΔG^0 value of -38.3 is transformed at a very slow rate by reacting with nZVI producing ethane as end product (Song and Carraway, 2005), 1,2-DCA has so far

failed to breakdown by reacting with nZVI. A conduct memory of the last operation and the second sec

Table 2.3: Half Reactions Gibb's Free Energies of Reductive Dechlorinations. The ΔG^0 electron values are calculated at standard conditions partial pressures 1 atm, pH=7.0 and [Cl]⁻=1mM. (Ref: Dolfing and Mueller)

di Buu tafang	网络拉马拉克拉拉拉西部瓦尔克马马		u në në dëvikaset nëse kat e	n de la serie d	
Electron	Half-reaction of reductive	ΔG^0	E ⁰		
Acceptor				(kJ/electron)	(mV)
				un 11 Hotten	
Pd ²⁺	$Pd^{2+} + 2e^{-}$	\rightarrow	Pd ⁰	-88.3	915
1				a da ta	Andreda i
PCE	$C_2Cl_4 + H^+ + 2e^-$	\rightarrow	$C_2HCl_3 + Cl^2$	-55.4	574
1,1,1 - TCA	$CCl_3-CH_3 + H^+ + 2e^-$	→	$CHCl_2-CH_3 + Cl^-$	-54.1	561
TCE	$C_2HCl_3 + H^+ + 2e^-$	\rightarrow	$C_2H_2Cl_2 + Cl^2$	-53.1_to -50.9	550 to
					527
1,1,2-TCA	$CHCl_2-CH_2Cl + H^+ + 2e^-$	\rightarrow	$C_2H_4Cl_2 + Cl^2$	-51.9 to -49.8	538 to
			an an ann an Arailtean an Arailte	en de la companya de La companya de la comp	510
VC	$C_2H_3Cl + H^+ + 2e^-$	\rightarrow	$C_2H_4 + Cl^-$	-43.4	450
1,2-DCE	$C_2H_2Cl_2 + H^+ + 2e^-$	\rightarrow	$C_2H_3Cl + Cl$	-40.6 to -38.3	420 to
2.155.					397
1,1-DCA	$C_2H_4Cl_2 + H^+ + 2e^-$	\rightarrow	$C_2H_5Cl + Cl^-$	-38.3	397
1,2-DCA	CH_2Cl - CH_2Cl + H^+ + 2e	\rightarrow	$C_2H_5Cl + Cl$	-36.2	375
Fe ²⁺	$Fe^{2+} + 2e^{-}$	→	Fe ⁰	(1 +42.5 (14)	-440
Zn ²⁺	$Zn^{2+} + 2e^{-}$	÷	Zn ⁰	+73.6	-763

Moreover, the dechlorination reactivity of chlorinated hydrocarbons is expected to be affected by the bond strength of the carbon-chlorine bond, the electron affinity of the carbon chlorine bond and the stability of the carbon-radical species resulting from an

initial electron transfer pathway (Larson and Weber, 1994). Table 2.4 illustrates the dependence of reactivity on structures of chlorinated ethanes where the order of reactivity of HCA, 1,1,2,2-TeCA and 1,2-DCA is linked to differences in the dissociation energy of the C-Cl bond. (Larson and Weber)

 Table 2.4 Relative Rates for the Vicinal Dechlorination of Chlorinated Ethanes in Reducing Sediment-Water Slurry (Larson and Weber, 1994)

Compound	k ^{obs} (min ⁻¹)	t _{1/2}	Bond dissociation	Electron Affinity of Associated
gradevices Tax Rev Point Course		n Alan ayan ayan Alan ayan ayan	kcal/mol)	chlorine(kcal/mol)
Hexachloroethane	1.9×10^{-2}	36 min	72	83
	royan di nakan			$ \begin{array}{c} \left(\left(\begin{array}{c} \left(\left(\begin{array}{c} \left(\left(\left(\begin{array}{c} \left($
1,1,2,2-	7.3 x 10 ⁻⁵	6.6 d	69.8	77.5
Tetrachloroethane	nad para p			skin za újávojav
			••	
1,2-Dichloroethane	<1.4 x 10 ⁻⁵	735 d	<47	70.5

Under the experimental conditions of the study by Larson and Weber, alkenes were formed at the end of reactions through the pathway of vicinal dechlorination (equation 2.15).

SMONT.

r photoche phe actual a l'alla fair constance de service phenone de la phenone de la service de la service de l

) where ${f CI}$, we ${f CI}$ is the result of the ${f CI}$ is the regular of the ${f H}$ gives the H strength regular to the result of the regular the regular to th H--H . (Equation 2.15) H_____ ▶ H--C1 Ĥ Ĥ

Vicinal Dechlorination

an 1956 (1966) - 网络白云 白云 新闻 (1966) - An 1967 (1966)

医含磷酸盐酸 品牌 法公司部

2.3.3 Degradation of 1,2-DCA with hydrogen and a catalyst sector and the

Although metal-based reduction has been found to be marginally successful in 1,2 DCA chemical dechlorination, catalytic hydrodechlorination (a dechlorination reaction in the presence of H₂ gas and a catalyst where chlorinated organic compounds are dechlorinated into benign compounds) has been reported to be successful in this respect (Vadlamannati et al 1999; Choi and Lee, 2000; Kovalchuk et al 2004; Orellana et al 2005). Here, the dechlorination reaction occurs between an organic molecule containing a C-Cl bond and gaseous H_2 in the presence of a highly selective metal catalyst, leading to the formation of HCl and a new C-H bond (Orellana et al 2005). It has been found that, for the process to be successful, the metal must be able to catalyze the dissociation of hydrogen (Kovalchuk et al 2004). According to Vadlamannati et al (1999), the catalyst participates in a catalytic cycle where the chlorinated molecules cover the metal surface. The reduction of highly reactive hydrogen adatoms remove the chlorine atoms and eventually form ethane (Vadlamannati et al, 1999). Group VIII-B metals (e.g. platinum, palladium, rhodium etc.) satisfy this condition and have shown good performance in the hydrodechlorination of 1,2-DCA forming ethane as end product (Fig 2.4) (Kovalchuk et al 2004; Vadlamannati et al 1999; Orellana et al 2005) when the temperature lies within the range of 200-250°C at a pressure of 101.325 kPa.

 $\begin{array}{c} CIH_2C-CH_2CI\\ H_2\\ H_2\\ CH_3-CH_3\\ H_2\\ H_2\\ CH_2=CH_2\end{array}$ 2H2 CH₃-CH₃ ←

Fig 2.5: Reaction Scheme for Catalytic Hydrodechlorination of 1,2-DCA (Ref: Kovalchuk et al, 2004)

Monochlorinated aliphatic hydrocarbons are also susceptible to catalytic hydrodechlorination; for example, methyl chloride dechlorination catalyzed by silica supported Ru, Rh, Pd, Pt, Ag lead to methane and hydrochloric acid (HCl) in the presence of H₂ (equation 2.16) (Fung and Sinfelt, 1987). Although Pd, Pt, Rh, Ni, Ru, Cu can effectively dechlorinate 1,2-DCA to ethane, the best performance of catalysts in terms of conversion of 1,2-DCA (~30%) has been found with Pd (Table 2.5).

remember of the art is will structure to a parallel an influence to a

onter des contrativos de la construcción de caráción de la defensa de la contra o contrapión de

 $H_4 + HCl H_2 \rightarrow CH_4 + HCl H_2 \rightarrow CH_4 + HCl He for the state of the first sector of (Equation 2.16) between the formula of the state of the formula of the state of the formula of the f$

Research	Temp (°C)	Catalyst	Conversion	Final Product
Group				
Heinrichs et al	350	Pd/SiO ₂	30%	90% Ethane
(1997)				
Vadlamannati	200	Pt/C	4%	92% Ethane, 2%
et al (1999)				Monochloroethane
Vadlamannati	200	Cu/C	0.6%	100% Ethylene
et al (1999)			м. А.	
Srebowata et al	230-250	Ni/SiO ₂	4%	70% Ethylene,
(2007)				30% Ethane
Srebowata et al	230-250	Ru/SiO ₂	2%	25% Ethylene,
(2007)				55% Ethane

Table 2.5: Summary of Catalytic Hydrodechlorination of 1,2-DCA

Therefore, it can be found that although the reductive treatment with nZVI particles failed to dechlorinate 1,2-DCA, catalytic hydrodechlorination in the laboratory at very high temperature, in the presence of a catalyst and gaseous hydrogen could transform 1,2-DCA into ethane and ethylene.

wells a second coupled decay repaired in Grand and App. 2010. The March

2.3.4 Modified Approach to Degrade 1,2-DCA with nZVI, Catalyst and Hydrogen So far, the beneficial application of zero-valent iron nanoparticles in aqueous phase for remediation of chlorinated hydrocarbons has been discoursed. However, 1,2-DCA has been found to be an exception in this respect due to its recalcitrance to reduction. On the other hand, gas-phase catalytic hydrodechlorination at an elevated temperature using an adequate catalyst could dechlorinate this compound. Nonetheless, the requirement of high temperature and gaseous H_2 makes this approach impractical for field remediation. Therefore, the study of an alternate methodology for in-situ 1,2-DCA degradation in contaminated aquifers is a worthwhile task. Since the main concept of hydrodechlorination to remediate 1,2-DCA lies in replacing the chlorine atoms by hydrogen, liquid phase reduction using an alternate source of hydrogen presents an interesting alternative to gas phase reduction of 1,2-DCA. This type of reduction has been favourably applied for dechloritation of polychlorinated biphenyles, chlorobenzenes and chlorotoluenes (Ukisu et al, 1997; 1998). In this case, relatively mild conditions with reaction temperatures rarely exceeding 100°C have been proved adequate for the dechlorination to take place. Moreover, instead of H₂ gas, various sources of hydrogen, like formic acid and formates, phosphinic acid and phosphinates, phosphorous acids and phosphates, alcohols, hydrazine or borohydrides, can be used (Ukisu et al 1998, Urbano and Marinas, 2001). The use of hydrogen donors has potential advantages on hydrogen transfer reduction compared with catalytic reduction with molecular H₂. The H₂ released from the hydrogen donors is transformed into atomic hydrogen (H[°]) onto the catalyst surface through catalyzed decomposition of H₂ (Zhou and Lim, 2010). This H^{*} reacts with chlorinated compounds, releases Cl- and forms hydrocarbons (Fig. 2.6) (Urbano et

al, 2001). According to Urbano et al. (2001), the hydrogen transfer mechanism may take place in various ways: the H_2 may serve as a proton, atom or hydride during its transfer from donor depending on the reagents and conditions.





When formic acid is used as the H_2 donor and Pd is used as catalyst to dechlorinate RCl, Kopinke et al (2004) proposed that, in the presence of Pd, RCl utilises directly the hydride hydrogen from the chemisorbed formate and breaks into RH and Cl⁻ (Fig 2.7a). In fact, under alkaline conditions, it is found that the relatively slow hydride mechanism dominates (Fig 2.7a), while at lower pH values the fast radical mechanism through H atoms is favoured under acidic or neutral conditions (Fig 2.7b) (Kopinke et al, 2004)



Fig 2.7: Possible mechanism of Pd-catalyzed hydroechlorination in the presence of formic acid a) hydride mechanism b) radical mechanism (Kopinke et al, 2004)

Literature reports indicate the combination of HCOOH, used as a H₂ donor, with Pd, is a powerful reductant. Under acidic or neutral conditions, formic acid was found to be as reactive as H₂. Kopinke et al (2004) observed that at 23°C and a pH = 3.5, 0.05mM chlorobenzene was completely dechlorinated using 5.5mM HCOOH and 250 mg/L Pd/Al₂O₃, forming benzene and chloride at the end of 20 hours. When isopropanol was used as hydrogen source, only 15% of chlorobenzene was converted to benzene after 2 weeks. The activity of the catalyst was tested and found to be reactive even after 2 weeks. Therefore, the partial dechlorination might be because the combination of Pd and

isopropanol was not an appropriate reductant under the applied mild conditions due to its kinetic nature and not because of the thermodynamic nature (the driving force for isopropanol reaction mechanism, $\Delta G^0 = -212$ kJ/ mol) (Kopinke et al, 2004). In another study, Ukisu et al (1997) used 2-propanol as hydrogen donor, for hydrogen transfer from 2-propanol to p-chlorotoluene in the presence of Rh/C, Pd/C and Pt/C at temperatures below 82°C. Although Pd has been proven to show the best catalytic properties in catalytic hydrodechlorination, Ukisu et al (1997) showed that, in this case, Rh exhibits better performance than Pd and the reaction was completed within 180 minutes while Pd showed only 20% degradation in that time period (Ukisu et al, 1998). Ukisu et al (1998) explained this difference in the dechlorination activity as a result of efficiency of hydrogen transfer process from 2-propanol on the catalyst surface.

Although nZVI itself performs as a source of hydrogen in aqueous phase for reduction of chlorinated solvents, it was completely unsuccessful to dechlorinate 1,2-DCA. Being encouraged by the idea of liquid phase reduction, along with nZVI, a suitable hydrogen donor and catalyst have been used in this research work to create a reducing environment that could degrade 1,2-DCA to a certain extent. The choice of hydrogen donor and catalysts were selected by analyzing the literature. Details of this selection to formulate different experimental conditions for 1,2-DCA has been resolved in the next chapter.

startel and party manager before no a la stage and elements with the state

2.4 SUMMARY

Nanoscale zero valent iron particles have been proposed as a preferred option for remediation of chlorinated solvents, such as chlorinated ethylenes and ethanes. The high surface area, nanoscale dimensions and high density of reactive sites on the surface enable nZVI particles to degrade contaminants at a faster and more complete manner than micro or macroscale iron systems. In aqueous environments, Fe⁰ is oxidized and this causes reduction of H₂O into H₂. Addition of noble metals to nZVI particles enhances the reactivity, the noble metals serving as catalysts. In a bimetallic nZVI system, the catalyst performs as the dissociator of H₂; absorbing and dissociating it into atomic H. This highly reactive form of hydrogen then attacks the adsorbed contaminants on the iron surface, replacing chlorine leading to the formation of benign products. However, in the absence of a catalyst, H₂ cannot contribute directly to the dechlorination. Moreover, the presence of a catalyst on the iron surface preserves the Fe⁰ core from being oxidized and thus maintains its reactivity for dechlorination. Both mono and bimetallic nZVI particles showed excellent performance in remediating chlorinated compounds like PCE, TCE, cis-DCE and VC. The final products were mainly saturated hydrocarbons. Although a number of chlorinated ethanes responded to the treatment with nZVI particles, 1,2-DCA was an exception because its chemical-physical properties did not allow it to respond positively towards the treatment involving nZVI. However, catalytic hydrodechlorination could successfully remediate this compound. The main strategy of this study is to create a nZVI based formulation combing the concept of catalytic hydrodechlorination and liquidphase reduction that will provide a positive methodology to dechlorinate 1,2-DCA successfully. elementes and alle contributions of the cost which elements.

(1) and sold for the second provide the second structure of the second structure of the second second second for the second s second s second seco

ference and the Archael to Lacron

2.5 REFERENCES

- Arnold, William A., William P. Ball, A. Lynn Roberts "Polychlorinated ethane reaction with zero-valent zinc" Journal of Contaminant Hydrology, 1999, 40, 183-200.
- Barnes, R.J, Olga Riba, M.N.Gardner, T.B.Scott and S.A. Jackman, "Optimization of nanoscale nickel/iron particles for the reduction of high concentration chlorinated aliphatic hydrocarbon solutions". <u>Chemosphere</u>, 2010, 79, 448-454.
- Bylaska, Eric, J, M. Dupuis and P.G.Tratnyek, "One electron transfer reactions of polychlorinated ethylenes: concerted and stepwise cleavages". J. Phys. Chem. A 2008, 112, 3712-3721
- Boyd, David, R, "The water we drink: an international comparison of drinking water standards and guidelines". Healthy Environment, Healthy Canadians Series, 2006, ISBN 1-897375-02-6.
- Choi, Young Heon and Wha Young Lee, "Effect of Ni loading and calcination temperature on catalyst performance and catalyst deactivation of Ni/SiO2 in the hydrodechlorination of 1,2-dichloropropane into propylene". <u>Catalysis Letters</u>, 2000, 67, 155-161
- De Wildeman, S., W. Verstraete, "The quest for microbial reductive dechlorination of C2 to C4 chloroalkanes is warranted". <u>Appl Microbiol Biotechnol</u>, 2003, 61, 94-102.
- Dyer, Mark, Erwin Van Heiningen, Jan Gerritse, "In situ bioremediation of 1,2-DCA under anaerobic condition". <u>Geotechnical and Geological Engineering</u>, 2000, 18, 313 334.
- Elliott, D.W. and Wei-xian Zhang, "Field Assessment of nanoscale bimetallic particles for groundwater treatment". <u>Environmental Science and Technology</u>, 2001, 35, 4922-4926.
- Feng, Jing nad Teik-Thye Lim, "Pathways and kinetics of carbontetrachloride and chloroform reductions by nanoscale Fe and Fe/Ni particles: comparison with commercial micro-scale Fe and Zn". Chemosphere 59 (2005) 1267-1277.

Field, J.A., R.Sierra-Alvarez, "Biodegradability of chlorinated solvents and related chlorinated compounds". <u>Eurochlor</u>, 2004.

Fung, S.C., J.H.Sinfelt, "Hydrogenolysis of methyl chloride on metals". Journal of <u>Catalysis</u>, 1987, 103, 220-223.

Health Canada, www.hc-sc.gc.ca/

- He, Feng, Dongye Zhao, "Manipulating the size and dispersibility of zero-valent iron nanoparticles by use of carboxymethyl cellulose stabilizers" <u>Environmental</u> <u>Science & Technology</u>, 2007, 41, 6216 6221.
- He, Feng, Dongye Zhao, "Preparation and characterization of a new class of starchstabilized bimetallic nnanoparticles for degradation of chlorinated hydrocarbons in water". Environmental Science & Technology 2005, 39, 3314 - 3320
- He, Feng, Dongye Zhao, "Stabilization of Fe-Pd nanoparticles with CMC for enhanced transport and dechlorination of TCE in soil and groundwater". <u>Ind. Eng.</u> <u>Chem</u>. Res, 2007, 46, 29-34.

Charles Street and Alexandra State

39

and Handson - David and the base

energine of Musich Breshold, in the Center

- Heinrichs, B., Delhez, P., Schoebrecht, J.P. and Pirard, J.P., "Pd-Ag sol gel catalysts for selective hydrodechlorination of 1,2-DCA to ethylene" Journal of Catalysis, 1997, 172, 322 1997.
- Johnstone R.A.W and Anna H. Wilby, "Heterogeneous catalytic transfer hydrogenation and its relation to other methods for reduction of organic compounds". <u>Chem. Rev</u>. 1985, 85, 129-170
- Kim, Y-H, Elizabeth R Carraway, "Reductive dechlorination of TCE by zero-valent bimetals" <u>Environmental Technology</u>, 2003, 24, 69-75.
- Klecka, G.M, C.L. Carpenter, S.J.Gonsior, "Biological transformations of 1,2-DCA in sub-surface soils and groundwater" Journal of Contaminant Hydrology, 1998, 34, 139 154.
- Kopinke Frank-Dieter, Katrin Mackenzie, Koechler, Robert Georgi Anette, "Alternative sources of H₂ for hydrodechlorination of chlorinated organic compounds in water on Pd catalysts". <u>Applied Catalysis A: General, 2004, 271,</u> 119-128.
- Kovalchuk, Vladimir I., Julie d'Itri, "Catalytic chemistry of chloro- and chlorofluorocarbon dehalogenation: from macroscopic level understanding".
 Applied Catalysis A: General, 2004, 271, 13-25.
- Larson, A.Richard and Eric J. Weber, "Reaction Mechanisms in Environmental Organic Chemistry" Lewis Publishers, 1994
- Lodolo, Andrea, "Further description: remediation option". <u>EUGRIS: Portal for Soil and</u> <u>Water Management in Europe</u>, 2007.
- Li, L., M.H. Fan, R.C. Brown, J.H. Van Leeuwen, J.J. Wang, W.H. Wang, Y.H. Song, and P.Y. Zhang, "Synthesis, properties, and environmental applications of nanoscale iron-based materials: a review". <u>Critical Reviews in Environmental Science and Technology</u>, 2006, 36, 405-431.
- Li, X.Q., D.W. Elliott, and W.X. Zhang, "Zero-valent iron nanoparticles for abatement of environmental pollutants: Materials and engineering aspects". Critical Reviews in Solid State and Materials Sciences, 2006, 31, 111-122.
- Lien, Hsing-Lung, Wei-Xian Zhang, "Hydrodechlorination of chlorinated ethanes by nanoscale Pd/Fe bimetallic particles". Journal of Environmental Engineering, 2005, 131.
- Lien, Hsing-Lung, Wei-Xian Zhang, "Nanoscale iron particles for complete reduction of chlorinated ethenes" <u>Colloids and Surfaces A: Physiochemical & Engineering</u> Aspects ,2001, 191, 97-105.
- Lin, Ching Jung, Shang Lien Lo, Ya Hsuan Liou, "Dechlorination of TCE in aqueous solution by noble metal modified iron" Journal of Hazardous Materials B, 2004, 116, 219-228.
- Liu, Y.Q., H. Choi, D. Dionysiou, and G.V. Lowry, "Trichloroethene
 - hydrodechlorination in water by highly disordered monometallic nanoiron". <u>Chemistry of Materials</u>, 2005a, 17, 5315-5322.
- Liu, Y.Q., S.A. Majetich, R.D. Tilton, D.S. Sholl, and G.V. Lowry, "TCE dechlorination rates, pathways, and efficiency of nanoscale iron particles with different properties". <u>Environmental Science & Technology</u>, 2005b,9, 1338-1345.
 Matheson, Leah J., P.G.Tratnyek, "Reductive dehalogenation of chlorinated methanes
- by iron metal". <u>Environmental Science & Technology</u>, 1994,28, 2045-2053.

- Marzorati, Massimo, Sara Borin, Lorenzo Brusetti, Daniele Daffonchio, Caterina Marcilli, Giovanna Carpani, Francesca de Ferra, "Response of 1,2dichloroethane-adapted microbial communities to ex-situ biostimulation of polluted groundwater". <u>Biodegradation</u>,2005, DOI 10.1007/ s 10532-005-9004-z Nobre, R.C.M, M.M.M. Nobre, "Natural attenuation of chlorinated organics in a shallow sand aquifer", Journal of Hazardous Materials, 2004, 110, 129-137.
- Nurmi, J.T., P.G. Tratnyek, V. Sarathy, D.R. Baer, J.E. Amonette, K. Pecher, C.M. Wang, J.C. Linehan, D.W. Matson, R.L. Penn, and M.D. Driessen, "Characterization and properties of metallic iron nanoparticles: Spectroscopy, electrochemistry, and kinetics". <u>Environmental Science & Technology</u>, 2005, 39, 1221-1230.
- Ordonez, Salvador, Herminio Sastre, Fernando V. Diez "Hydrodechlorination of aliphatic organochlorinated compounds over commercial hydrogenation catalysts" Applied Catalysis B: Environmental, 2000, 25, 49 – 58.
- Orellana, F., G. Pecchi, P. Reyes, "Selective hydrodechlorination of 1,2-DCA over Pd-Sn/SiO2 catalysts". J. Chil. Chem. Soc, 2005, 50.
- Ponder, S.M., J.G. Darab, and T.E. Mallouk, "Remediation of Cr(VI) and Pb(II) aqueous solutions using supported, nanoscale zero-valent iron". <u>Environmental</u> <u>Science & Technology</u>, 2000, 34, 2564-2569.
- Ponder, S.M., J.G. Darab, J. Bucher, D. Caulder, I. Craig, L. Davis, N. Edelstein, W. Lukens, H. Nitsche, L.F. Rao, D.K. Shuh, and T.E. Mallouk, "Surface chemistry and electrochemistry of supported zerovalent iron nanoparticles in the remediation of aqueous metal contaminants". <u>Chemistry of Materials</u>, 2001, 13, 479-486.
- Rhodes W.D, K. Lazar, V. I Kovalchuk, J. d'Itri, "Hydrogen-assisted 1,2dichloroethane dechlorination catalyzed by Pt-Sn/SiO2: Effect of the Pt/Sn ratio". Journal of Catalysis, 2002, 211, 173-182.
- Sakulchaicharoen, Nataphan, D.M.O'Carroll and Jose .E. Herrera, "Enhanced Stability and dechlorination activity of pre-synthesis stabilized nanoscale FePd particles". Journal of Contaminant Hydrology. 2010, 118, 117-127.
- Saleh, Navid, K. Sirk, Y. Liu, T. Phenrat, B.Dufour, K.Matyzasewski, R.D.Tilton, G.V.Lowry, "Surface modifications enhance nanoiron transport and NAPL targeting in saturated porous media" Environ. Eng. Sci, 2007, 24, 45-57.
- Schrick, B., J.L. Blough, A.D. Jones, and T.E. Mallouk, "Hydrodechlorination of trichloroethylene to hydrocarbons using bimetallic nickel-iron nanoparticles". <u>Chemistry of Materials</u>, 2002, 14, 5140-5147.
- Song Hocheol, Elizabeth R Carraway, "Catalytic hydrodechlorination of chlorinated ethenes by nano-scale zero-valent iron" <u>Applied Catalysis B: Environmental</u> 2008, 78, 53-60
- Song, Hocheol, Elizabeth R Carraway, "Reduction of chlorinated ethanes by nanoscale zero-valent iron: kinetics, pathways and effects of reaction conditions". Environmental Science and Technology, 2005, 39, 6237-6245.
- Srebowata, Anna, M. Sadowska, W. Juszczyk, Z. Kaszkur, Z. Kowalczyk, M. Nowosielska, Z. Karpinski, "Hydrogen-assisted dechlorination of 1,2dichloroethane over silica-supported nickel-ruthenium catalysts". <u>Catalysis</u> <u>Communications</u>, 2007, 8, 11-15.

- Tratnyek. Paul G, Richard L Johnson, "Nanotechnologies for environmental cleanup". <u>Nanotoday</u>, May 2006, Volume 1 Number 2
- Ukisu, Yuji, Satoshi Kameoka, Tatsuo Miyadera, "Rh-based catalysts for catalytic dechlorination of aromatic chloride at ambient temperature" <u>Applied Catalysis B:</u> <u>Environmental</u>, 1998, 18, 273-279.
- United States Environmental Protection Agency, "Nanotechnology for site remediation : Fact sheet", Solid Waste and Emergency Response (5203P), 2008, EPA 542-F-08-009.
- Urbano, F.J., J.M. Marinas, "Hydrogenolysis of organohalogen compounds over palladium supported catalysts". Journal of Molecular Catalysis A: Chemical 2001, 173, 329 345.
- Vadlamannati, Lalith S., Vladimir I. Kovalchuk, Julie d'Itri, "Dechlorination of 1,2-DCA catalyzed by Pt-Cu/C: unravelling the role of each metal". <u>Catalysis Letters</u>, 1999, 58, 173 – 178.
- Vanstone, Nancy, M. Elsner, G. Lacrumpe-Couloume, S. Mabury, B.S. Lollar, "Potential for identifying abiotic chloroalkane degradation mechanisms using carbon isotope fractionation" <u>Environmental Science & Technology</u>, 2008, 42, 126-132.
- Wang, C.B. and W.X. Zhang, "Synthesizing nanoscale iron particles for rapid and complete dechlorination of TCE and PCBs". <u>Environmental Science & Technology</u>, 1997, 31, 2154-2156.
- Wang, Xiangyu, Chao Chen, Ying Chang, Huiling Liu, "Dechlorination of chlorinated methanes by Pd/Fe bimetallic nanoparticles" Journal of Hazardous Materials 2009, 161, 815-823.
- Wang, Xiangyu, Chao Chen, Huiling Liu, Jung Ma, "Characterization and evaluation of catalytic dechlorination activity of Pd/Fe bimetallic nanoparticles" <u>Ind. Eng.</u> <u>Chem. Res</u> 2008, 47, 8645-8651.
- Yan, Weile, Andrew A. Herzing, Xiao-Qin Li, C.J.Kiely, Wei-Xian Zhang, "Structural evolution of Pd-doped nanoscale zero-valent iron (nZVI in aqueous media and implications for particle aging and reactivity". <u>Environmental Science</u> <u>& Technology</u>, 2010, 44, 4288-4294.
- Zaan, Bas Van der, Jasperien de Wart, Huub Rijnaarts, , Willem M. de Vos, Kauke Smidt, Jan Gerritse, "Degradation of 1,2-DCA by microbial communities from river sediment at various redox conditions". <u>Water Research</u>, 2009,43, 3207-3216.
- Zhang, W.X., "Nanoscale iron particles for environmental remediation: An overview". Journal of Nanoparticle Research, 2003, 5, 323-332.
- Zhang, W.X., C.B. Wang and H.L. Lien, "Treatment of chlorinated organic contaminants with nanoscale bimetallic particles", <u>Catalysis Today</u>, 1998, 40, 387-395.
- Zhang, Wei-Xian, Daniel W Elliot, "Application of iron nanoparticles for groundwater remediation" <u>Remediation Spring 2006</u>.
- Zhou, Jao, Yaozhong Li and Teik-Thye Lim,"Catalytic hydrodechlorinaion of chlorophenols by Pd/Fe nanoparticles: Comparisons with other bimetallic systems, kinetics and mechanisms". <u>Separation and Purification Technology</u>, 2010, 76, 206-214.

3.0 DEGRADATION OF 1,2-DCA WITH NZVI PARTICLES

du Lorense open of Dully and repairly and themselv dealers and without du

3.1 INTRODUCTION

Chlorinated Organic Compounds (COCs) are difficult to remediate directly under most natural or abiotic conditions and are persistent in the environment for decades to centuries, causing widespread contamination in both groundwater and soil. The application of nanoscale zero-valent iron particles (nZVI) for abiotic remediation of chlorinated aliphatics, aromatics and polychlorinated biphenyls is a promising technology

(Zhang et al, 1998; He and Zhao, 2005; Li et al, 2006; Wang et al 2008; Sakulchaichareon et al, 2010). Nanoscale zero-valent iron particles-are very effective electron donors possessing a standard reduction potential (E_0) of - 0.44 V. In an aqueous solution, Fe⁰ is oxidized to Fe²⁺ and forms H₂ by reduction of H₂O (equations 3.1 and 3.2). When iron particles are doped with a catalytic metal (e.g., Pd, Pt, Ni or Ag), the catalyst adsorbs the H₂ and dissociates it into atomic H (equation 3.3) which then attacks the adsorbed chlorinated contaminant (RCl in equation 3.4) on the iron surface, leading to dehalogenation of the organic substrate (equation 3.4).

 $Fe^{0} + 2H_{2}O \rightarrow Fe^{2+} + H_{2} + 2OH^{-} \qquad (Equation 3.1)$ $2Fe^{0} + O_{2} + 2H_{2}O \rightarrow 2Fe^{2+} + 4OH^{-} \qquad (Equation 3.2)$ $H_{2} \xrightarrow{\text{catalyst}} 2H^{\bullet} \qquad (Equation 3.3)$ $Fe + H_{2}O + H^{\bullet} + RCl \longrightarrow RH + HCl + FeO_{x} \qquad (Equation 3.4)$

Many COCs of concern (e.g., tetrachloroethylene PCE, trichloroethylene TCE, cisdichloroethylene cis-DCE) are rapidly and completely dechlorinated without the formation of toxic by-products (e.g., vinyl chloride) using nZVI doped with a catalytic metal known as bimetallic nZVI systems (He and Zhao, 2005; Lien and Zhang, 2001). Both mono and bimetallic nZVI particles exhibited a good performance in degradation of chlorinated ethanes like hexachloroethane (HCA), pentachloroethane (PCA), 1,1,2,2tetrachloroethane (1,1,2,2-TeCA) and 1,1-dichloroethane (1,1-DCA) forming ethane at the end of the reaction (Lien and Zhang, 2005; Song and Carraway, 2005).

ndrie Plane an ind galaga dan manin de type el concernificant las de danseters

44

Although chlorinated ethanes like HCA, PCA, 1,1,2,2-TeCA and 1,1-DCA showed response to nZVI dechlorination, 1,2-dichloroethane (1,2-DCA) has been resistant to degradation using nZVI technology (Ferrey et al. 2004; Zhang et al 1998; Song and Carraway 2005; De Wildeman and Verstraete 2003, Kopinke et al 2004). For instance, Song and Carraway (2005) found no measurable reduction of 1,2-DCA (conversion below 5%) even after 40 days of reaction (nZVI/1,2-DCA=83 wt ratio). The limited reactivity of nZVI to 1,2-DCA has been ascribed to factors such as bond strength of the carbon-chlorine bond, the low electron affinity of the carbon-chlorine bond and the low stability of the carbon-radical species resulting from the initial electron transfer step required for dechlorination (Larson and Weber, 1994). Generally, the higher the oxidation state of the carbon atom, the more responsive it is to reduction and the more resistant to oxidation (De Wildeman and Verstraete, 2003). 1,2-DCA with an oxidation state of -1 exhibits a low tendency to reduction under conventional conditions.

na Maline ang kana ing Kanali kanali ka dipuning sa ana bana an dan si a**n kaping muta ang Kaping C**anada na m

As discussed earlier, in the nZVI system the source of H₂ is water. Hydrogen generation occurs through Fe⁰ corrosion and this hydrogen is used during the transformation of the chlorinated compounds into benign products. Therefore, introducing a more efficient H₂ source in the dechlorination system could potentially enable a more deffective dechlorination process. Previous studies have shown that a H₂ source coupled with an active hydrogenation catalyst (such as, Pd, Rh, Ni or Pt) can reduce reticent organic halides such as- chlorotoluene, chlorobenzene, and polychlorinated biphenyls. (Johnstone and Wilby, 1985, Ukisu et al 1998; Kopinke et al 2004). The adsorption of hydrogen released from the hydrogen donors onto the catalyst surface influences the dechlorination rate (Urbano et al, 2001). Formic acid has been found as an effective hydrogen source and a combination of formic acid and Pd acts as a powerful reductant to dechlorinate compounds like chlorobenzene (Kopinke et al, 2004). Alcohols have been proposed as hydrogen sources as well. In this case, the alcohol acts both as a hydrogen source and solvent for the dechlorination reaction. However, in this case, the addition of a base (e.g. NaOH, KOH, NH₄OH) is necessary to neutralize the HCl formed because HCl can deactivate the catalysts. (Urbano and Marinas, 2001; Ukisu et al, 1998). The use of alcohols as hydrogen donors coupled with a hydrogenation catalysts such Rh, Pd and Pt has been proved successful in the dechlorination of contaminants like p-chlorotoluene (Ukisu et al 1998). Nonetheless, the literature lacks reports on the use of these systems for the dechlorination of 1,2 DCA. Nov they workly is space proceeding, (1 20 Cheeve S

The studies discussed in Chapter 2 and briefly discussed in this section indicate that, while thermodynamically feasible, the right conditions for dechlorination of 1,2 DCA are

ee particulty with the active suggest Stress Actives (Article (2048), ACROS)

yet to be found. Based on results observed on the dechlorination of other organic species, it can be hypothesized that a combination of liquid-phase reduction using hydrogen donors, a catalyst and nZVI should be able to initiate the degradation of 1,2-DCA under the right experimental conditions. In our study formic acid and isopropanol (IPA) were selected as hydrogen donors while Rh and Pt along with Pd were used as catalysts. The selection of these specific systems were based on reports indicating that both Rh and Rh-Pt catalysts showed significant performance in dechlorinating chlorotoluene in the presence of 2-propanol and NaOH (Ukisu et al, 1998) whereas HCOOH catalyzed by Pd was as reactive as gaseous H₂ to completely degrade chlorobenzene (Kopinke et al, 2004). The objectives of this study are i) to find the nZVI-based formulation to dechlorinate 1,2-DCA under aqueous conditions similar to those in the field. ii) to optimize the experimental conditions for this process and iii) to investigate the degradation pathways under which 1,2-DCA dechlorination takes place.

3.2 MATERIALS AND METHODS

3.2.1 Chemicals

The following chemicals were used for experiments as received: 1,2-DCA(>99%, A.C.S. Reagent, Sigma-Aldrich), TCE (99+%, extra pure, stabilized, ACROS Organics), nhexane (>95% GC, Fluka Analytical), formic acid (98%, GR ACS, EMD Chemicals Inc), isopropyl alcohol (for GC, HPLC, Residue Analysis, Spectophotometry; EMD Chemicals Inc), FeSO₄.7H₂O (99+%, A.C.S. Reagent, Sigma-Aldrich), NaBH₄ (98+%, ACROS), K₂PdCl₆ (99%, ACROS Organics), Pd-acetate (99.98%, trimer, Pd min 47%; Alfa Aesar), Rh-acetate (dimmer, 99.99%; Alfa Aesar), PtBr₂ (99.9%, Pt min 54.5%; Alfa Aesar),

n 1919 i la coma contributa di Mala di coma dal adale da la dificia da la Romana da la Romana da na danima.

sodium carboxymethyl cellulose (90K, Sigma-Aldrich), sodium chloride (EMD Chemicals Inc), Scotty Analyse Gases, N₂ (ultra high purity, PRAXAIR), Gas Mix (5% H₂ balance Ar, PRAXAIR), He (PRAXAIR), Compressed air (PRAXAIR), 5% CH₄ balance Ar (PRAXAIR).

3.2.2 Synthesis of non-stabilized nZVI Particles

The set of the set of the second set of the

The non-stabilized nZVI particles were synthesized using NaBH₄ to reduce an aqueous solution of FeSO₄.7H₂O following the protocol of Schrick et al (2002). The synthesis procedure was carried out in an anaerobic chamber to avoid Fe oxidation. In order to remove dissolved O₂, deionized (DI) water used for the experiments was purged with purified N₂ for at least 2 hours before synthesis. FeSO₄.7H₂O was added to DI water to achieve 0.161M iron. The reaction mixture was homogenized using a magnetic stirrer operating at 600 rpm. After complete dissolution of the iron salt, an equal volume of 0.322M aqueous solution of NaBH₄ was added drop wise (1drop/ sec) to the iron solution to reduce ferrous iron to its zero-valent form (Equation 3.5):

 $2Fe^{2+} + BH_4 + 3H_2O \rightarrow 2Fe^0 + H_2BO_3 + 4H^+ + 2H_2$ (Equation 3.5)

相关的 化合理 建立方式 医多子肌的 使人的 的复数使用的复数 化乙基酸

Upon addition of the NaBH₄ solution, the clear yellowish solution rapidly changed to a dark black suspension, indicating the formation of Fe nanoparticles. After complete addition of NaBH₄, the solution was mixed for about 30 minutes until H₂ gas evolution ceased. The resulting nZVI particles were thoroughly rinsed with DI water (100 mL/g).

In order to prepare Pd/Fe bimetallic nanoparticles, an aqueous slurry of Fe nanoparticles was equilibrated with an 3.38×10^{-6} M ethanolic solution of $[Pd(C_2H_3O_2)_2]_3$. This mixture was stirred for 30 minutes. This resulted in the reduction and subsequent deposition of Pd on the iron surface (equation 3.6):

 $Pd^{2+} + Fe^{0} \rightarrow Pd^{0} + Fe^{2+}$ (Equation 3.6)

The palladium-doped nanoiron particles were then washed with DI water, collected in 120 mL reaction vials and capped with Teflon Mininert Valves.

Transmission Electron Microscopy (TEM) images give information about the shape, size and size distribution of particles. A very dilute solution of Pd/Fe nZVI particles was prepared in acetone or IPA in order to take their TEM images. This solution was then dispersed for $20 \sim 30$ minutes. On a copper lacey carbon grid, a few drops of nZVI solution was deposited, stored in the fume hood until the acetone or IPA is completely evaporated from the grid and then it was analyzed by the TEM. TEM analysis was performed using a FEI Titan 80-300 Cryo-in situ Transmission Electron Microscopes. Fig 3.1 shows the TEM image of Pd/Fe bimetallic particles.

he menode administration of the least of the least of the last the last the last of the last sector of the

achung but the added to show of De relegandeles and they collected following the

sour parachies to complete the first of the background



Fig 3.1 TEM image of Pd/Fe bimetallic nZVI particles with scale bar 100 nm. (Fig Courtesy: Nataphan Sakulchaicharoen) '

For the preparation of Rh/Fe bimetallic particles, Rh-acetate ($C_8H_{16}O_8Rh_2$) dissolved in DI water (4.82 x 10⁻⁴ M) was added to the aqueous slurry of Fe particles and then equilibrated for 30 minutes. In some experiments, a mixture of catalyst precursors (Rh-Pt and Pd-Rh-Pt) was used following a co impregnation protocol. PtBr₂ served as the precursor of Pt. However, since this compound is insoluble in pure water, it was first dissolved in a 1M NaBr solution (through comlexation by Br⁻ ions) and then mixed with the aqueous solution of Rh-acetate (Rh-Pt/Fe) and/ or Pd-Rh-acetate (Pd-Rh-Pt/Fe). This mixture was then added to slurry of Fe nanoparticles and then collected following the same procedure described above for the Pd-Fe material.

3.2.3 Synthesis of CMC-stabilized nZVI Particles

The synthesis procedure was carried in an anaerobic chamber. CMC was added to the solution mixture before the reduction step with borohydride. Before synthesis, both DI water and 3% CMC (90K) stock aqueous solution were purged with purified N_2 for at least 2 hours to remove dissolved O_2 .

A volume of 50mL of an aqueous solution of FeSO₄.7H₂O (0.161M) prepared in DI water was added to 125 mL of 3% aqueous solution of CMC to give a final concentration of 1.5% of CMC. This mixture was stirred continuously with a mixer for about 15 minutes to complete the formation of the Fe-CMC complex where the carboxylic groups of CMC, after being almost fully disassociated, interacted with the Fe²⁺ cations. After this period, 50mL of an aqueous NaBH₄ solution (0.322M) was added drop wise (1drop/ sec) to the CMC-Fe²⁺ solution. After addition of the total volume NaBH₄, the mixture was allowed to homogenize for about 30 more minutes and then 25 mL of 2.35 x 10⁻³ M of K₂PdCl₆ dissolved in a 0.005M NaCl solution was added to the CMC-nZVI solution in order to prepare bimetallic nanoparticles. Pd was deposited on the nZVI surface according to equation 3.7:

$PdCl_6^{2^{\circ}} + 2Fe^0 \rightarrow Pd^0 + 2Fe^{2+} + 6Cl^{\circ} \qquad (Equation 3.7)$

teres histori di laginge sitar ayang raga anala kining teran berajar datara kamentanan

evandersel waa baajang beshare haad parella. Uniterate as a figti 12020.

The synthesized iron particles were collected in a 120 mL reaction vial and were immediately capped with Teflon Mininert Valves. Fig 3.2 shows the TEM image of CMC (90K) stabilized Pd/Fe nZVI particles.



Fig 3.2 TEM image of CMC (90K) stabilized Pd/Fe bimetallic nZVI particles with scale bar 5 nm (Fig Courtesy: Nataphan Sakulchaicharoen)

3.3 BATCH EXPERIMENTS

Batch experiments were carried out in 120 mL amber bottles to test the reactivity of the laboratory synthesized nanosized zero-valent iron particles for the dechlorination of 1,2-DCA and TCE. The purpose of using amber bottles was to restrict any kind of photodegradation of the chlorinated solvents. Along with the reaction vials, blank experiments were conducted without metal particles. Stock solutions of both 1,2-DCA and TCE were prepared in isopropanol and appropriate volumes of the stock solutions were spiked through gas-tight syringes in both vials to achieve the desired concentration of the chlorinated solvents. In those experiments requiring formic acid, an appropriate amount of this chemical was added to the vials before 1,2-DCA spiking. About 25~50% of inert headspace was maintained in the vials to allow for good homogenization of the

Table 3	.1 indicates the	different	experimental	conditions	for 1,2-DCA and	TCE
lechlorin	nation testing.			n na sana na s Sana sana sana sana sana sana sana sana		
		international and a second se			n territoria da successa en la seconda successa en la seconda en la seconda en la seconda en la seconda en la s Seconda en la seconda en la Seconda en la seconda en la	
n an						
and a second			n an an an ann an an an an an an an an a			
		in in the second se				
• • •	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2			and a second		n - 1 2 2 2 2 2 2 3 3 3 3 3 3 3 3 3 3 3 3 3
2 2		a and a second s				
• • • • • • • • • • • • • • •	-	a A A A A A A A A A A A A A A A A A A A		۲۰۰۰ ۲۰۰۰ ۲۰۰۰ ۲۰۰۰ ۱۹۰۰ - ۲۰۰۰ - ۲۰۰۰ ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰ - ۲۰۰۰		
	politikan karata Politikan Step Han Politikan		and Alabas Alabas Alabas	•		
n an generation of the second se		in a second seco				i i i i i i i i i i i i i i i i i i i
· · · ·						· · ·
in in the state of						

Exp #.	Chlorinated	Fe ⁰ conc	Catalyst	Temp	H ₂ donor /	Degradation
	Compound	(g/L)	(wt $\%$ of Fe ⁰)	(°C)	Stabilizer added	.after 7 days
$\sim \sum_{i=1}^{n-1} g_i^{(i)} + g_i^{(i)} g_i^{(i)} - g_i^{(i)} g_i^{(i)} - g_i^$	(mg/L)			ALCON BE AN		n konte
1.	TCE=100	r.5	none	25	none	25% after 6
		n gelfter i	aparta sa Nasa	All the second	nde ander ander en de service	days
2.	TCE= 100	1.5	Pd=0.1%	25	none	100% after
101. 301.0	21、880-300-3440-244 <u>4</u> -3	$\lambda = \sum_{i=1}^{N-1} \left(\lambda_{i} \lambda_{i} \right) = \left(\sum_{i=1}^{N-1} \left(\lambda_{i} \lambda_{i} \right) \right)$	an tha an that th	dan etki te	ale a la calendaria da c	3 hours
3.	1,2-DCA= 50	1.5	Pd=0.5%	25	none	none
	es en trates Novalis.			전화가 관람		
4.	1,2-DCA= 5	20	Pd=0.2%	25	none	7%
A REAL PROPERTY AND	a nivela la secta de secta.	Alter BBare			et kan de state de la seconda de la secon Seconda de la seconda de la Seconda de la seconda de la	Shahuzzho
5.	1,2-DCA=10	20	Pd=0.2%	25	none	8%
	Net and the second s					i se a comerciano de la comerciana de la co
6.	1,2-DCA=5	2.5	Pd=0.2%	25	HCOOH=	7%
					250 mg/L	
7	1,2-DCA= 10	5	Pd=0.2%	35	HCOOH=	11%
	grawn yn yn cae de ywyn.			a dia mandri dia minina	500 mg/L	-
8.	1,2-DCA=10	10	Rh=0.5%	35	IPA and NaOH	none
9,	1,2-DCA=10	10	Rh=0.2%, Pt	35	IPA and NaOH	none
			= 0.1%			
		10	D1 0 P	07		
10.	1,2-DCA=10	10	Rh=0.7%,	35	IPA and NaOH	none
2.34 97	t glera trongt	1 - A - A - A - A - A - A - A - A - A -	Pt = 0.35%		Re La Galacia de la A	
11	10.004 10	10	D1 1 4007	25		
	1,2-DCA=10	IU A A A A A A A A A A A A A A A A A A A	Rn=1.42%	. 33	IPA and NaOF	none
10	10004 10	10	Pl = 1.42%	25		1007
12.	1,2-DCA=10	10 States a tradition	Rn=0.2%	33	HCOOH=	12%
			Pu=0.2%		500mg/L	
12 .	10004 10		Pl = 0.1%	25		007-
13.	1,2-DCA=10	3	Pu=0.5%	23	1000 me/I	970
	1 CE = 10 (alter $7 down)$		geographics of f	ng ang kang kang sang sang sang sang sang sang sang s	CMC-15%	
14	TCE-10	5	Pd-0.5%	25	HCOOH-	not detected
14.		a serie de series		20	500 mg/I	
					CMC=1.5%	
15	1.2 - DCA - 10	5	Pd=0.5%	45	HCOOH=	17%
1.5.	1,2-DCA-10	5	1 4-0.570		500 mg/I	1110
1.00				in the second second	CMC=1.5%	·
16	12 - DCA - 10	10	Pd=0.2%	35	HC00H=	12%
10.	1,2-10-10		1 4-0.270		500mo/I	
1	1	1	1	1	1	J

 Table 3.1: Experimental Conditions for 1,2-DCA and TCE Dechlorination

3.4 ANALYTICAL METHODS

3.4.1 Solvent Extraction Analysis

The concentrations of 1,2-DCA and TCE were analyzed using the Solvent Extraction method. n-Hexane was used as the extracting solvent. At selected time intervals, 0.25~1 mL of an aqueous aliquot was collected from both the reaction and blank vials and transferred to a 2-mL GC vial containing 1 mL of n-Hexane. This GC vial was shaken for $g_{(\mu)}$ about 10 seconds and then kept undisturbed for about 2 hours, to allow for equilibration. After this time, the organic phase was removed for GC analysis. A 1 µL of extract was withdrawn by the autosampler for GC analysis.

The sample was analyzed for the determination of the concentration of 1,2-DCA and TCE using an Agilent 7890 Gas Chromatograph equipped with a DB-624 capillary column (75m x 0.45 mm x 2.55 μ m) and an Electron Capture Detector (ECD). The temperature program used for the analysis was: 35°C for 12 min, then 5°C / min to 60°C held for 1 min; 17°C/ min to 200°C held for 5 min. A sample volume of 1 μ L was injected in split less mode. Standard calibration curves for 1,2-DCA and TCE were used to calculate the aqueous concentration of the solvents in the reaction mixture. The standards were prepared using isopropanol and DI water and were analyzed using the same method as the samples. The retention time of 1,2-DCA and TCE found were about 15.77 min and 18.21 min respectively for this temperature program. Along with 1,2-DCA and TCE, 1,2-DCE was also quantified during the analysis for identification of the formation of chlorinated intermediates during the dechlorination reaction.

the predentities of character along the constant energy is along an it, which has the first an

an advect to be been to a new the due may not the reproduct be the product be been been been and

Grandiad

3.4.2 Headspace Analysis management of the holds and the destantion of

Static headspace analysis was done to detect the hydrocarbons produced during dechlorination reactions on an Agilent 7890 GC equipped with a 7693 Auto sampler, a Mass detector (5975C MSD) and Flame Ionization Detector (FID). A GS-Gas Pro Column (3.0m x 320 µm) was used for this analysis. The temperature program used under this program was 35°C for 5 min, then 10°C / min to 220°C held for 7 min. Sample injection was carried out using both manual injection and auto-sampling. For the case of manual injection, a 250µL gas-tight syringe was used to take the gaseous sample from the headspace of the reaction and blank vial. For the case of auto sampling, two different methods were used. In the first case, 1 mL of aqueous sample withdrawn from the reaction mixture was put into a 2mL GC vial. This sample was heated at 50°C for 1 min and stirred at a speed of 1000 rpm before a 250µL headspace sample injection into the GC. The second case involved 2 mL of liquid sampling from the reaction mixture into a 10 mL autosampler GC glass vial. This was followed by heating the vial at 85°C for 28 min and stirring at 1500 rpm followed by injection of a 2.50 mL headspace sample to the GC. The hydrocarbons were identified based on their mass spectra. Quantification was carried out by performing a calibration using a gaseous mixture of ethane, propane, butane, pentane, hexane, ethylene, propylene, 1-butylene, 2-butylene, pentylene and 1hexylene of known concentration in nitrogen.

3.4.3 Chloride Analysis () and find of the second decaded and for the second decaded to the

The production of chloride from the dechlorination reaction of 1,2-DCA and TCE was monitored for selected experiments. For this purpose, the aqueous phase left in the 2 mL

能是原则,这些方面把这些活动。他们,他们们都是自己的命令的性情的。""你们不知道,你不知道

GC vial after n-Hexane extraction was used. The evolution of chloride was monitored using a Waters 717 plus Autosampler High Performance Liquid Chromatograph (HPLC) equipped with a conductivity detector and an IC pakTM anion column (4.6m x 50 mm). An isocratic program was used. The eluant used was a mixture of 12% acetonitrile in deionized water and the flow rate was set to 1.2 mL/min at a pressure of 595 psi. The injection volume was 100 μ L. The calibration for chloride evolution was done using different standards of NaCl solution (0.01, 0.05, 0.5, 1 and 5 mM).

a na Alexandra (a carabet ha e a bab patampalemente (Carbo de pa

3.5 RESULTS AND DISCUSSION

3.5.1 The Formulation for 1,2-DCA Dechlorination

Before attempting 1,2-DCA dechlorination, a number of experiments were carried out to confirm the general appropriateness of the nZVI synthesis procedure for dechlorination of chlorinated hydrocarbons and to compare results to those reported in the literature. TCE was selected as the representative contaminant since it is well known that TCE can be readily reduced by nZVI (Zhang et al, 1998; Kim and Carraway, 2002; Schrick et al, 2002; Liu et al, 2005; Lien and Zhang, 2007). In these initial experiments, TCE was dechlorinated with both mono and bimetallic nZVI. In the case of monometallic nZVI (TCE = 100 mg/L, nZVI/TCE = 15 wt. ratio, Exp # 1, Table 3.1), about 25% of TCE was degraded at the end of 6 days (results not shown) whereas with bimetallic nZVI (TCE = 100 mg/L, nZVI/TCE = 15 wt. ratio, Pd=0.1 wt%, Exp # 2, Table 3.1) TCE was completely dechlorinated in one hour (Fig. 3.3). Observed degradation products were mainly ethane with a significant amount of ethylene. This is in agreement with the results obtained by Lien and Zhang (2001)! They used a higher ratio of nZVI/TCE (250) with

0.05wt% Pd and obtained ~50% TCE degradation at the end of 12 minutes while, in this study, (nZVI/TCE=15, Pd=0.1wt%) it took ~30 minutes to achieve the same % of degradation. This difference in rate of reaction is mainly due to the different nZVI/TCE ratio. A small amount of 1-butylene and 2-butylene also appeared as coupling products as the reaction proceeded further in this study; these products have been previously observed by others (Liu et al 2005). Unfortunately, due to analytical error only 32% of carbon mass balance was recovered in the experiment. The total recovery of chloride ions was 86.5% of the total expected chloride based on the disappearance of TCE (Fig 3.4). Though some TCE could be lost due to volatilization it is more likely that chloride is adsorbed onto the nZVI surface (Barnes et al, 2010).





医治疗 医心理学的 医内静脉 法建立法理事件

vet, a the flear le tafe

김 사람님님 같은



Fig 3.4: Evolution of Chloride due to degradation of 100 mg/L of TCE with 1.5g/L of nZVI and 0.1 wt% Pd at 25°C (Exp #2)

The initial set of 1,2-DCA experiments were designed based on the successful TCE dechlorination experiments with Pd/Fe bimetallic particles. Experiments were carried out for the degradation of 50 mg/L of 1,2-DCA (nZVI/ 1,2-DCA = 30 wt. ratio, Pd = 0.5 wt%, Exp # 3, Table 3.1). Pd loading was increased to 0.5 wt% based on the work of Orellana et al (2005) using 0.5 wt% Pd supported on SiO₂ to catalytically dehydrogenate 1,2-DCA at 300°C in the presence of gaseous H₂. After 3 days, no 1,2-DCA degradation was observed in this experiment. This unsuccessful result might be due to a lower nZVI/1,2-DCA ratio, compared to Lien and Zhang (2005) (where, nZVI/1,2-DCA = 250 wt. ratio, though this high ratio was unable to degrade 1,2-DCA). It should also be noted that the temperature used here was much lower than Orellana et al (2005) used. As such

the kinetic barrier needed for dechlorination under these specific conditions may not have been achieved. To address these two issues, a new set of experimental conditions were developed; these include the use of a higher nZVI/1,2-DCA ratio and increased temperature to decrease the reaction time and cut experiment time down.

As discussed in Chapter 2, Vanstone et al (2008) degraded about 30% of 15 mg/L of 1,2-DCA at room temperature in 12 days using large quantities of Zn^0 (Zn/1,2-DCA = 80,000 wt. ratio). They selected Zn^0 for this experiment because it is known as a highly reactive zero-valent metal with a standard reduction potential of -0.76V, which facilitates experiments with slowly reacting chlorinated compounds (e.g. 1,1-DCA, 1,2-DCA) (Vanstone et al, 2008). The one electron reduction potential for Zn^0 when reacting with 1,2-DCA is -0.558V at {Cl-} =1mM in the presence of all other species at unit activity (Arnold et al, 1999). However, Zn⁰ becomes easily oxidized in the sub-surface affecting its dechlorination capacity (Wang et al, 2010). Moreover, the use of Zn^0 in the field is limited as it releases Zn^{2+} , a harmful metal ion (Vanstone et al, 2008). On the other hand, the application of nZVI to dechlorinate COCs is a preferred option due to its high efficiency and environmentally benign nature; though the reduction potential for the reaction between 1,2-DCA and nZVI was not quantified because of its resistance to nZVI reactivity. Furthermore, Pd doped nZVI corrodes at a lower rate than Zn⁰ (Wang et al, 2010). The use of Pd together with Fe^{0} reduces the activation energy for dechlorination. For instance, for dechlorination of PCE with nZVI, the activation energy is 44.9 kJ/mol whereas, the use of Pd/Fe reduces the activation energy to 31.1 kJ/mol (Lien and Zhang, A. States of 1135 e posteriore 2007). Therefore, instead of Zn^0 , in this study Pd/Fe bimetallic nZVI has been selected. and a later of the e. is the case, they fills concurring
Since Vanstone's results indicate that a very large excess of reductant is required to carry 1,2-DCA dechlorination, a set of experiments were designed using a high concentration of highly active bimetallic Pd/Fe particles (1,2-DCA= 5 mg/L, Fe⁰/1,2-DCA = 4000 wt ratio, Pd = 0.2 wt%, Exp #4, Table 3.1), The result of this experiment is shown in Fig 3.5.



Fig 3.5: Degradation of 5 mg/L of 1,2-DCA with 20g/L of nZVI and 0.2 wt% Pd at 25°C (Exp # 4)

Comparison of 1,2-DCA in the reaction and blank vials suggests disappearance of 7% of the 1,2-DCA after 7 days; however no chloride was detected in the reaction vials. The amount of chloride that would have been evolved due to 7% of 1,2-DCA dechlorination is 0.125 mg/L; which is lower than the detection limit of the chloride analysis method used (0.5 mg/L). Hydrocarbons were not measured in this experiment. To confirm whether degradation was indeed taking place a second experiment under similar conditions was carried out; however, in this case, the 1,2-DCA concentration was

doubled (1,2-DCA = 10 mg/L, Fe⁰/1,2-DCA= 2000 wt ratio, Pd = 0.2 wt% Exp #5, Table 3.1). Here, 8% of the initial 1,2-DCA disappeared after 7 days (Fig 3.6) and ethane and propane were detected in the reaction vial as degradation products. This indicates that 1,2-DCA degradation was taking place, though at a relatively slow pace.



Fig 3.6: Degradation of 10 mg/L of 1,2-DCA with 20g/L of nZVI and 0.2 wt% Pd at 25°C (Exp # 5)

These results indicate that 1,2-DCA degradation can be achieved using Pd doped nZVI particles at relatively high loadings (e.g., nZVI/1,2-DCA= 2000 wt. ratio, nZVI= 20g/L), which is rather high compared to typical field applications that can go as high as 10 g/L. Given this, alternate nanometal formulations that include the addition of different hydrogen donors and catalysts were explored to lower nZVI loadings while increasing dechlorination rates.

3.5.2 Formic Acid as H₂ Source

Previous reports indicate that formic acid (HCOOH) is an efficient H_2 donor under acidic and neutral conditions (Kopinke et al, 2004). When formic acid is used together with Pd for dechlorination it has been observed that the relatively slow hydride mechanism dominates while under acidic or neutral conditions the fast radical mechanism through H atoms' is favoured (Kopinke et al, 2004) Indeed, it is reported that a combination of formic acid and Pd is very active for chlorobenzene dechlorination (Kopinke et al, 2004).

Based on the results reported in the literature, a suite of experiments was designed to decrease iron loadings and achieve dechlorination of 1,2-DCA using H₂ donors. The first system evaluated was formic acid/Pd. Thus, the next set of experiments were carried to dechlorinate 5 mg/L of 1,2-DCA in the presence of HCOOH as a H₂ source (nZVI/1,2-DCA=500 wt. ratio, Pd=0.2 wt%, HCOOH/1,2-DCA = 50 wt ratio, Exp #6 in Table 3.1) These conditions were selected based on a report by Kopinke et al (2004) (HCOOH/ Chlorobenzene = 45 wt ratio, Pd/Al₂O₃= 0.5 wt.%, 23°C). Our results using this formulation (Fig 3.7) show that at the end of 7 days, around 7% dechlorination of 1,2-DCA was achieved in the presence of HCOOH. This is similar to the results obtained in Exp #5 which required large ratios of nZVI/1,2-DCA (2000 wt. ratio) and definitely offers a better and more economic methodology. The presence of ethane and propane in the headspace analysis of reaction vial confirmed the degradation as no hydrocarbons were detected in the blank vials. A chloride analysis was unsuccessful because of interference by formate ions.



Fig 3.7: Degradation of 5 mg/L of 1,2-DCA with 2.5g/L of nZVI, 0.2 wt% Pd and 250 mg/L HCOOH at 25°C (Exp # 6)

To get a faster rate of dechlorination, the reaction temperature was increased to 35° C. (1,2-DCA=10 mg/L, nZVI/1,2-DCA=500 wt ratio, Pd=0.2 wt%, HCOOH/1,2-DCA=50 wt ratio, Exp #7 in Table 3.1 and Fig 3.8). At the end of 7 days, 11% degradation was observed (compared to 7% observed when running the same experiment at 25°C).

anyway to the exclusion alteration with a second approved that delive to tably

The heat parents of the latence of the second se

Line Filler and Hells and Arthur medium lighterers

, she a sudaa addida 7 maru danad ay dhe spacifu waxda ya amarata daha a

and was used as a history a dependent direction of Platt and Physics

63



Fig 3.8: Degradation of 10 mg/L of 1,2-DCA with 5g/L of nZVI, 0.2 wt% Pd and 500 mg/L HCOOH at 35°C (Exp # 7)

3.5.3 Optimization of Experimental Conditions

The experiments done so far showed that when HCOOH was used as a hydrogen source and Pd as a catalyst with nZVI at a temperature of 35°C, it was possible to degrade 1,2-DCA up to 11% at the end of 7 days. The next objective was to optimize the experimental conditions to get even better results. For that purpose, instead of HCOOH, isopropanol was selected as hydrogen source. This selection was based on a study done by Ukisu et al (1997) where 2-propanol was used as a hydrogen donor in the presence of NaOH and Rh as catalysts to dechlorinate chlorobenzene. This system displayed the ability to fully degrade chlorobenzene within 3 hours. Based on this specific results we attempted to use Rh-doped Fe nanoparticles for 1,2-DCA degradation (1,2-DCA=10 mg/L, nZVI/1,2-DCA= 1000 wt ratio, Rh=0.5 wt%, reaction medium isopropanol (IPA), NaOH/CI=3

E une par plant d'Alle Carlotte de la company

molar ratio, Exp # 8 in Table 3.1). The reaction temperature was kept at 35°C. However, at the end of 5 days, no 1,2-DCA degradation was observed in the reaction vial.

Previous reports indicate that the mixture of Rh-Pt as a catalyst (Rh=2 wt%, Pt=1 wt%) leads to successful dechlorination for the specific case of p-chlorotoluene at 27°C (Ukisu et al, 1998). The activity of the catalyst was shown to increase at higher Rh loadings (1 – 3 wt.%) (Ukisu et al, 1998). Thus, different amounts of Rh and Pt (Rh=0.2%, Pt=0.1%; Rh=0.7%, Pt=0.35% and Rh=1.42%, Pt=1.42%) were used to prepare several doped-nZVI formulations (Exp #9, 10 and 11 respectively in Table 3.1) in the presence of IPA and 0.61M NaOH but neither of these formulations could successfully dechlorinate 1,2-DCA (Appendix A). Ukisu et al (1998) have proposed that dechlorination activity is intimately linked to H₂ transfer and adsorption efficiencies. Therefore it can be proposed that the activation of H₂ by Rh and Pt might not be as efficient as that of Pd, leading to poor results in 1,2-DCA dechlorination.

Another nZVI formulation containing 0.2 wt% Pd, 0.2 wt% Rh and 0.1 wt% Pt was tested as well. In this particular case HCOOH was used as hydrogen donor for the catalytic hydrodechlorination of 10 mg/L of 1,2-DCA at 35° C (nZVI/1,2-DCA= 1000 wt ratio, Exp #12 in Table 3.1) instead of IPA. This particular formulation yielded 12% degradation of 1,2 DCA at the end of 7 days (Fig 3.9). This result is very similar to Exp # 7 suggesting that dechlorination takes places due to the presence of HCOOH and Pd rather than due to the use of Pt and Rh in this formulation. To confirm this, Exp#16 (nZVI/1,2-DCA= 1000 wt ratio, 0.2 wt% Pd, HCOOH= 500 mg/L, Table 3.1) was done

e andre de la collection de la contractione de la constitue de la constitue de la constitue de la constitue de



Fig 3.9: Degradation of 10 mg/L of 1,2-DCA with 10g/L of nZVI, 0.2 wt% Pd, 0.2 wt% Rh, 0.1 wt% Pt in 500 mg/L HCOOH at 35°C (Exp # 12)

Thus, up to this point, our results indicate that the most active formulation is nZVI/1,2-DCA=500 wt. ratio, Pd=0.2 wt% and HCOOH/1,2-DCA=50 wt ratio (Exp# 7, Table 3.1). Our next step was to improve the observed reaction rates through stabilization of the iron nanoparticles. It is well known that nZVI particles have tendency to form larger flocs and induce agglomeration which in turn, decreases reactivity. Several stabilizers have been used with nZVI to diminish agglomeration, decrease particle size and therefore increase reactivity (Ponder et al, 2001; He and Zhao, 2005; Saleh et al, 2007; He and Zhao, 2007). Among different stabilizers, carboxymethyl cellulose (CMC) has been proved to be the most effective (He and Zhao, 2007). As a result, the next set of experiments for this

project was conducted using CMC stabilized Pd/Fe bimetallic particles in the presence of HCOOH along with increasing the Pd content to 0.5%.(1,2-DCA=10 mg/L, nZVI/1,2-DCA=500 wt ratio, Pd=0.5 wt% HCOOH/1,2-DCA=50 wt ratio, CMC (90K) 1.5 wt% of nZVI, Exp # 13 in Table 3.1) It is expected that the addition of the stabilizer during synthesis would lead to smaller iron particle size, in turn increasing the available surface area of the nZVI formulation (Schrick et al, 2002; He and Zhao, 2005; He and Zhao, 2007; Sakulchaicharoen et al, 2010), leading to a higher 1,2-DCA degradation rates. Under these conditions, around 9% of 1,2- DCA degradation was observed after 7 days (Fig 3.10). Though it was expected that inclusion of CMC will make the particle size much smaller which will increase reactivity (Sakulchaicharoen et al, 2010) and a higher dose of Pd would give a faster reactivity of 1,2-DCA, it was observed that the rate of degradation was lower than that achieved in Exp #7 (11% dechlorination after 7 days) where no CMC was added. The failure of CMC to increase reactivity can be rationalized using the explanation by Phenrat et al (2009) who proposed that when stabilizers are adsorbed on nZVI surfaces, reactivity is decreased primarily due to blocking of the iron active sites. On the other hand, according to Wang et al (2009) though an increase in Pd loading increases the rate of dechlorination, this is true only at loadings below 0.2% Pd weight. A further increase in Pd content might hinder Fe⁰ corrosion and thus hydrogen formation by covering the active reactive sites of nZVI particles (Wang et al, 2009). Unfortunately no experiments were done without CMC and with Pd=0.2% at 45°C keeping all other experimental conditions same to confirm the role of temperature on increase in reactivity.



Fig 3.10: Degradation of 10 mg/L of 1,2-DCA with 1.5% CMC stabilized 5g/L of nZVI, 0.5 wt% Pd, in 500 mg/L HCOOH at 25° C (Exp # 13)

The next set of experiments was carried out at 45°C, while keeping all other experimental conditions unchanged (Exp #15, Table 3.1), to check if increase in temperature could improve the rate of degradation.

gene (20. days of the opposite equipment of the even of our backfield by the lower in the second of the second of

decision programme and suggification of the decision of the decision of the presence in the

医尿道的 经保持保证 医外外的 医外外的

hereither of the second as a second should be called a second be and the state of the second se

the



Fig 3.11: Degradation of 10 mg/L of 1,2-DCA with 1.5% CMC stabilized 5g/L of nZVI, 0.5 wt% Pd, in 500 mg/L HCOOH at 45°C (Exp # 15)

Fig 3.11 shows the results of this experiment. 17% of 1,2-DCA degradation was observed at the end of 7 days. A replicate of this experiment was done at the same time which also gave 17% degradation (Appendix F). To the best of our knowledge, this is the first time in which significant degradation of 1,2-DCA using nZVI has been achieved. The dechlorination reaction lead to the formation of ethane and propane as degradation products from the very first day of reaction; the amount of these gases increased as dechlorination progressed (Fig: 3.11). However, no vinyl chloride was present in the reaction vial.

Since the experiments continued up to $10\sim12$ days, it was necessary to check the reactivity of nanoscale iron particles towards the end. For that purpose, about 10 mg/L of

TCE was spiked in the reaction vial (1,2-DCA=10 mg/L, nZVI/1,2-DCA=500 wt. ratio, Pd=0.5 wt%, CMC=1.5%, HCOOH/1,2-DCA=50 wt. ratio) after 7 days (Exp #13, Table 3.1). GC analysis demonstrated instant dechlorination of TCE just after its spiking in the reaction vial, which confirmed the reactivity of nZVI particles even after 7 days. That means, bimetallic nZVI particles stabilized with CMC stay active even after 7 days of reactivity with 1,2-DCA. However, despite being highly reactive towards TCE, the nZVI particles could not do any further degradation of 1,2-DCA after 10~12 days. This might be explained by proposing a tendency for utilization of specific nZVI reactive sites by 1,2-DCA.

3.5.4 Degradation pathways during the dechlorination of 1,2-DCA

According to our results it can be concluded that the combination of nZVI, Pd as catalyst, HCOOH as hydrogen source, CMC as stabilizer at a temperature of 45°C (1,2-DCA=10 mg/L, nZVI/1,2-DCA=500 wt. ratio, Pd=0.5 wt%, CMC=1.5%, HCOOH/1,2-DCA=50 wt. ratio Exp #15, Table 3.1) produced the best performance of all experiments done to dechlorinate 1,2-DCA. The dechlorination of 1,2-DCA evolved about 44% of ethane and 56% of propane (Fig:3.11). After 7 days, a very small amount of ethylene, propylene, butane and butylene were also observed as coupling products. The observed formation of ethane is consistent with previous reports where it has been demonstrated that the use of Pd with nZVI as a catalyst for dechlorination of C2 halocarbons yields ethane (Lien and Zhang, 2005) Moreover, catalytic hydrodechlorination of 1,2-DCA also forms ethane as an end product when only Pd is used as catalyst (Kovalchuk et al, 2004; Orellana et al, 2005). Arnold et al (1999) have proposed a scheme for the dechlorination of vicinal (a

term used to describe the location of two identical atoms which are bonded to adjacent carbon atoms) chloroethanes and chlorinated ethylenes with Zn^0 . According to their hypothesis, reductive β -elimination is the only pathway through which species containing the requisite α , β pair of chlorine atoms react with Zn^0 . Therefore, the pathway of 1,2-DCA degradation on highly reactive Zn^0 is reductive β -elimination (vicinal elimination) involving net transfer of two electrons to the organohalides and the net loss of two nucleophiles (CI⁻) from neighbouring atoms (Fig 3.12).



Fig 3.12: Dechlorination of 1,2-DCA (Ref: Vanstone et al, 2008)

However, ethylene was not observed during the first 6~7 days in our experiment. This can be explained based on the fact that Arnold et al (1999) used Zn^0 without any catalyst, while in our case Pd, a very powerful hydrogenation catalyst was used. It is plausible to hypothesize that though ethylene was formed through β -elimination, it was immediately hydrogenated by the Pd catalyst. However, the pathway for the formation of propane has not been explained in the literature. In case of one molecule of ethane production, 2 hydrogen replaces 2 chloride ions from one molecule of 1,2-DCA. Whereas, it might be proposed that, when the degradation product is propane, 4 hydrogen replaces 4 chloride ions from 3 molecules of 1,2-DCA and forms 2 molecules of propane. Since both ethane

and propane are evolved from the very first day of degradation, it plausible to assume that propane is not coming from ethane.

Figure 3.13 shows a proposed reaction scheme that can be used to rationalize the observed results we obtained for dechlorination of 1,2-DCA with Pd/Fe in the presence of HCOOH:



$4CI^{-}$

2

 $\begin{array}{c} \mathrm{gamma}_{\mathrm{H}} & \mathrm{ga$

 $\frac{2H_2}{H} = \frac{2H_2}{H} = \frac{1}{H} = \frac{1}{H}$

where α is the $\mathbf{H}_{\mathbf{H}}^{\alpha}$ and $\mathbf{H}_{\mathbf{H}}^{\alpha}$ are $\mathbf{H}_{\mathbf{H}}^{\alpha}$ and $\mathbf{P}_{\mathbf{r}}^{\alpha}$ optime where α is a definition of the state of the state

nan Allen - Millinger englige ann a statistic and a statistic statistics

Fig. 3.13: Proposed 1,2-DCA transformation pathway while reacting with bimetallic Pd/Fe and HCOOH.

In order to compare the degradation products from dechlorination of both TCE and 1,2-DCA, the hydrocarbon evolved during TCE degradation were monitored along with that for 1,2-DCA. When TCE reacted with bimetallic Pd/Fe nanoparticles, (Exp#14, Table 3.1), the degradation products formed were 97% ethane, 1.17% propane and 1.19% butane. Due to the relatively high Pd concentration if some amount of ethylene was formed, it would rapidly be hydrogenated to ethane. No chlorinated byproducts like VC were observed during the reaction period. This contrasts with the results obtained by Lien and Zhang (2001) where with bimetallic Pd/Fe nanoparticles, TCE formed ~87% of ethane and 5.8~7.1% of ethylene within two hours (Lien and Zhang, 2001). On the other hand, Liu et al (2005) found 80% ethane and 20% C3-C6 coupling products due to TCE dechlorinatioin with nZVI.

One main drawback in this project during hydrocarbon analysis was that it was not possible to get a 100% carbon mass balance due to analytical error. Instead of 100%, only \sim 33% of carbon mass balance was possible to calculate. This might be explained as the fact, during calculation of amount of hydrocarbons, PV=nRT has been used were P= 1 atm pressure has been considered. But in reality, in the reaction vial with sample the pressure of hydrocarbons evolved after reactivity is likely to be more than 1 and that changes the hydrocarbon quantity affecting the carbon mass balance. In our case for 1,2-DCA dechlorination ethane comprised ~44% of the observed degradation products whereas for TCE, ethane was the main product covering (97% of total the hydrocarbons generated at the end of reaction). On the other hand, for the case of 1,2-DCA degradation

A conjuly association of the first state of the first state of the first state of the first state of the first

propane was the main end product (~56%). However, only 1.17% of TCE converted to propane.

offer of the William The advantioners and the second definition of the arthresis and the

3.6 SUMMARY AND CONCLUSIONS and a state of the state of t

The lower C-Cl bond dissociation energy (<47 Kcal/mole) of 1,2-DCA makes its dechlorination difficult in reductive environments. Nonetheless the thermodynamic free energy dechlorination value (-36.2 KJ/electrons) indicates that 1,2-DCA dechlorination is thermodynamically possible. Though different researchers report the inability of nZVI particles to dechlorinate 1,2 DCA, by exploring and adopting proper experimental conditions, 1,2-DCA can successfully be dechlorinated using nZVI formulations.

A very high concentration of nZVI (20g/L) catalyzed by 0.2 wt% Pd degraded 1,2-DCA by ~8% after 7 days of time. But this amount of nZVI is very high for field application. Therefore, to find a more convenient methodology, the combination of a hydrogen donor, nZVI and a catalyst was used. The use of formic acid as hydrogen donor with a lower concentration of nZVI (5g/L) and 0.2% Pd at 35°C was mildly successful in degrading 1,2-DCA (11% after 7 days). The reactivity was even higher (~18% after 7 days), when both the temperature and the Pd content were increased to 45°C and 0.5 wt% respectively and nZVI was stabilized using 1.5% CMC. It can be postulated that 1,2-DCA is breaking following dihaloelimination into ethylene and propylene; but due to the utilization of a high concentration of Pd, the ethylene and propylene get instantly hydrogenated into ethane and propane. Both ethane and propane evolve from the very first day of reaction and increase progressively. In this study, the present nZVI based dechlorination methodology offers a 1,2-DCA degradation of about 17%. Therefore the future step of this study should be to improve the rate of degradation by modifying the current experimental approach. Through an improvement in the experimental conditions such using different hydrogen donors (e.g. formates instead of formic acid), maintaining a lower pH during the reaction period as well as raising the temperature might develop a better dechlorination methodology.

Reserved S.J. (2010) Reading to be for the set of the set of the field on only process and the set of the structure of the set of the properties dependence on the system of the force of the system of the set of the field of the set of the set of the set of the force of the system of the set of

- 20. (* 1991). Berta Min Mindelson, 1997) nähet, Tin dig bil oper heftelles 4,2400A 1998 – Andreas Constituent (* 1999). 1998 – Andreas Heftelley (* <mark>Anglin alge, 1999).</mark> 1993 – Stor
- o i seogé g<mark>agounar a kara ranjor in a l'1</mark>1 annéa goleziak bagoganikak an ana 2011 - Alite Argan<u>it ar AM</u> Atasa marikabané se oraz moz, singtatan géodra alitega org 2011 - Alite Arganit<u>ar AM</u> Atasa marikabané se organitega (alitegatikat) ang
- ng Stagn Brianger Schwenz (1995). Brief, 19 million oor ar ar of schweregensbagt, softwereg - The second and a support share of the second schedule state of the spiniteral software (1995) - The second schedule schedule of the Schedule (1996). Brief and schedule schedule (1996).
- er er som de som kannen ander en som en er er som er s I verstande som er so I verstande som er s
- and "DARD" and subdemptions in East for a substrate of the mainter of the matter of practicity of the second s second s
- o for an monormalistic construction and the first of the formation of the statement of the second second second The formation of the second second second and the second second second second second second second second second The component of the formation of the second sec
- la an 1919 Francisco Carlos en acepto franceiras decidas hadden do 7708 de corporada. A comuna qui franch-maigne di Praticipa e 1800, 23, 29,735
- APRESIS recently available for functions of consists of manipulations of the law of the law of the second secon second sec
- Nophelie (1996) (1996) (1996) (1996) (1996) (1996) (1996) (1996) (1996) (1997) (1997) Methodolaet (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997) (1997)

3.7 REFERENCES

- Arnold, William A., William P. Ball, A. Lynn Roberts "Polychlorinated ethane reaction with zero-valent zinc" <u>Journal of Contaminant Hydrology</u>, 1999, 40, 183-200.
- Barnes J. Robert, O.Riba, M.N.Gardner, T.B.Scott, S.A.Jackman and I.P Thompson, "Optimization of nano-scale nickel/iron particles for the reduction of high concentration chlorinated aliphatic hydrocarbon solutions". Chemosphere, 2010, 79, 448-454.
- Bulushev, A. Dmitri, Julian R.H. Ross, "Vapour phase hydrogenation of olefins by formic acid over a Pd/C catalyst", <u>Catalysis Today</u>, 2011, 163, 42-46.
- Campbell,T.J, D.R.Burris, A.L.Roberts, J.R.Wells, "Trichloroethylene and tetrachloroethylene reduction in a metallic iron-water vapour batch system", <u>Environmental Toxicology and Chemistry</u>, 1997, 16, 625-630.
- De Wildeman, S., W. Verstraete, "The quest for microbial reductive dechlorination of C2 to C4 chloroalkanes is warranted". <u>Appl Microbiol Biotechnol</u>, 2003, 61, 94-102.
- Dyer, Mark, Erwin Van Heiningen, Jan Gerritse, "In situ bioremediation of 1,2-DCA under anaerobic condition" <u>Geotechnical and Geological Engineering</u>, 2000, 18, 313 334.
- Garson, Anthony and Florence Epron, "Use of formic acid a sreducing agent for applicatioin in catalytic reduction of nitrate in water", <u>Water Research</u>, 2005, 39, 3073-3081.
- He, Feng, Dongye Zhao, Chris Paul "Field assessment of carboxymethyl cellulose stabilized iron nanoparticles for insitu destruction of chlorinated solvents in source zones" Water Research, 2010, 44, 2360-2370
- He, Feng, Dongye Zhao, "Manipulating the size and dispersibility of zero-valent iron nanoparticles by use of carboxymethyl cellulose stabilizers" <u>Environmental</u> <u>Science & Technology</u>, 2007, 41, 6216 6221.
- He, Feng, Dongye Zhao, "Preparation and characterization of a new class of starchstabilized bimetallic nnanoparticles for degradation of chlorinated hydrocarbons in water". Environmental Science & Technology 2005, 39, 3314 - 3320
- He, Feng, Dongye Zhao, "Stabilization of Fe-Pd nanoparticles with CMC for enhanced transport and dechlorination of TCE in soil and groundwater". <u>Ind. Eng.</u> <u>Chem.</u> Res, 2007, 46, 29-34.
- Johnstone R.A.W and Anna H. Wilby, "Heterogeneous catalytic transfer hydrogenation and its relation to other methods for reduction of organic compounds". <u>Chem. Rev</u>. 1985, 85, 129-170.
- Kim, Y-H, Elizabeth R Carraway, "Reductive dechlorination of TCE by zero-valent bimetals" <u>Environmental Technology</u>, 2003, 24, 69-75.
- Klecka, G.M, C.L. Carpenter, S.J.Gonsior, "Biological transformations of 1,2-DCA in sub-surface soils and groundwater" Journal of Contaminant Hydrology, 1998, 34, 139 154.
- Kopinke Frank-Dieter, Katrin Mackenzie, Koechler, Robert Georgi Anette, "Alternative sources of H2 for hydrodechlorination of chlorinated organic

compounds in water on Pd catalysts". <u>Applied Catalysis A: General, 2004, 271, 119-128</u>.

- Kovalchuk, Vladimir I., Julie d'Itri, "Catalytic chemistry of chloro- and chlorofluorocarbon dehalogenation: from macroscopic level understanding". <u>Applied Catalysis A: General, 2004, 271, 13-25</u>.
- Larson, A.Richard and Eric J. Weber, "Reaction Mechanisms in Environmental Organic Chemistry" Lewis Publishers, 1994.
- Li, X.Q., D.W. Elliott, and W.X. Zhang, "Zero-valent iron nanoparticles for abatement of environmental pollutants: Materials and engineering aspects". <u>Critical Reviews in Solid State and Materials Sciences</u>, 2006, 31, 111-122.
- Lien, Hsing-Lung, Wei-Xian Zhang, "Hydrodechlorination of chlorinated ethanes by nanoscale Pd/Fe bimetallic particles". Journal of Environmental Engineering, 2005, 131.
- Lien, Hsing-Lung, Wei-Xian Zhang, "Nanoscale iron particles for complete reduction of chlorinated ethenes" <u>Colloids and Surfaces A: Physiochemical & Engineering Aspects</u>, 2001, 191, 97-105.
- Lien, Hsing-Lung, Wei-Xian Zhang, "Nanoscale Pd/Fe bimetallic particles: catalytic effects of palladium on hydrodechlorination". <u>Applied Catalysis B:</u> <u>Environmental</u>, 2007, 77, 110-116.
- Liu, Y.Q., S.A. Majetich, R.D. Tilton, D.S. Sholl, and G.V. Lowry, "TCE dechlorination rates, pathways, and efficiency of nanoscale iron particles with different properties". <u>Environmental Science & Technology</u>, 2005,9, 1338-1345.
- Omole, A.Marcells, Isaac O. K'Owino, Omowunmi A. Sadik, "Palladium nanoparticles for catalytic reduction of Cr(VI) using formic acid". <u>Applied</u> <u>Catalysis B: Environmental</u>, 2007, 76, 158-167.
- Orellana, F., G. Pecchi, P. Reyes, "Selective hydrodechlorination of 1,2-DCA over Pd-Sn/SiO2 catalysts". J. Chil. Chem. Soc, 2005, 50.
- Phenrat, Tanapon, Y.Liu, R.D.Tilton and G.V.Lowry,"Adsorbed polyelectrolyte coatings decrease Fe0 nanoparticle reactivity with TCE in water: conceptual model and mechanisms" <u>Environmental Science & Technology</u>, 2009, 43, 1507 1514.
- Sakulchaicharoen, Nataphan, D.M.O'Carroll and Jose .E. Herrera, "Enhanced Stability and dechlorination activity of pre-synthesis stabilized nanoscale FePd particles". Journal of Contaminant Hydrology. 2010, 118, 117-127.
- Schrick, B., J.L. Blough, A.D. Jones, and T.E. Mallouk, "Hydrodechlorination of trichloroethylene to hydrocarbons using bimetallic nickel-iron nanoparticles". <u>Chemistry of Materials</u>, 2002, 14, 5140-5147.
- Song Hocheol, Elizabeth R Carraway, "Reduction of chlorinated ethanes by nanosized zero-valent iron: kinetics, pathways and effects of reaction condition". Environmental Science & Technology, 2005, 39, 6237 – 6245
- Ukisu, Yuji, Satoshi Kameoka, Tatsuo Miyadera, "Rh-based catalysts for catalytic dechlorination of aromatic chloride at ambient temperature" <u>Applied Catalysis B:</u> <u>Environmental</u>, 1998, 18, 273-279.
- Urbano, F.J., J.M. Marinas, "Hydrogenolysis of organohalogen compounds over palladium supported catalysts". Journal of Molecular Catalysis A: Chemical 2001, 173, 329 345.

- Vanstone, Nancy, M. Elsner, G. Lacrumpe-Couloume, S. Mabury, B.S. Lollar, "Potential for identifying abiotic chloroalkane degradation mechanisms using carbon isotope fractionation" <u>Environmental Science & Technology</u>, 2008, 42, 126-132.
- Wang, Xiangyu, Chao Chen, Huiling Liu, Jung Ma, "Characterization and evaluation of catalytic dechlorination activity of Pd/Fe bimetallic nanoparticles" <u>Ind. Eng.</u> <u>Chem. Res</u> 2008, 47, 8645-8651.
- Wang, Xiangyu, Chao Chen, Ying Chang, Huiling Liu, "Dechlorination of chlorinated methanes by Pd/Fe bimetallic nanoparticles" <u>Journal of Hazardous</u> <u>Materials</u> 2009, 161, 815-823.
- Wang Zhiyuan, Weilin Huang, Pingan Peng, Donna E. Fennell,"Rapid transformation of 1,2,3,4-TCDD by Pd/Fe catalysts". <u>Chemosphere</u>, 2010, 78, 147-151.
- Zaan, Bas van der, J.De Weert, Hubb Rijnaarts, W.M.de Vos, Hauke Smidt, Jan Gerritse, "Degradation of 1,2-dichloroethane by microbial communities from river sediment at various redox conditions". <u>Water Research</u>, 2009, 43, 3207-3216.
- Zhang, W.X., "Nanoscale iron particles for environmental remediation: An overview". Journal of Nanoparticle Research, 2003, 5, 323-332.
 - Zhang, W.X., C.B. Wang and H.L. Lien, "Treatment of chlorinated organic contaminants with nanoscale bimetallic particles", <u>Catalysis-Today</u>, 1998, 40, 387-395

出现 计行行 化合物 医神经 人名德尔 医白色 建成 医月经病 化合理 化合理 化合理 化化合理

Refer to be a state of the boot of the state of state gas when each of the state of the state

a hannen er som standelsen er staten i staten i staten i staten i staten i staten er staten er staten i staten

e and a control of price of the state of the transformer of the book trapped

na landa na malas, comas locorde Fe¹ come ha antino fore lander da

. A selection of the distribution and has been and the dealers for the factor in the second second

The selection in advised states, while we will a fight open data and for any sub-chair open.

s non est the service of quicky from MINE and notices relies standards holder

Zhang, Wei-Xian, Daniel W Elliot, "Application of iron nanoparticles for groundwater remediation" <u>Remediation Spring 2006.</u>

last to the first first and the set of the s

4.0 SUMMARY AND RECOMMENDATIONS

4.1 SUMMARY AND CONCLUSIONS

Through a combination of nZVI, a hydrogen source, a hydrogenation catalyst, and a nZVI suspension stabilizer a suitable reductive environment to dechlorinate 1,2-DCA has been created. This particular approach can be applied in the field relatively easily since it requires neither high temperature nor gaseous hydrogen, unlike catalytic hydrodechlorination. In addition reaction times are fast meaning it will not take decades

to centuries to breakdown 1,2-DCA, as required for subsurface microorganisms.

The results from this study indicate that:

• nZVI by itself cannot initiate 1,2 DCA dechlorination, even at unusually high Fe⁰/1,2 DCA ratios. The presence of a hydrogenation catalyst is required in order to provide highly active hydrogen species able to carry out 1,2 DCA degradation.

A hydrogen donor and a noble metal catalyst is required to decrease the amount of nZVI required to degrade 1,2-DCA. This is because, in the nZVI system, hydrogen generation occurs through Fe^0 corrosion that is used during the transformation of the chlorinated compounds into benign products. Introducing an efficient H₂ source along with an active hydrogenation catalyst can substitute a part of the hydrogen required from nZVI and reduce reticent organic halides

through reaction with the active form of hydrogen adsorbed in the catalyst. This will need less excess iron making the process more economical.

ter her with an happened a transferration at the transfer denses (it provides and

• As demonstrated in this work when compared to the other catalysts (e.g., Rh and Pt) Pd is the most effective, removing chlorines from 1,2-DCA thus yielding ethane. Rh and Pt were unable to catalyze 1,2-DCA degradation despite the presence of another hydrogen source (i.e., isopropanol). This might be due to the inadequate activation of hydrogen by Rh and Pt for 1,2-DCA dechlorination.

universita de la construcción de la

êns la color aplaced navelenka în car na cola le n

• The degradation pathway observed for 1,2-DCA followed a β -elimination mechanism followed by a very rapid hydrogenation of the unsaturated products formed (e.g. ethylene and propylene) leading to the formation of ethane and propane.

4.2 RECOMMENDATIONS

This work found an appreciable amount of 1,2-DCA dechlorination using nanoscale zerovalent iron particles, which was, to date, almost impossible. After 7 days ~17% of the 1,2-DCA was dechlorinated and little additional was degraded up to 12 days. Under the existing experimental conditions this was the highest observed conversion. However, it can be proposed that by optimizing the experimental conditions, 1,2-DCA conversion can be increased even more. For that purpose, a number of steps can be taken for future work. Since hydrogen donor plays a vital role in the dechlorination of 1,2-DCA, a more appropriate hydrogen source could be examined. Some reports indicate that formate salts are superior to formic acid as hydrogen donors (Rajagopal and Spatola (1995). Therefore, a next step in improving 1,2 DCA dechlorination efficiency is to carry out the experiments using formate salts instead of formic acid (e.g., Cs and K).

Since the rate of chemical reaction increases with an increase in temperature, the next step of the experiments in the lab can be carried over at higher reaction temperatures. This will help in reducing the duration of the experiments when evaluating different nZVI formulations.

Another important factor controlling the reactivity is the pH. At low pH, Fe^{0} corrosion is faster leading to the production of more hydrogen through reduction of water which can potentially increase the dechlorination rates. Therefore, if the pH can be controlled throughout the duration of the reactivity, that might help to improve the dechlorination rate.

APPENDIX A



Comparison of reactivity between 1,2-DCA and different catalysts

Figure A: Comparison of reactivity of different catalysts when nZVI=10 mg/L, 1,2-DCA=10 mg/L, temp=35°C i) Rh=0.5% 1% ii) Rh=0.2%, Pt=0.1%, iii) Rh=0.7%, Pt=0.35%, iv) Rh=0.2%, Pd=0.2%, Pt=0.1% v) blank

She in Earla

化氯化化 美国教师 静脉放动 化二氯合合物 或可能完成 知识

ίç-1



Chromatogram from GC/MS during dechlorination of 1,2-DCA

Fig B: Chromatogram for degradation products of 1,2-DCA:- retention time for i) ethane = 3.15 min ii) propane = 7.6 min, iii) 1,2-DCA

APPENDIX C



Fig C-1: Calibration Curve for TCE



Fig C-2: Calibration Curve for 1,2-DCE

ç.

APPENDIX D



Fig D-1: Calibration Curve for 1,2-DCA (Headspace Analysis)



Fig D-2: Calibration Curve for 1,2-DCA (Liquid Analysis)

19

APPENDIX E



Calibration Curves for Hydrocarbons





Fig E-2: Calibration Curve for Propane (Headspace Analysis)

APPENDIX G





Fig G: Degradation of 10 mg/L of 1,2-DCA with 20g/L of nZVI, 0.2 wt% Pd, at 25°C (Results of Exp # 8 & 11)

ц.

<u>ç</u>,

- **G** (







G

No.	Conc of Chlorinated Solvent (mg/L)	Conc of Fe ⁰ (gm/L)	% of Pd	% of CMC	Conc of HCOOH (mg/L)	Temp (°C)	% of Degradation after 7 days	Degradation C-H	Products Cl-
1	1,2- 25	1	0.1	0.8	x	25	no	not done	not done
	DCA=	а 					no		
2 	DCA = 25	1.5	0.5	0.8	X	25	degradation	not done	not done
3	1,2- DCA= 50	5	0.2	x	X	60	no degradation	not done	0%
4	1,2- DCA= 50	1.5	0.5	X	X	60	no degradation	not done	0% 6/535
5	1,2- DCA= 5	20	0.2	X	X	25	8%	Ethane and propane	below det Stion limit
6	1,2- DCA= 5	2.5	0.2		250	25	7%	Ethane and propane	below detection limit
7	1,2- DCA= 5	20	0.2	x	X	25	6%	not done	not done
8	1,2- DCA= 10	20	0.2	X	X	25	8%	Ethane and propane	not done
	· •	·			• •				

 Table I-1: Experiments with 1,2-DCA

No.	Conc o Chlorina Solver	of ated nt	Conc of Fe ⁰ (gm/L)	% of Pd	% of CMC	Conc of HCOOH (mg/L)	Temp (°C)	% of Degradation after 7 days	Degra Proc	dation lucts
	(mg/L	J :		100 - 100 -	1				C-H	CI-
9	1,2- DCA=	10	5	0.2	x	500	25	9%	not done	not done
10	1,2- DCA=	10	5	0.2	x	500	25	8%	not done	8 times more
11	1,2- DCA=	10	20	0.2	x	X	25	8%	Ethane and propane	2.6 times more
12	1,2- DCA=	10	10	0.2	x	500	35	12%	Ethane and propane	60%
13	1,2- DCA=	10	5	0.2	X	500	25	8%	Ethane and propane	not done
14	1,2- DCA=	10	20	0.2	x	x	25	17% decrease after 25 days	Ethane and propane	not done
	TCE =	10	_	0.5		500	25	100%		*
15	1,2- DCA=	5	5	0.5		500	25	.9%	not done	not done

 Table I-2: Experiments with 1,2-DCA (contd)

 Table I-3: Experiments with 1,2-DCA (contd)

No	Conc of Chlorinated	Conc of Fe ⁰	% of	% of	Conc of HCOOH	Temp	% of	Degrad Produ		
	Solvent (mg/L)	(gm/L)	Pđ	CMC	(mg/L)	(°C)	Degradation	С-Н	Cl-	
• .	•	: :						40%		<u> </u>
1	1,2- DCA= 10	5	0.5	1.5	500	25	12%	ethane, 60% propane	not done	
2	1,2- DCA= 10	5	0.5	1.5	500	45	17%	44% ethane, 56% propane	not done	
		4 13						49%		
3	1,2- DCA= 10	5	0.5	1.5	500	45	15%	ethane, 51% propane	not done	n i Anno Sin Anno Anno Anno
<u> </u>]
										not Joan

		general de la composition de l		n na sana sa Marina sa sa				алар алар алар алар Алар	
	Conc of							Degra	adation ducts
No.	Chlorinated Solvent (mg/L)	of Fe ⁰ (gm/L)	% of Pd	% of Pt	% of Rh	Temp (°C)	% of Degradatio	n C-H	Cl-
; 1	1,2- DCA= 5	5	X	X	0.5	35	no degradation	not done	,not done
2	1,2- DCA= 10	10	X	0.1	0.2	35	no degradation	not done	nōt done
3. 3.	1,2- DCA= 10	10	X	1.42	1.42	35	no degradatio	not done	not
4	1,2- DCA= 10	10		0.35	0.7	35	no degradation	not done	not done
5	1,2- DCA= 10	10		0.1	0.2	35	12%	Ethane and Propane	not done
				: : : : :					
						• • • • • • • • • • • • • • • • • • •			and Smith
			1	; • .: ;					1903) (C.C.T. <mark>S</mark> .C.
				1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1					
		· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·	,		an a shara an		a series and a series a

Table I-4: Experiments with 1,2-DCA (contd)

Table J-1: Experiments with TCE											
	Conc of Chlorinated	Conc	% of	% of	нсоон	% of	Degradation Products				
	Solvent (mg/L)	(gm/L)	Pd	CMC	(mg/L)	Degradation	C-H	Cl-			
1	TCE = 700	10	X	• X • • •	X • • • •	32%	not done	not done			
2	TCE = 700	10	x	С. Х.	x	26%	not done	33% <u>of</u> expected Cl-			
3	TCE = 100	1.5	X	× ×	X	61%	not done	nct lone			
4	TCE = 100	1.5	() - X	X	x	37%	Ethane and Ethene	not done			
5	TCE = 100	1.5	. X .	X	X	31%	Ethane and Ethene	11% of expected Cl-			
- 6	TCE = 4.4	1.9	X	x	х	31%	not done	not done			
7	TCE = 100	1.5	. 0.1	x	X	100%	not done	not done			
8	TCE = 100	1.5	• 0.1	X X	X	100%	not done	50% of expected Cl-			
9	TCE = 100	1.5	0.1	X	X	100%	not done	not done			
10	TCE = 10	5	0.2	x	500	100%	not done	not done			
11 8	TCE = 100	1.5	0.1	x	x	100%	not done	not done			
-12	TCE = 100	5	0.2	x	500	100%	not done	not done			

. (;

	Conc of	Conc					Degradatio	n Products
No.	Chlorinated Solvent (mg/L)	of Fe ⁰ (gm/L)	% of Pd	% of CMC	HCOOH (mg/L)	% of Degradation	С-Н	Cl-
- 13	TCE = 100	1.5	0.1	X	× ,	100%	Ethane, Ethene, 1- Butene, 2- Butene	117% of expected Cl-
14	TCE = 100	1.5	0.1	X	X	100%	Ethane, Ethene, 1- Butene, 2- Butene	118% cf expected Cl-
15	TCE = 100	1.5	0.1	X	X	100%	Ethane, Ethene, 1- Butene, 2- Butene	98% of expected Cl-
16	TCE = 100	1.5	0.1	X	X	-100%	Ethane, Ethene, 1- Butene, 2- Butene	84% of expected Cl-
17	TCE = 100	1.5	0.1	X	X	100%	Ethane, Ethene, 1- Butene, 2- Butene	126% cf expected Cl-
18	TCE = 100	1.5	0.1	X	X	100%	Ethane, Ethene, 1- Butene, 2- Butene	not done

 Table J-2: Experiments with TCE (contd)
APPENDIX J-3

No.	Conc of Chlorinated Solvant		% of	% of	HCOOH	% of	Degradation Products	
	(mg/L)	(gm/L)	ru		(IIIg/L)	Degradation	С-Н	Cl-
19	TCE = 100	5	0.2	X	500	100%	Total = 31.26% (Ethane = 31.11%, Propane = 0.073% Butane = 0.307)	not cone
20	TCE= 10	5	0.5	1.5	500	100%	Total = 31.26% (Ethane = 31.11%, Propane = 0.073% Butane = 0.307%)	not done
. 21	TCE= 10	5	0.5	1.5	500	100%	Total = 30.44% (Ethane = 30.33%, Propane = 0.089% Butane = 0.023%)	not done
22	TCE= 10	5	0.5	1.5	500	100%	Total = 25.34% (Ethane = 24.26%, Propane = 0.199% Butane = 1.073%)	not
23	TCE= 10	5	0.5	1.5	500	100%	Total = 34.79% (Ethane = 34.41%, Propane = 0.144% Butane = 0.232%)	not done
24	TCE= 10	5	0.5	1.5	500	100%	Total = 54.81% (Ethane = 54.37%, Propane = 0.403% Butane = 0.44%)	not done
25	TCE= 10	5	0.5	1.5	500	100%	Total = 31.197% (Ethane = 31.06%, Propane = 1.512% Butane = 0.136%)	not d_{ξ} ne

 Table J-3: Experiments with TCE (contd)